



King's Research Portal

DOI: 10.1016/j.physrep.2016.06.005

Document Version Peer reviewed version

Link to publication record in King's Research Portal

Citation for published version (APA): Marsh, D. J. E. (2016). Axion cosmology. *Physics Reports*, *643*, 1-79. https://doi.org/10.1016/j.physrep.2016.06.005

Citing this paper

Please note that where the full-text provided on King's Research Portal is the Author Accepted Manuscript or Post-Print version this may differ from the final Published version. If citing, it is advised that you check and use the publisher's definitive version for pagination, volume/issue, and date of publication details. And where the final published version is provided on the Research Portal, if citing you are again advised to check the publisher's website for any subsequent corrections.

General rights

Copyright and moral rights for the publications made accessible in the Research Portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognize and abide by the legal requirements associated with these rights.

•Users may download and print one copy of any publication from the Research Portal for the purpose of private study or research. •You may not further distribute the material or use it for any profit-making activity or commercial gain •You may freely distribute the URL identifying the publication in the Research Portal

Take down policy

If you believe that this document breaches copyright please contact librarypure@kcl.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.

Accepted Manuscript

Axion cosmology

David J.E. Marsh

 PII:
 S0370-1573(16)30155-7

 DOI:
 http://dx.doi.org/10.1016/j.physrep.2016.06.005

 Reference:
 PLREP 1911

To appear in: Physics Reports

Accepted date: 16 June 2016



Please cite this article as: D.J.E. Marsh, Axion cosmology, *Physics Reports* (2016), http://dx.doi.org/10.1016/j.physrep.2016.06.005

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

Manuscript

Axion Cosmology

David J. E. $Marsh^1$

King's College London, Strand, London, WC2R 2LS, United Kingdom

Abstract

Axions comprise a broad class of particles that can play a major role in explaining the unknown aspects of cosmology. They are also well-motivated within high energy physics, appearing in theories related to CP-violation in the standard model, supersymmetric theories, and theories with extra-dimensions, including string theory, and so axion cosmology offers us a unique view onto these theories. I review the motivation and models for axions in particle physics and string theory. I then present a comprehensive and pedagogical view on the cosmology and astrophysics of axion-like particles, starting from inflation and progressing via BBN, the CMB, reionization and structure formation, up to the present-day Universe. Topics covered include: axion dark matter (DM); direct and indirect detection of axions, reviewing existing and future experiments; axions as dark radiation; axions and the cosmological constant problem; decays of heavy axions; axions and stellar astrophysics; black hole superradiance; axions and astrophysical magnetic fields; axion inflation, and axion DM as an indirect probe of inflation. A major focus is on the population of ultralight axions created via vacuum realignment, and its role as a DM candidate with distinctive phenomenology. Cosmological observations place robust constraints on the axion mass and relic density in this scenario, and I review where such constraints come from. I next cover aspects of galaxy formation with axion DM, and ways this can be used to further search for evidence of axions. An absolute lower bound on DM particle mass is established. It is $m_a > 10^{-24}$ eV from linear observables, extending to $m_a \gtrsim 10^{-22}$ eV from non-linear observables, and has the potential to reach $m_a \gtrsim 10^{-18}$ eV in the future. These bounds are weaker

Preprint submitted to Elsevier

ACCEPTED MANUSCRIPT

if the axion is not all of the DM, giving rise to limits on the relic density at low mass. This leads to the exciting possibility that the effects of axion DM on structure formation could one day be detected, and the axion mass and relic density measured from cosmological observables.



Conte	nts
1 Int	roduction
2 Mo	odels
2.1	The QCD Axion
	2.1.1 The Strong- CP Problem and the PQ Solution
	2.1.2 PQWW axion
	2.1.3 KSVZ axion
	2.1.4 DFSZ axion
2.2	Anomalies, Instantons, and the Axion Potential
2.3	Couplings to the Standard Model
2.4	Axions in String Theory
3 Pr	oduction and Initial Conditions
3.1	Symmetry Breaking and Non-Perturbative Physics
3.2	The Axion Field During Inflation
	3.2.1 PQ symmetry unbroken during inflation, $f_a < H_I/2\pi$
	3.2.2 PQ symmetry broken during inflation, $f_a > H_I/2\pi$.
3.3	Cosmological Populations of Axions
	3.3.1 Decay Product of Parent Particle
	3.3.2 Decay Product of Topological Defect
	3.3.3 Thermal Production
	3.3.4 Vacuum Realignment
4 Th	e Cosmological Axion Field
4.1	Action and Energy Momentum Tensor
4.2	Background Evolution
4.3	Misalignment Production of DM Axions
	4.3.1 Axion-Like Particles
	4.3.2 The QCD Axion

	1
	T
	2
	3
	4
	5
	6
	7
	8
	0
1	ב ר
T	0
1	1
1	2
1	3
1	4
1	5
1	6
1	7
1	<i>'</i>
T	8
Τ	9
2	0
2	1
2	2
2	3
2	4
⊿ ∩	т Г
2	5
2	6
2	7
2	8
2	9
3	0
3	1
2	2
د ~	⊿ ∽
3	3
3	4
3	5
3	б
3	7
3	8
2 2	q
1	0
4	1
4	T
4	2
4	3
4	4
4	5
Δ	6
1	-
4	1
4	8
4	9
5	0
5	1
5	2
5	_
5	.≺
F	3 1
5	3 4 5
55	3 4 5
5 5 5	3 4 5 6
5 5 5 5	3 4 5 6 7
5 5 5 5 5 5	3 4 5 6 7 8
5 5 5 5 5 5 5 5 5	3 4 5 6 7 8 9
555556	3 4 5 6 7 8 9 0
5555566	345678901
555555666	3456789012
5555556666	34567890122

ACCEPTED MANUSCRIPT

K

		4.4.1	Initial Conditions	. 58
		4.4.2	Early Time Treatment	. 60
		4.4.3	The Axion Effective Sound Speed	. 61
		4.4.4	Growth of Perturbations and the Axion Jeans Scale	. 63
		4.4.5	Transfer Functions: Relation to WDM and Neutrinos .	. 67
	4.5	Non-lin	nearities and the Schrödinger Picture	. 71
	4.6	Simula	ating axion DM	. 74
	4.7	My Tv	vo Cents on BEC	. 76
5	Con	straint	ts from the CMB and LSS	81
0	5.1	The P	rimary CMB	81
	5.2	The M	latter Power Spectrum	. 84
	5.3	Combi	ned Constraints	. 88
	5.4	Isocury	vature and Axions as a Probe of Inflation	. 89
6	Gal	axy Fo	rmation	94
	6.1	The H	alo Mass Function	. 94
	6.2	Constr	caints from High- z and the EOR \ldots	. 97
	6.3	Halo I	Density Profiles	. 100
	6.4	ULAs	and the CDM Small Scale Crises	. 104
7	Axi	ons an	d Accelerated Expansion	109
	7.1	Axions	s and the Cosmological Constant Problem	. 109
	7.2	Axion	Inflation	. 114
		7.2.1	Natural Inflation and Variants	. 116
		7.2.2	Axion Monodromy	. 118
0	0			110
8	Gra	vitatio	and Interactions with Black Holes and Pulsars	119
	8.1	Black .	Hole Superradiance	. 120
	8.2	Pressu	re Oscillations and Pulsar Timing	. 122
9	Nor	on-Gravitational Interactions 124		
	9.1	Stellar	Astrophysics	. 125

 $\begin{array}{r} 48\\ 49\\ 50\\ 51\\ 52\\ 53\\ 55\\ 57\\ 58\\ 59\\ 60\\ 62\\ 63\\ 64 \end{array}$ 65

ACCEPTED MANUSCRIPT

9.2	"Light Shining Through a Wall"		
9.3	Vacuum Birefringence and Dichroism		
9.4	9.4 Axion Mediated Forces		
9.5	Direct Detection of Axion DM	131	
	9.5.1 Haloscopes and ADMX	131	
	9.5.2 Nuclear Magnetic Resonance and CASPEr	132	
9.6	Heavy Axions and Axion Decays	135	
9.7	Axion Dark Radiation	138	
9.8	Axions and Astrophysical Magnetic Fields		
	9.8.1 CMB Spectral Distortions	140	
	9.8.2 X-ray Production	142	
9.9	Cosmological Birefringence		
10 Coi	ncluding Remarks	145	
A The	eta Vacua of Gauge Theories	148	
	in the second		
B EF'	Γ for Cosmologists	150	
C Frie	edmann Equations	152	
D Cos	smological Fluids	153	
2 000	inorogram / innas	100	
Е Вау	ves Theorem and Priors	155	
F Deg	generacies and Sampling with ULAs	157	
${ m G}~{ m She}$	th-Tormen Halo Mass Function	160	

	1
	2
	3
	Δ
	-
	5
	6
	7
	8
	0
_	9
Τ	0
1	1
1	2
1	2
1	1
Ŧ	4
1	5
1	6
1	7
1	8
1	0
Ŧ	9
2	0
2	1
2	2
ົ	2
2	<u>з</u>
2	4
2	5
2	6
2	7
~	~
2	8
2	9
3	0
3	1
2	- -
3	2
3	3
3	4
3	5
2	6
2	0
3	7
3	8
3	9
4	0
1	1
4	1
4	2
4	3
4	4
۵	5
-	2
4	ю
4	7
4	8
4	9
5	0
5	1
5	1
5	2
5	3
5	4
5	5
5	ر م
5	6
5	7
5	8
5	a
ر ح	2
6	U
6	1
6	2
~	
6	3

1. Introduction

As Weinberg said, "physics thrives on crisis" [1]. In 1989 when Weinberg wrote that famous review, he said that physics was short on crises. Happily, these days, thanks in large part to the advent of precision cosmology, it is full of them.

The standard cosmological model is described by just six numbers: two for initial conditions, one for dark matter (DM), one for the baryons, one for cosmic structure formation and reionization, and one for the cosmological constant (c.c.). Each of these numbers presents a problem for our understanding of fundamental physics. The initial conditions appear close to scale invariant: producing such initial conditions requires a period of rapid acceleration (or slow deceleration) in the early Universe, a state of affairs that cannot be realised in the usual hot big bang. Dark matter constitutes the vast majority of matter in the Universe, and no particle in the standard model of particle physics can fit the role of being stable, cold, and weakly coupled. The standard model also provides no obvious way to tip the matter-anti-matter asymmetry in favour of baryons instead of anti-baryons. Structure formation and reionization are sensitive to the initial conditions, matter content, and complex astrophysical processes in ways that we are only just learning. And then finally there is Weinberg's problem of the c.c..

In 1989 Weinberg selected just the c.c. as a major problem: even without precision cosmology, it was clear that the theoretical expectations about this number were wildly off the mark. All of the other problems were known at that time, but without the precision measurements we have today their importance could easily be debated and there was no need to call "crisis." We are no longer in that position of blissful ignorance: all the numbers in the standard cosmological model need to be considered and their theoretical implications taken seriously.

In seeking a unified view of the problems presented by precision cosmology, we will focus in this review on a class of particles known as axions. Ever since the earliest days of the QCD axion it has been realised that it offers an exceptionally good DM candidate. With the advent of string theory and the corresponding profusion of axion-like particles (ALPs), axions have come to play important roles in inflation and the generation of cosmological initial conditions, and in the solution of the c.c. problem. String axions also offer the possibility to resolve problems of structure formation inherent in more vanilla models of DM. Axions can even assist in baryogenesis thanks to their role in CP-violation. A summary of constraints and probes of axion cosmology, as a function of axion mass, is shown in Fig. 1.

A large portion of this review will focus on ALPs in the mass range

$$10^{-33} \text{ eV} \lesssim m_a \lesssim 10^{-18} \text{ eV}$$
. (1)

I will refer to axions in this mass range as ultralight axions, or ULAs. The lower bound is of order the present day Hubble constant, $H_0/h = M_H = 2.13 \times 10^{-33} \text{ eV} = 100 \text{ km s}^{-1} \text{ Mpc}^{-1}$, and reflects constraints on axion dark energy (DE). The upper bound is related to the baryon Jeans scale, and reflects a distinctive role of ULAs in cosmological structure formation and reionization. This vast range of axion masses can be probed using the tools that led us to our crises in the first place, i.e. those of precision cosmology: the cosmic microwave background (CMB), large scale structure (LSS), galaxy formation in the local Universe and at high redshift, and by the epoch of reionization (EOR).

It is worth noting here, for clarity, that the word "axion" can take on a variety of meanings. It was first coined by Wilczek [2] to name the particle associated to the axial anomaly in QCD and the Peccei-Quinn [3] solution to the strong-CP problem. It is so named after the eponymous American laundry detergent, using the axial anomaly to clean up the mess of CP symmetry in the strong interactions [4]. The QCD axion acquires mass from QCD chiral symmetry breaking, giving a one parameter model described by the axion decay constant, f_a . In quantum field theory, the term can apply generally to any pseudoscalar Goldstone bosons of spontaneously broken global chiral symmetries, typically giving a two parameter model with (m_a, f_a) . In string theory and supergravity,

the term "axion" is more general and can refer either to such matter fields, or to pseudoscalar fields associated to the geometry of compact spatial dimensions [5]. In these theories there are typically many axion fields, each with a number of free parameters in their potentials and kinetic terms. In this review, we will use the term in its most general sense for a light pseudoscalar field (indeed in some cosmological cases, apart from naturalness considerations, even the distinction between scalar and pseudoscalar will be irrelevant).

Since the QCD axion was first proposed in 1977-1978, there have been many reviews written on axion physics. Many such reviews and published lecture notes focus on the QCD axion and its role in solving the strong-*CP* problem [6, 7], as well as its important cosmological role [8]. Of ALPs, there are technical reviews of axions in field theory and string theory [9, 5], as well as reviews of axions in astrophysics [10], and of axion inflation [11]. There is also a vast number of reviews in the field of axion direct detection [12, 13, 14, 15, 16]. It is the purpose of this review firstly to focus on ULAs, the cosmology of which has not been reviewed before, and with a particular emphasis on methods of modern precision cosmology, including computational aspects both analytic and numerical, and with an eye to data. Secondly, it is to bring together the disparate topics of other axion reviews into one place, expressing the unity of axion particle physics and cosmology: a task, which, to my knowledge, has not been fully addressed since the review of Ref. [9], more than 30 years ago in this very journal.

Notes

Useful notation and equations for cosmology are defined in the Appendix. I (mostly) use units where $c = \hbar = k_B = 1$ and express everything in terms of either electronvolts, eV, solar masses, M_{\odot} , parsecs, pc, or Kelvin, K, depending on the context. The Fourier conjugate variable to x is k and my Fourier convention puts the 2π 's under the dk's. I use the reduced Planck mass, $M_{pl} = 1/\sqrt{8\pi G} = 2.435 \times 10^{27} \text{ eV}$, and a "mostly positive" metric signature.

ACCEPTED MANUSCRIPT





2. Models

A classic review of models for axions in particle physics and string theory is Ref. [9], where many more details are given. A modern review of axions in string theory is Ref. [5], and for pedagogical introductions and phenomenology see e.g. Refs. [17, 14]. This section is intended only as an overview: we will wave our hands through the particle physics computations, and wave them even more wildly through the string theory. This section is also self-contained, and can be skipped for those interested only in cosmology and astrophysics. The salient points for cosmology are repeated in Section 3.1.

2.1. The QCD Axion

2.1.1. The Strong-CP Problem and the PQ Solution

QCD suffers from the "strong-CP problem." A topological (total derivative) term is allowed in the Lagrangian:

$$\mathcal{L}_{\theta \text{QCD}} = \frac{\theta_{\text{QCD}}}{32\pi^2} \text{Tr } G_{\mu\nu} \tilde{G}^{\mu\nu} , \qquad (2)$$

where $G_{\mu\nu}$ is the gluon field strength tensor, $\tilde{G}^{\mu\nu} = \epsilon^{\mu\nu\alpha\beta}G_{\alpha\beta}/2$ is its dual, and the trace is over the adjoint representation of SU(3) (a notation I drop from now on).¹ This term arises due to the so-called " θ -vacua" of QCD [18], which are discussed in Appendix A.

The θ term is *CP* violating and gives rise to an electric dipole moment (EDM) for the neutron [19]:

$$d_n \approx 3.6 \times 10^{-16} \theta_{\rm QCD} \, e \, \, {\rm cm} \,, \tag{3}$$

where e is the charge on the electron. The (permanent, static) dipole moment is constrained to $|d_n| < 2.9 \times 10^{-26} e \text{ cm} (90\% \text{ C.L.})$ [20], implying $\theta_{\text{QCD}} \lesssim 10^{-10}$.

¹I have chosen the normalization for the gluon field, A_{μ} , appropriate for the vacuum topological term, which takes $\theta_{\rm QCD} \in [0, 2\pi]$. In this normalization the gluon kinetic term is $-G_{\mu\nu}G^{\mu\nu}/4g_3^2$, where g_3 is the SU(3) gauge coupling constant.

This is a true fine tuning problem, since θ_{QCD} could obtain an $\mathcal{O}(1)$ contribution from the observed *CP*-violation in the electroweak (EW) sector [21], which must be cancelled to high precision by the (unrelated) gluon term. Specifically, the measurable quantity is

$$\theta_{\rm QCD} = \tilde{\theta}_{\rm QCD} + \arg \, \det M_u M_d \,, \tag{4}$$

where $\tilde{\theta}$ is the bare quantity and M_u , M_d are the quark mass matrices.²

The QCD axion is the dynamical pseudoscalar field coupling to $G\tilde{G}$, proposed by Peccei and Quinnn (PQ) [3], which dynamically sets $\theta_{\rm QCD} = 0$ via QCD non-perturbative effects (instantons) [23]. The simple idea is that there is a field, ϕ , which enjoys a shift symmetry, with only derivatives of ϕ appearing in the action. Taking $\theta_{\rm QCD} = C\phi/f_a$, where ϕ is the canonically normalized axion field, f_a is the axion decay constant and C is the "colour anomaly" (discussed in Section 2.2), this is a symmetry under $\phi \to \phi + \text{const.}$ Then, as long as shift symmetry violation is induced only by quantum effects as $(C\phi/f_a)G\tilde{G}$, any contribution to $\theta_{\rm QCD}$ can be absorbed in a shift of ϕ . The action, and thus the potential induced by QCD non-perturbative effects, only depends on the overall field multiplying $G\tilde{G}$. If the potential for the shifted field is minimized at $C\phi/f_a = 0 \mod 2\pi$, then the strong CP problem is solved. In fact, a theorem of Vafa and Witten [23] guarantees that the instanton potential in more detail in Section 2.2.

The axion mass, m_a , induced by QCD instantons can be calculated in chiral perturbation theory [24, 2]. It is given by

$$m_{a,\text{QCD}} \approx 6 \times 10^{-6} \text{ eV}\left(\frac{10^{12} \text{ GeV}}{f_a/\mathcal{C}}\right).$$
 (5)

This is a (largely) model-independent statement, and the approximate symbol, " \approx ," takes model and QCD uncertainties into account. If f_a is large, the QCD

²The phase of the quark mass matrix is not measured, but could be $\mathcal{O}(1)$. *CP*-violation in the standard model leads to a calculable minimum value for θ_{QCD} even in the axion model (e.g. Ref. [22]).

axion can be extremely light and stable, and is thus an excellent DM candidate [25, 26, 27].

We will consider three general types of QCD axion model:³

- The Peccei-Quinn-Weinberg-Wilczek (PQWW) [3, 24, 2] axion, which introduces one additional complex scalar field only, tied to the EW Higgs sector. It is excluded by experiment.
- The Kim-Shifman-Vainshtein-Zakharov (KSVZ) [28, 29] axion, which introduces heavy quarks as well as the PQ scalar.
- The Dine-Fischler-Srednicki-Zhitnitsky (DFSZ) [30, 31] axion, which introduces an additional Higgs field as well as the PQ scalar.

2.1.2. PQWW axion

The PQWW model introduces a single additional complex scalar field, φ , to the standard model as a second Higgs doublet. One Higgs field gives mass to the *u*-type quarks, while the other gives mass to the *d*-type quarks (a freedom of the model is the choice of which doublet, if not a third field, gives mass to the leptons). This fixes the representation of φ in $SU(2) \times U(1)$. The whole Lagrangian is then taken to be invariant under a global $U(1)_{PQ}$ symmetry, which acts with *chiral* rotations, i.e. with a factor of γ_5 . These chiral rotations shift the angular part of φ by a constant. The PQ field couples to the standard model via the Yukawa interactions which give mass to the fermions as in the usual Higgs model. The invariance of these terms under global $U(1)_{PQ}$ rotations fixes the PQ charges of the fermions.

Just like the Higgs, φ has a symmetry breaking potential (see Fig. 2):

$$V(\varphi) = \lambda \left(|\varphi|^2 - \frac{f_a^2}{2} \right)^2 \,, \tag{6}$$

 $^{^{3}}$ One can also construct more general particle physics models along these lines with multiple ALPs as well as the QCD axion, but we will not discuss such models in detail. We consider all ALPs within a string theory context in Section 2.4.

ACCEPTED MANUSCRIPT



Figure 2: A symmetry breaking potential in the complex φ plane. The vev of the radial mode is $f_a/\sqrt{2}$ and the axion is the massless angular degree of freedom at the potential minimum.

and takes a vacuum expectation value (vev), $\langle \varphi \rangle = f_a/\sqrt{2}$ at the EW phase transition. Just as for the Higgs, this fixes the scale of the vev $f_a \approx 250$ GeV.

There are four real, electromagnetically (EM) neutral scalars left after EW symmetry breaking: one gives the Z-boson mass, one is the standard model Higgs [32, 33], one is the heavy radial φ field, and one is the angular φ field. The angular degree of freedom appears as $\langle \varphi \rangle e^{i\phi/f_a}$ after canonically normlaizing the kinetic term. The field ϕ is the axion and is the Goldstone boson of the spontaneously broken $U(1)_{\rm PQ}$ symmetry.

The axion couples to the standard model via the chiral rotations and the PQ charges of the standard model fermions, e.g. expanding in powers of $1/f_a$ the quark coupling is $m_q(\phi/f_a)i\bar{q}\gamma_5 q$. The chiral anomaly [34] then induces couplings to gauge bosons via fermion loops⁴ $\propto \phi G\tilde{G}/f_a$ and $\propto \phi F\tilde{F}/f_a$, where F is the EM field strength. The gluon term is the desired term and leads to the PQ solution of the strong-CP problem. Notice that all axion couplings come

⁴See Appendix B for a heuristic description of effective field theory (EFT).

suppressed by the scale f_a , which in the PQWW model is fixed to be the EW vev. In the PQWW model f_a is too small, the axion couplings are too large, and it is excluded, e.g. by beam-dump experiments [9]. The PQWW axion is also excluded by collider experiments such as LEP (see the recent compilation of collider constraints in Ref. [35], and Section 9.6).

In the KSVZ and DFSZ models, which we now turn to, the PQ field, φ , is introduced independently of the EW scale. The decay constant is thus a free parameter in these models, and can be made large enough such that they are not excluded. For this reason, both the KSVZ and the DFSZ axions are known as *invisible axions*. On the plus side, in these models the axion is stable and is an excellent DM candidate with its own phenomenology.

2.1.3. KSVZ axion

The KSVZ axion model introduces a heavy quark doublet, Q_L, Q_R , each of which is an SU(3) triplet, and the subscripts represent the charge under chiral rotations. The PQ scalar field, φ , has charge 2 under chiral rotations, but is now a standard model singlet. The PQ field and the heavy quarks interact via the PQ-invariant Yukawa term, which provides the heavy quark mass:

$$\mathcal{L}_{\rm Y} = -\lambda_Q \varphi \bar{Q}_L Q_R + h.c. \,, \tag{7}$$

where the Yukawa coupling λ_Q is a free parameter of the model. As in the PQWW model, there is a global $U(1)_{\rm PQ}$ symmetry which acts as a chiral rotation with angle $\alpha = \phi/f_a$, shifting the axion field. Global $U(1)_{\rm PQ}$ symmetry is spontaneously broken by the potential, Eq. 6.

At the classical level, the Lagrangian is unaffected by chiral rotations, and φ is not coupled to the standard model. However at the quantum level, chiral rotations on Q affect the $\tilde{G}G$ term via the chiral anomaly [34]:

$$\mathcal{L} \to \mathcal{L} + \frac{\alpha}{32\pi^2} G\tilde{G},$$
 (8)

where I have used that in the KSVZ model the colour anomaly is equal to unity (see Section 2.2).

At low energies, after PQ symmetry breaking, φ takes a vev and the Q fields obtain a large mass, $m_Q \sim \lambda_Q f_a$. The Q fields can then be integrated out. The chiral anomaly induces the axion coupling to $\tilde{G}G$ as a "memory" of the chiral rotation applied at high energy. At the level of EFT, the induced topological term is the only modification to the standard model Lagrangian: the KSVZ axion has no unsuppressed tree-level couplings to standard model matter fields.

There is an axion-photon coupling in this model that can be calculated via loops giving the EM anomaly. It's value depends on the electromagnetic charges assigned to the Q fields. The canonical choice is that they are uncharged and the axion-photon coupling is induced solely by the longitudinal mode of the Z-boson (see e.g. Ref. [36]). Other couplings can also be induced by loops and mixing, since Q must be charged under SU(3). Couplings will be listed and discussed further in Section 2.3.

2.1.4. DFSZ axion

The DFSZ axion couples to the standard model via the Higgs sector. It contains two Higgs doublets, H_u , H_d , like in the PQWW model, however the complex scalar, φ , which contains the axion as its angular degree of freedom, is introduced as a standard model singlet. Again, global $U(1)_{PQ}$ symmetry is imposed and spontaneously broken by the potential, Eq. (6).

The PQ and Higgs fields interact via the scalar potential:

$$V = \lambda_H \varphi^2 H_u H_d \,. \tag{9}$$

This term is PQ invariant for φ with $U(1)_{PQ}$ charge +1, and the Higgs fields each with charge -1. As in the KSVZ model, PQ rotations act by shifting the axion by $\phi/f_a \rightarrow \phi/f_a + \alpha$. When the PQ symmetry is broken and φ obtains a vev, the parameters in the Higgs potential, and the coupling constant, λ_H , must be chosen such that the Higgs fields remain light, consistent with the observed 125 GeV standard model Higgs [32, 33], and the EW vev, $v_{\rm EW} = \sqrt{\langle H_u \rangle^2 + \langle H_d \rangle^2}$.

The Higgs must also couple to all the standard model fermions, providing

ACCEPTED MANUSCRIPT

their mass through Yukawa terms as usual, e.g.

L

$$\mathcal{L}_Y \supset \lambda_u \bar{q}_L u_R H_u$$
.

(10)

In order for this to be PQ invariant the standard model fermions must be charged under $U(1)_{PQ}$. After EW symmetry breaking, H is replaced by its vev, inducing axial current couplings between the axion and standard model fermions from the chiral term in the fermion mass matrix: $m_u(\phi/f_a)i\bar{u}\gamma_5 u$. This axial current in turn induces the coupling between the axion and $G\tilde{G}$ via the colour anomaly. The difference between KSVZ and DFSZ is that for DFSZ this term is induced by light quark loops calculated at low energy, rather than via the integrating out of a heavy quark. In the DFSZ model all of the standard model quarks are charged under the PQ symmetry, giving rise to a larger colour anomaly, C = 6.

The same fermion loops induce the axion-photon coupling, $\phi F \tilde{F}$, which is computed via the electromagnetic anomaly. Freedom in this model appears through the lepton charges: we are free to choose whether it is H_u or H_d that gives mass to the electron via $H_{u,d} \bar{\ell}_L e_R$. The axion-photon coupling is the sum of quark and lepton loops, and the different lepton PQ charges give different values for the anomaly, and thus the coupling (see Section 2.3).

The use of the Higgs in DFSZ leads to a number of important consequences that differentiate it from KSVZ. Firstly, in the DFSZ model there are tree-level couplings between the axion and standard model fermions, via the chiral terms in the mass matrix. Secondly, the EW sector is modified by the addition of an extra axial Higgs field, A, with mass of order the EW scale. This is constrained by collider data, and could potentially be discovered at the LHC, just like the additional Higgs fields of supersymmetry (SUSY, see e.g. Refs. [37, 38]).

2.2. Anomalies, Instantons, and the Axion Potential

A PQ rotation on a field x_i with PQ charge $Q_{PQ,i}$ acts as

$$x_i \to e^{i\mathcal{Q}_{\mathrm{PQ},i}\phi/f_a} x_i \,. \tag{11}$$

The rotation is chiral, meaning that, if x_i is a spinor, left and right handed components of x_i have opposite charges (for the two-component spinor $\psi = (\psi_L, \psi_R)$ one introduces a factor of γ_5 to achieve this).

The axion model is set up so that at the classical level the Lagrangian is invariant under such transformations, which leads to the shift symmetry of the axion field, $\phi \rightarrow \phi + \text{const.}$ At the quantum level, however, PQ rotations of quarks are anomalous, meaning that the quantum theory violates the classical symmetry. This affects the QCD topological term, and shifts it by an amount $\propto (\phi/f_a)G\tilde{G}$. The question we now wish to answer is: what is the constant of proportionality?

The constant of proportionality is called the *colour anomaly* of the PQ symmetry, and is given by (e.g. Ref. [39]):

$$\mathcal{C}\delta_{ab} = 2\mathrm{Tr} \ \mathcal{Q}_{\mathrm{PQ}}T_a T_b \,, \tag{12}$$

where the trace is over all the fermions in the theory, and T_a are the generators of the SU(3) representations of the fermions (e.g. for the triplet these are the Gell-Mann matrices). A PQ rotation now shows up in the action as

$$S \to S + \int d^4x \frac{\mathcal{C}}{32\pi^2} \frac{\phi}{f_a} \operatorname{Tr} G_{\mu\nu} \tilde{G}^{\mu\nu} \,. \tag{13}$$

Although the topological term in the QCD action, Eq. (2), does not affect the classical equations of motion, it does affect the vacuum structure, and the vacuum energy depends on $\theta_{\rm QCD}$. This is because of the existence of *in*stantons and the so-called θ -vacua of QCD (for more details, see Ref. [18] and Appendix A). These emerge because the non-Abelian gauge group, SU(3), can be mapped onto the symmetry group of the space-time boundary, allowing for topologically-distinct field configurations [18]. The different vacua of QCD are labelled by the value of $\theta_{\rm QCD}$. The vacuum energy is [40, 41]:

$$E_{\rm vac} \propto \cos \theta_{\rm QCD} \sim \theta_{\rm QCD}^2$$
 (14)

However, because the θ -vacua are topologically distinct, no process allows for

transitions between them, and the energy cannot be minimized.⁵ Introducing a field that couples to $G\tilde{G}$, as the axion does, means that the vacuum energy now depends on the linear combination $E_{\text{vac}}(\theta_{\text{QCD}} + C\phi/f_a)$.

Using the shift symmetry on ϕ to absorb any contribution to $\theta_{\rm QCD}$, the vacuum energy is

$$E_{\rm vac} \propto \cos\left(\frac{\mathcal{C}\phi}{f_a}\right)$$
 (15)

The vacuum energy now depends on a dynamical field, and so can be minimized by the equations of motion.

The colour anomaly sets the number of vacua that ϕ has in the range $[0, 2\pi f_a]$. Because ϕ is an angular variable, we must have a symmetry under $\phi \rightarrow \phi + 2\pi f_a$. This implies that the colour anomaly must be an integer (this can always be achieved by normalization [39]). Because it sets the number of vacua, the colour anomaly is also known as the *domain wall number*, $C = N_{\text{DW}}$ (see Section 3.3.2). Dynamics of ϕ send it to one of these vacua, which is the essence of the PQ mechanism.

In this way, the instantons are said to induce a mass for the axion. Let's investigate this in the DFSZ model, though the argument is more general. The relevant terms in the Lagrangian are:

$$m_q \bar{q}q + \frac{N_{\rm DW}\phi}{32\pi^2 f_a} G\tilde{G} \,. \tag{16}$$

Applying a chiral rotation to the quarks by an angle $\alpha = N_{\rm DW} \phi / f_a$ shows up as an interaction between the axion and the quarks:

$$\cos(N_{\rm DW}\phi/f_a)m_*(\bar{u}u+\bar{d}d)+\sin(N_{\rm DW}\phi/f_a)m_*(\bar{u}i\gamma_5u+\bar{d}i\gamma_5d)\,,\qquad(17)$$

where $m_* = m_u m_d / (m_u + m_d)$.

After the QCD confinement transition at $T \sim \Lambda_{\text{QCD}}$ we can replace the quark bilinears with their condensates, $\langle q\bar{q} \rangle$. Expanding for large f_a we see that the cosine term introduces a mass (i.e. ϕ^2 term) for the axion proportional to

⁵There is a "superselection rule" such that $\langle \theta | \text{Anything} | \theta' \rangle = \delta_{\theta \theta'}$.

 $-(m_u + m_d)\langle q\bar{q}\rangle/f_a^2 = m_\pi^2 f_\pi^2/f_a^2$, where m_π is the pion mass and f_π is the pion decay constant.

At lowest order the sine term introduces a Yukawa-like interaction between axions and quarks, and renormalizes the axion mass. The interaction allows for the quark condensate to appear in the axion two-point function. The structure of the interaction is such that the η' meson dominates this effect and the axion mass is renormalized to

$$m_a^2 = \frac{m_\pi^2 f_\pi^2}{(f_a/N_{\rm DW})^2} \frac{m_u m_d}{(m_u + m_d)^2} \left\{ 1 + \frac{m_\pi^2}{m_\eta^2} \left[-1 + \mathcal{O}\left(1 - \frac{m_\pi}{m_\eta}\right) \right] \right\}.$$
 (18)

The masses of the mesons are known [42], and the η' is substantially heavier than the π . If the masses were the same, the quantum effects would cancel, and the axion would be massless. QCD non-perturbative effects are responsible for lifting the η' above the π . Any non-perturbative physics will do the job, but it happens that the lifting is due to the same instantons that are responsible for the θ -vacua. This is why we say that QCD instantons give mass to the axion for $T < \Lambda_{\rm QCD}$. The non-perturbative effects break the axion shift symmetry down to the discrete shift symmetry, $\phi \to \phi + 2\pi f_a/N_{\rm DW}$, and the axion is a pseudo Nambu-Goldstone boson (pNGB).

The axion potential generated by QCD instantons is

$$V(\phi) = m_u \Lambda_{\rm QCD}^3 \left[1 - \cos\left(\frac{N_{\rm DW}\phi}{f_a}\right) \right] \,. \tag{19}$$

The cosine form comes from the dependence of the vacuum energy on $\theta_{\rm QCD}$ in the lowest order instanton calculation [40], and I have applied a constant shift such that V is minimized at zero, i.e. I have assumed a solution to the cosmological constant problem. The instanton potential given here is the zero temperature potential: we will discuss temperature dependence in Section 4.3.2, as it is important when computing the axion relic abundance.

QCD is not the only non-abelian gauge theory in the standard model, there is also SU(2) in the EW sector, and SU(2) instantons also contribute to the axion potential. The weak force breaks CP, and the SU(2) instantons lead to a shift in the minimum of the axion potential away from the CP-conserving value.

The instanton action for a gauge group with coupling g_i is (this is typical of nonperturbative effects, and can be seen e.g. via dimensional transmutation [40])

$$S_{\text{inst.}} = \frac{8\pi^2}{g_i^2} \,. \tag{20}$$

This action sets the co-efficient in front the axion potential from a given sector as $V_i(\theta) \propto \cos \theta e^{-S_{\text{inst.}}(g_i)}$. Taking $g = g_{EW} \ll g_3$ we see that the potential from W-bosons only weakly breaks CP compared to the QCD term. For more details, see Ref. [9].

We have so far discussed instantons and non-perturbative physics in the standard model, but the story can be extended to encompass general pNGBs, including ALPs. The steps are:

- There is a global U(1) symmetry respected by the classical action.
- Spontaneous breaking at scale f_a leads to an angular degree of freedom, ϕ/f_a , with a shift symmetry.
- The U(1) symmetry is anomalous and explicit breaking is generated by quantum effects (instantons etc.), which emerge with some particular scale, Λ_a . Because of the classical shift symmetry, these effects must be non-perturbative.
- Since ϕ is an angular degree of freedom, the quantum effects must respect the residual shift symmetry $\phi \rightarrow \phi + 2n\pi f_a$.

In this picture a pNGB or ALP obtains a periodic potential $U(\phi/f_a)$ when the non-perturbative quantum effects "switch on." The mass induced by these effects is $m_a \sim \Lambda_a^2/f_a$.

2.3. Couplings to the Standard Model

The couplings of the QCD axion are computed in Ref. [39]. Other references include Refs. [9, 36, 43].

The QCD axion is defined to have coupling strength unity to $G\tilde{G}$, via the term in Eq. (2), replacing $\theta_{\rm QCD} \rightarrow \phi/(f_a/N_{\rm DW})$. Any ALP must couple more

weakly to QCD (e.g. Ref [44]), and in any case a field redefinition can often define the QCD axion to be the linear combination that couples to QCD, leaving ALPs free of the QCD anomaly.

Axion couplings to the rest of the standard model are defined by symmetry, and in specific models can be computed in EFT. The axion is a pseudoscalar Goldstone boson with a shift symmetry, so all couplings to fermions must be of the form

$$\partial_{\mu}(\phi/f_a)(\bar{\psi}\gamma^{\mu}\gamma_5\psi).$$
(21)

The form of this coupling, as an axial current, means that the force mediated by axions is *spin-dependent* and only acts between spin-polarised sources (see Section 9.4). Thus no matter how light the axion, it transmits no long-range scalar forces between macroscopic bodies. This has the important implication that, in an astrophysical setting, ULAs are not subject to the simplest fifth-force constraints like light scalars such as (non-axion) quintessence are.

For example, in the DFSZ model, a coupling of the form Eq. (21) is obtained from the $H\bar{\psi}\psi$ term after symmetry breaking and a PQ rotation, with the value of the co-efficient set by the PQ charge of the fermions. Such a term is generated at one loop in the KSVZ model.

A coupling to EM of the form:

$$\phi \vec{E} \cdot \vec{B} = -\phi F_{\mu\nu} \tilde{F}^{\mu\nu} / 4 \tag{22}$$

is generated if there is an EM anomaly (see below).

On symmetry grounds we can write a general interaction Lagrangian, applicable at low energies (after PQ symmetry breaking and non-perturbative effects have switched on):

$$\mathcal{L}_{\text{int}} = -\frac{g_{\phi\gamma}}{4}\phi F_{\mu\nu}\tilde{F}^{\mu\nu} + \frac{g_{\phi N}}{2m_N}\partial_\mu\phi(\bar{N}\gamma^\mu\gamma_5N) + \frac{g_{\phi e}}{2m_e}\partial_\mu\phi(\bar{e}\gamma^\mu\gamma_5e) - \frac{i}{2}g_d\phi\bar{N}\sigma_{\mu\nu}\gamma_5NF^{\mu\nu}$$
(23)

where $\sigma^{\mu\nu} = \frac{i}{2} [\gamma^{\mu}, \gamma^{\nu}]$, and here N is a nucleon (proton or neutron). The coupling $g_{\phi\gamma}$ has mass-dimension -1 and is proportional to $1/f_a$; the coupling g_d has mass dimension -2 and is also proportional to $1/f_a$. The couplings

 $g_{\phi e}$ and $g_{\phi N}$ are dimensionless in the above conventions, but are related to commonly-used dimensionful couplings $\tilde{g}_{\phi e,N} = g_{\phi e,N}/(2m_{e,N}) \propto 1/f_a$. Notice how all dimensionful couplings are suppressed by $1/f_a$, which is a large energy scale. This is why axions are weakly coupled, and evade detection. Note the similarity to the suppression of quantum-gravitational effects by $1/M_{pl}$.

In generic ALP models the couplings to the standard model are taken as free parameters that and can be very much less than they are in the QCD case if, e.g., they are loop suppressed, or forbidden on symmetry grounds. In specific models, the couplings of ALPs can be computed (e.g. Refs. [45, 46]).

Expressions for all standard model couplings of the QCD axion can be found in, e.g. Ref. [43] (though the notation differs slightly). The EDM coupling, g_d , is discussed in Ref. [47]. In this section, we will only discuss the two-photon coupling in detail, following Ref. [36]. We define:

$$g_{\phi\gamma} = \frac{\alpha_{\rm EM}}{2\pi (f_a/\mathcal{C})} c_{\phi\gamma} \,, \tag{24}$$

where $\alpha_{\rm EM} \approx 1/137$ is the EM coupling constant and $c_{\phi\gamma}$ is dimensionless. The dimensionless coupling obtains contributions from above the chiral symmetry breaking scale, via the EM anomaly, and below the chiral-symmetry breaking scale, by mixing with the longitudinal component of the Z-boson [39]:

$$c_{\phi\gamma} = \frac{\mathcal{E}}{\mathcal{C}} - \frac{2}{3} \cdot \frac{4 + m_u/m_d}{1 + m_u/m_d}, \qquad (25)$$

where \mathcal{E} is the EM anomaly:

$$\mathcal{E} = 2 \mathrm{Tr} \ \mathcal{Q}_{\mathrm{PQ}} \mathcal{Q}_{\mathrm{EM}}^2 \,, \tag{26}$$

and $Q_{\rm EM}$ are the EM charges

We see clearly here how the KSVZ and DFSZ models differ. In KSVZ we only have the heavy Q fields with PQ charge, and so the value of $c_{\phi\gamma}$ is fixed by the EM charge assigned to this field. Model dependence in KSVZ occurs if we introduce additional heavy quarks with PQ and EM charges. In the DFSZ model, all the standard model fermions carry PQ charges. Model dependence in DFSZ occurs because the coupling depends on the lepton PQ charges, i.e. whether H_u or H_d gives mass to the leptons. If H_u gives mass to the leptons, $c_{\phi\gamma}$ also depends on the ratio of Higgs vevs, $\tan\beta = \langle H_u \rangle / \langle H_d \rangle$.

The QCD axion has certain canonical choices for the model dependence. For KSVZ one takes a single EM neutral Q field. For DFSZ the H_d gives mass to the leptons, allowing for SU(5) unification. For $m_u/m_d = 0.6$, the couplings are then:

$$c_{\phi\gamma} = -1.92 \text{ (KSVZ)}; \quad c_{\phi\gamma} = 0.75 \text{ (DFSZ)}.$$
 (27)

2.4. Axions in String Theory

As is well known, string theory requires the existence of more spacetime dimensions than our usual four: 10 in the case of the critical superstring, and 11 in the case of M-theory [48, 49, 50]. The additional spacetime dimensions must be "compactified," that is, rolled up and made compact, with a small size. Typically, for appropriate phenomenology containing some unbroken SUSY and chiral matter, the compact manifold must be "Calabi-Yau" [51]. The supergravity description of string theory contains antisymmetric tensor fields: for example, the antisymmetric partner of the metric, B_{MN} , is present in all string theories.

Axions arise as the Kaluza-Klein (KK) zero modes of the antisymmetric tensors on the Calabi-Yau [52]. The number of axions present depends on the topology of the compact manifold, and in particular is determined by its Hodge numbers. Many Calabi-Yau manifolds are known to exist, and the distribution peaks for Hodge numbers in the dozens [53], as shown in Fig. 3 for the Kreuzer-Skarke [54] list. Furthermore, axions arising in this way are massless to all orders in perturbation theory thanks to the higher-dimensional gauge invariance. The axions then obtain mass by non-perturbative effects, such as instantons. Thus axions, with symmetry properties similar to those axions in field theory that we have already discussed, are an extremely generic prediction of string theory, in the low-energy four-dimensional limit [5]. This scenario has come to be known



Figure 3: The distribution of Hodge numbers $h^{1,1}$ and $h^{1,2}$ for the known Calabi-Yau manifolds in the Kreuzer-Skarke [54] list. Note that the frequency (=number of occurrences) color scale is logarithmic. There is a huge peak in the distribution at $h^{1,1} \approx h^{1,2} \approx 30$, which implies that a compactification picked at random from this list is most likely to contain of the order of 30 axions.

as the string axiverse [17].⁶

Let's flesh out the discussion above with some simple examples and observations. I will use notation for forms, which can be found in e.g. Ref. [55].

A (p+1)-form field strength F_{p+1} appears in the action as:

$$S \supset -\frac{1}{2} \int F_{p+1} \wedge \star F_{p+1} = -\frac{1}{2(p+1)!} \int d^D x \sqrt{-g_D} F_{\mu_1 \cdots \mu_{p+1}} F^{\mu_1 \cdots \mu_{p+1}} , \quad (28)$$

where D is the number of spacetime dimensions, and g_D is the D-dimensional metric determinant. The equation of motion is dF = 0, implying F_{p+1} can be written as $F_{p+1} = dA_p$, since $d^2 = 0$ (this is just like the EM field strength and the usual vector potential). A general solution which is homogeneous and isotropic in the large dimensions is found by decomposing the potential A into

⁶Of course, there are many subtleties, and not all the axions present in the spectrum may survive to low energies. I defer to the references for discussion of this topic.

ACCEPTED MANUSCRIPT

the basis of harmonic *p*-forms, $\omega_{p,i}$, on the compact manifold:

$$A_p = \frac{1}{2\pi} \sum a_i(x) \omega_{p,i}(y) \Rightarrow a_i = \int_{C_{p,i}} A_p, \qquad (29)$$

where $C_{p,i}$ are *p*-cycles in the compact space, *x* are co-ordinates in the large 3 + 1 dimensions, *y* are co-ordinates in the compact space, and for symmetry under *CP*, $a_i(x)$ is a pseudoscalar.

The sum in Eq. (29) runs over the number of harmonic forms, and expresses the topologically distinct ways that F can be "wrapped" on the compact space. The number of basis p-forms is determined by the number of homologically nonequivalent p-cycles, i.e. by the p^{th} Betty number, b_p . For example, taking the decomposition Eq. (29) for the two-form B mentioned above, we would count the number of two-cycles, and for the C_4 four-form of Type IIB theory, we would count the number of four-cycles.⁷ For a Calabi-Yau three-fold (three complex dimensions, six real dimensions), all the b_p are determined by the two Hodge numbers $h^{1,1}$ and $h^{1,2}$ (see, e.g., Chapter 9 of Ref. [50], and Fig. 3 above).

The axions of Eq. (29) are closed string axions. Each closed string axion is partnered into a complex field $z_i = \sigma_i + ia_i$ where σ_i is a scalar modulus (saxion) field controlling the size of the corresponding *p*-cycle. The moduli come from KK reduction of the Ricci scalar as usual, and their pairing with axions is a consequence of SUSY, which demands the existence of the appropriate form fields in supergravity. Open string axions also exist in string theory, and are more like the field theory axions we discussed previously. Open string axions live on spacetime filling branes supporting gauge theories and are the phases of

⁷Take a simple example in non-string theory jargon. Imagine a vector field, A_{μ} with field strength $F_{\mu\nu}$ in 3+1 large dimensions, and a two dimensional compact space in the shape of a doughnut (or two-torus). There are two distinct ways the vector field can wrap the doughnut: along the tube, or all the way around. These are the distinct one-cycles of the torus. The vector field has co-ordinates in the large dimensions also, but if these are to be homogeneous and isotropic, the only dependence can be as a (pseudo)scalar expressing how wrapping varies from place to place. The two fields necessary are the axions: the KK zero-modes of the A field wrapped on the one-cycles.

matter fields, φ , which break global PQ symmetries. Open string axions might be related to closed string axions by gauge/gravity duality [56, 57].

We have just seen the basics of how string theory gives rise to axions and moduli, the number of which is determined by the topology of the compact space. Next we must ask what determines the spectrum of axion masses and decay constants.

After KK reduction of Eq. (28) the $a_i(x)$ fields are found to be massless, i.e. there are only kinetic terms for them in the action, implying a shift symmetry. The shift symmetry descends from the higher-dimensional gauge invariance of F, and so is protected to all orders in perturbation theory.

In Type IIB theory, the axion kinetic term resulting from KK reduction of the action for the C_4 four-form potential is (for the full axion action in Type IIB theory, see e.g. Ref. [14])

$$S \supset -\frac{1}{8} \int \mathrm{d}a_i \mathcal{K}_{ij} \wedge \star \mathrm{d}a_j \,, \tag{30}$$

where \mathcal{K}_{ij} is the Kähler metric,

$$\mathcal{K}_{ij} = \frac{\partial^2 K}{\partial \sigma_i \partial \sigma_j} \,, \tag{31}$$

and K is the Kähler potential, which depends on the moduli. KK reduction kinetically mixes the axions and couples them to the moduli via the Kähler metric. Canonically normalizing the kinetic terms and diagonalizing the Kähler metric, we see that it is the moduli that determine the axion decay constants, since the canonical kinetic term is $\mathcal{L}_{\text{kin.}} = -f_{a,i}^2 (\partial a_i)^2/2$. In particular we have that, parametrically,⁸

$$f_{a,i} \sim \frac{M_{pl}}{\sigma_i} \lesssim M_{pl} \,, \tag{32}$$

where the dimensionless modulus σ_i measures the volume of the corresponding *p*-cycle in string units, i.e. $\sigma_i = \text{Vol}_i/l_s^p$, for string length l_s . The volume should be larger than the string scale in order for the effective field theory description

 $^{^{8}}$ I have assumed that the size of the cycle is of order the size of the manifold. See Refs. [5, 58] for more details.

to be valid, giving the inequality. This may be related to be a general feature, known as the "weak gravity conjecture," following from properties of black holes [59].⁹ We return to this question in the context of inflation in Section 7.2.

Axions in string theory can obtain potentials from a variety of non-perturbative effects (see e.g. Refs. [5, 17, 58, 63]). In general, instantons provide a contribution to the superpotential, W for the axion field $a = \phi/f_a$:

$$W = M^3 e^{-S_{\text{inst.}} + ia}, \qquad (33)$$

where $S_{\text{inst.}}$ is the instanton action and M is the scale of instanton physics, which in string theory may be the Planck scale. If SUSY is broken at a scale m_{SUSY} then the axion potential at low energies is

$$V(\phi) = \Lambda_a^4 [1 - \cos(\phi/f_a)] \text{ with } \Lambda_a^4 = m_{\text{SUSY}}^2 M_{pl}^2 e^{-S_{\text{inst.}}} .$$
(34)

A non-Abelian gauge group has instantons with action given by Eq. (20). In string theory, the moduli couple to the gauge kinetic term for a non-Abelian group realized by a stack of *D*-branes wrapping the corresponding cycle, and the gauge coupling $g^2 \propto 1/\sigma$ (this occurs e.g. in Type IIB theory for gauge theory on a stack of *D*7 branes filling 3+1 spacetime and wrapped on the same four-cycles as C_4). Thus, if an axion obtains mass from these instantons as above, we find that the axion mass scales exponentially with the cycle volumes:

$$m_a^2 \sim \frac{\mu^4}{f_a^2} e^{-\#\sigma_i} ,$$
 (35)

where μ is a hard scale. In general, from the above, we expect $\mu = \sqrt{m_{\text{SUSY}} M_{pl}}$. If the moduli are stabilised by perturbative SUSY breaking effects giving $m_{\sigma} \sim m_{\text{SUSY}} \gg m_a$ then the moduli can be set to constant values at late times in cosmology and the axion mass will be a constant (for dynamical moduli as dark energy, see Refs. [64, 65]).

⁹The relation of the conjecture to axion decay constants is only well formulated in the case of a single axion. Consider, for example, the two axion model of Ref. [60] has a decay constant $\sim 3.25M_{pl}$. Our simplistic description here has ignored the phenomenon of alignment [61, 62].

The two observations, Eqs. (32,35), form the key basis for the phenomenology of the axiverse. Thanks to the exponential scaling of the potential energy scale with respect to the moduli, string axions will have masses spanning many orders of magnitude. The axion decay constants will (generally) be parametrically smaller than the Planck scale, and are expected to span only a small range of scales due to the power-law scaling with the moduli.

Let's end this discussion with a few examples of explicit string theory constructions displaying the above properties. The so-called "model independent axion" in heterotic string theory emerges from compactification of B_{MN} on twocycles. It has decay constant $f_a = \alpha_{\rm GUT} M_{pl}/2\sqrt{2}\pi$ and the shift symmetry of the axions is broken by wrapped NS-5 branes with $S_{\rm inst.} = 2\pi/\alpha_{\rm GUT}$ [5]. Gauge coupling unification at $\alpha_{\rm GUT} = 1/25$ gives $f_a \sim 1.1 \times 10^{16}$ GeV.

The *M*-theory axiverse [66] is realized as a compactification of M-theory on a G_2 manifold, with axions arising from the number of three-cycles. The G_2 volume is small, fixing one heavy string-scale axion by leading non-perturbative effects, and giving $f_a \approx 10^{16}$ GeV. The remaining axions obtain potentials from higher order effects, and are hierarchically lighter. Fixing the GUT coupling requires that an additional axion take a mass $m_{a,GUT} \approx 10^{-15}$ eV. The other axions in the theory will be distributed around these characteristic values according to the scalings we have discussed.

The Type IIB axiverse [67] is a LARGE volume Calabi-Yau compactification [68, 69], with axions arising from C_4 as discussed above. At least two axions are required in this scenario, one of which is the almost-massless volume-axion associated to the exponentially large volume-modulus, and the other is again associated to the GUT coupling. The volume, \mathcal{V} , is exponentially large in string units and gives the decay constant of the volume-axion as $f_a \approx 10^{10}$ GeV. Other light axions are associated to perturbatively fixed moduli, since they must obtain masses only from higher order effects. Larger values of the effective decay constant for very light axions with $m_a \sim H_0$ can be achieved in this scenario by alignment [70].

3. Production and Initial Conditions

3.1. Symmetry Breaking and Non-Perturbative Physics

Let's briefly review the general picture for axions given in the previous section, highlighting how this is relevant to axion cosmology in the very early Universe. Two important physical processes determine this behaviour. Symmetry breaking occurs at some high scale, f_a , and establishes the axion as a Goldstone boson. Next, non-perturbative physics becomes relevant, at some temperature $T_{\rm NP} \ll f_a$, and provides a potential for the axion.

Giving substance to this chain of events: the axion field, ϕ , is related to the angular degree of freedom of a complex scalar, $\varphi = \chi e^{i\phi/f_a}$. The radial field, χ , obtains the vev $\langle \chi \rangle = f_a/\sqrt{2}$ when a global U(1) symmetry is broken (see Fig. 2). The field χ is heavy, and f_a is the PQ symmetry breaking scale. The axion is the Goldstone boson of this broken symmetry , and possesses a shift symmetry, $\phi \to \phi + \text{const.}$, making it massless to all orders in perturbation theory. Non-perturbative effects, for example instantons, "switch on" at some particular energy scale and break this shift symmetry, inducing a potential for the axion, $V(\phi)$. The potential must, however, respect the residual discrete shift symmetry, $\phi \to \phi + 2n\pi f_a/N_{\text{DW}}$, for some integer n, which remains because the axion is still the angular degree of freedom of a complex field. The potential is therefore *periodic*.

The scale of non-perturbative physics is Λ_a and the potential can be written as $V(\phi) = \Lambda_a^4 U(\phi/f_a)$, where U(x) is periodic, and therefore possesses at least one minimum and one maximum on the interval $x \in [-\pi, \pi]$. We can choose the origin in field space such that U(x) has its minimum at x = 0.¹⁰ It is common practice to assume a solution to the cosmological constant problem such that the minimum is also obtained at U(0) = 0 (see Section 7.1 for further discussion).

¹⁰When $x \neq 0$ is associated to the breaking of *CP* symmetry, as is the case for the QCD axion, a theorem of Vafa and Witten [23] guarantees that the induced potential has a minimum at the *CP*-conserving value x = 0.

A particularly simple choice for the potential is then

$$V(\phi) = \Lambda_a^4 \left[1 - \cos\left(\frac{N_{\rm DW}\phi}{f_a}\right) \right], \qquad (36)$$

where $N_{\rm DW}$ is an integer, which unless otherwise stated I will set equal to unity. I stress that the potential Eq. (36) is not unique and without detailed knowledge of the non-perturbative physics it cannot be predicted. For example, so-called "higher order instanton corrections" might appear, as $\cos^n \phi/f_a$ (see e.g. Ref. [71]). The form of the potential given by Eq. (36) is, however, a useful benchmark for considering the form of axion self-interactions.

We can study axions in a model-independent way if we consider only small, $\phi < f_a$, displacements from the potential minimum. In this case, the potential can be expanded as a Taylor series. The dominant term is the mass term:

$$V(\phi) \approx \frac{1}{2} m_a^2 \phi^2 \,, \tag{37}$$

where $m_a^2 = \Lambda_a^4/f_a^2$. The symmetry breaking scale is typically rather high, while the non-perturbative scale is lower. The axion mass is thus parametrically small.

Let's consider some possible values for these scales. The QCD axion (see Section 2.1) is the canonical example, where we have that $\Lambda_a^4 \approx \Lambda_{\rm QCD}^3 m_u$ with $\Lambda_{\rm QCD} \approx 200$ MeV and m_u the u-quark mass, and 10⁹ Gev $\lesssim f_a \lesssim 10^{17}$ GeV. The lower limit on f_a comes from supernova cooling [72, 73] (see Section 9.1), while the upper limit comes from black hole superradiance [74] (BHSR, see Section 8.1). This leads to an axion mass in the range 4×10^{-10} eV $\lesssim m_{a,\rm QCD} \lesssim$ 4×10^{-2} eV.

In string theory models (see Section 2.4), things are much more uncertain. The decay constant typically takes values near the GUT scale, $f_a \sim 10^{16}$ GeV [5], though lower values of $f_a \sim 10^{10-12}$ GeV are possible [67]. In specific, controlled, examples one always finds $f_a \leq M_{pl}$ for individual axion fields. The "weak gravity conjecture" places some constraints on realising super-Planckian decay constants within quantum gravity [59].¹¹ The potential energy

¹¹Collective behaviour of multiple axion fields further complicates matters. We will return

scale in string models depends exponentially on details of the compactification, and large hierarchies between the non-perturbative scale and the string scale can easily be achieved. Explicitly, $\Lambda_a \sim \mu e^{-\sigma}$, where μ is the hard non-perturbative scale (e.g. SUSY breaking), and σ is a modulus field describing the size of the compact dimensions in string units: small changes in σ produce large changes in Λ_a for fixed μ . String models are expected to possess a large number of axions, with each axion associated to a different modulus. String axions thus have a mass spectrum spanning a vast number of orders of magnitude from the string scale down to zero. In particular, string models can realise a spectrum such as Eq. (1).

The axion mass is protected from quantum corrections, since these all break the underlying shift symmetry and must come suppressed by powers of f_a . For the same reason, self-interactions and interactions with standard model fields are also suppressed by powers of f_a (for the self-interactions, we can see this easily by expanding the cosine potential to higher orders). This renders the axion a light, weakly interacting, long-lived particle. These properties are protected by a symmetry and as such the axion provides a *natural* candidate to address cosmological problems that can be solved using a light scalar field. Axions can be used to drive inflation, to provide DM, and to provide DE.

Taking only the mass term from the potential for simplicity, the homogeneous component of the axion field obeys the equation

$$\ddot{\phi} + 3H\dot{\phi} + m_a^2\phi = 0.$$
(38)

This is the equation of a simple harmonic oscillator with time dependent friction determined by the Friedmann equations, Eqs. (B2). In general, the axion mass will be temperature dependent, as the non-perturbative effects switch on. We will study this equation in detail in Section 4. An important stage in the evolution of the axion field is the transition form over-damped to under-damped

to this topic in Section 7.2. A large literature surrounds the question of super-Planckian axions in string theory, see e.g. Refs. [75, 71, 76, 77, 78, 79, 80], and references therein.

motion, which occurs when $H \sim m_a$, and the axion field begins oscillating.

3.2. The Axion Field During Inflation

This section refers explicitly to DM axions as a spectator fields during inflation.¹² Inflation driven by an axion field is discussed in Sec. 7.2.

The temperature of the Universe during inflation is given by the Gibbons-Hawking [81] temperature (Hawking radiation emitted from the de-Sitter horizon):

$$T_I = \frac{H_I}{2\pi} \,, \tag{39}$$

where H_I is the inflationary Hubble scale. This temperature determines whether the PQ symmetry is broken or unbroken during inflation, with each scenario giving rise to a different cosmology.

The inflationary Hubble scale is tied to the value of the tensor-to-scalar ratio, r_T :

$$\frac{H_I}{2\pi} = M_{pl}\sqrt{A_s r_T/8}\,. \tag{40}$$

where A_s is the scalar amplitude. Ever since the observation of the first acoustic peak in the CMB [82, 83, 84], we have known that $r_T < 1$ and that cosmological fluctuations are dominantly scalar and adiabatic, with $\sqrt{A_s} \sim 10^{-5}$ first measured by COBE [85]. This sets, very roughly, $H_I \leq 10^{14}$ GeV. The most up-todate constraints come from the combined analysis of *Planck* and BICEP2 [86], which give $A_s = 2.20 \times 10^{-9}$, $r_T < 0.12$ and thus

$$\frac{H_I}{2\pi} < 1.4 \times 10^{13} \text{ GeV}.$$
(41)

¹²I assume a standard, single-field, slow-roll inflationary model throughout these notes, as it gives us a concrete setting for performing calculations and comparing to data. I further assume (for the most part) that the Universe is radiation dominated from the end of inflation, and in particular when $V(\phi)$ switches on. The general principles, however, can be used as a guide for computing in non-standard cosmologies. The important aspects to consider are: when does symmetry breaking occur with respect to the epoch when initial conditions are set; what is the energy scale at which initial conditions are set; what dominates the energy density when the non-perturbative physics giving rise to $V(\phi)$ becomes relevant?

High scale single-field slow-roll inflation has observably large tensor modes, $r_T \gtrsim 10^{-3}$, and requires super-Planckian motion of the inflaton [87]. We will discuss the importance of CMB tensor modes to axion phenomenology in more detail in Section 5.4.

3.2.1. PQ symmetry unbroken during inflation, $f_a < H_I/2\pi$

This scenario occurs when $f_a < H_I/2\pi$. A large misalignment population of ULA DM (our main focus in these notes) requires $f_a \sim 10^{16}$ GeV, and so this scenario is irrelevant to that model. This is an important scenario for the QCD axion, however, since it applies to the ADMX [88] sensitivity range of $f_a \sim 10^{12}$ GeV in the case of high scale standard inflation.

During inflation, fluctuations induced by the Gibbons-Hawking temperature are large enough that the U(1) symmetry is unbroken and φ has zero vev. After inflation, the symmetry breaks when the radiation temperature drops below f_a . At this point, χ obtains a vev and each causally disconnected patch picks a different value for $\phi/f_a = \theta_{PQ}$. Since the decay constant is larger than the scale of non-perturbative physics, the axion has no potential at this time, and θ_{PQ} thus has no preferred value. Therefore, in each Hubble patch θ_{PQ} is drawn at random from a uniform distribution on $[-\pi, \pi]$. The horizon size $R \sim 1/H$ when the PQ symmetry is broken. The symmetry is broken in the early Universe, and the present day Universe is made up of many patches that had different initial values of θ_{PQ} .

Given the θ_{PQ} distribution, it is possible to compute the average value of the square of the axion field, $\langle \phi^2 \rangle$. As we will see later, this value fixes the axion relic density produced by vacuum realignment in this scenario (see Sections 3.3 and 4.3). However, it is clear that there are $\mathcal{O}(1)$ fluctuations in the axion field from place to place on scales of order the horizon size when non-perturbative effects switch on ($R \sim 10$ pc today for the QCD axion). These large fluctuations have been conjectured to give rise to so-called "axion miniclusters" [89]. Fluctuations of this type are non-adiabatic, but are *not scale invariant* and give rise to additional power only on scales sub-horizon at PQ symmetry breaking.

The breaking of global symmetries gives rise to topological defects. A broken U(1) creates axion strings, while having $N_{\rm DW} > 1$ in Eq. (36), as in the DFSZ QCD axion model, gives rise to domain walls. When the PQ symmetry breaks after inflation, a number of such defects will remain in the present Universe. Domain walls, if stable, are phenomenologically disastrous, since their energy density scales like $1/a^2$ and they can quickly dominate the energy density of the Universe [90]. They can be avoided if $N_{\rm DW} = 1$ in Eq. (36), which is possible in the KSVZ axion model, although other mechanisms to avoid their disastrous consequences exist (e.g. Ref. [91]). Cosmic strings have a host of additional phenomenology. Perturbations seeded by strings and the decay of domain walls may lead to the existence of heavy axion clumps [92]. For our purposes, the most important impact of axion strings is that their decay can source a population of relic axions, which is discussed below.

The important phenomenological aspects of the unbroken PQ scenario are:

- The average (background) initial misalignment angle is not a free parameter: $\langle \theta_{a,i}^2 \rangle = \pi^2/3.$
- Phase transition relics are present. Their consequences must be dealt with.
- Existence of axion miniclusters?

3.2.2. PQ symmetry broken during inflation, $f_a > H_I/2\pi$

This scenario occurs when $f_a > H_I/2\pi$. It is particularly relevant for GUT scale axions, and all axion DM models combined with low-scale inflation.

As in the previous scenario, PQ symmetry breaking establishes causally disconnected patches with different values of θ_{PQ} , and produces topological defects. However, the rapid expansion during inflation dilutes all the phase transition relics away.¹³ It also stretches out each patch of θ_{PQ} , so that our current Hubble volume began life at the end of inflation with a single uniform

¹³Recall that one of the original motivations for inflation was as a solution to the monopoloe problem of GUT theories [93, 94, 95].
value of θ_{PQ} everywhere. This initial value of θ_{PQ} is completely random. It is again drawn from a uniform distribution, but the existence of many different Hubble patches means that values of θ_{PQ} arbitrarily close to zero or π cannot be excluded, except on grounds of taste or anthropics.

Fluctuations in θ_{PQ} , which later seed structure formation with axion DM, are generated in two different ways in this scenario. Firstly, as we will show in Section 4.4, the axion field has a gravitational Jeans instability. Axion DM will fall into the potential wells established by photons in the radiation era (which were in turn established by quantum fluctuations during inflation). This leads to adiabatic fluctuations.

The second source of axion fluctuations are inflationary isocurvature modes. When the PQ symmetry is broken during inflation, the axion exists as a massless field (or in any case, one with $m_a \ll H_I$). All massless fields in de Sitter space undergo quantum fluctuations with amplitude

$$\delta\phi = \frac{H_I}{2\pi} \,. \tag{42}$$

The amplitude of the power spectrum of these perturbations is proportional to r_T . In de Sitter space, the power spectrum would be scale invariant. Slow roll inflation imparts a red tilt. The isocurvature spectral index is the same as the tensor spectral index, and is also fixed by H_I via inflationary consistency conditions.

Just like tensor modes, DM isocurvature perturbations of this type do not give rise to a large first acoustic peak in the CMB, and are thus constrained to be sub-dominant. The latest *Planck* constraints give $A_I/A_s < 0.038$ [96]. As we will discuss in detail in Section 5.4, this typically forbids the compatibility of $f_a \gtrsim 10^{11}$ GeV axion DM and an observably large r_T .

Isocurvature perturbations also give rise to a backreaction contribution to the homogeneous field displacement (see e.g. Ref. [97])

$$\begin{aligned} \langle \phi_i^2 \rangle &= f_a^2 \theta_{a,i}^2 + \langle \delta \phi^2 \rangle \,, \\ &= f_a^2 \theta_{a,i}^2 + (H_I / 2\pi)^2 \,. \end{aligned} \tag{43}$$

The backreaction sets a minimum value to the misalignment population of axions that can be significant in high scale inflation for heavier ALPs, $m_a \gtrsim 10^{-12}$ eV, and the QCD axion.

The important phenomenological aspects of the broken PQ scenario are:

- The average (background) initial misalignment angle is a free parameter, with a minimum value fixed by backreaction.
- Isocurvature perturbations are produced. Their consequences must be dealt with.
- Use as a probe of inflation?

3.3. Cosmological Populations of Axions

The relic density of axions is $\rho_a = \Omega_a \rho_{\rm crit}$. In cosmology we often discuss the physical density, $\Omega_a h^2$, by factoring out the dimensionless Hubble parameter, h, from the critical density. This gives $\rho_a = \Omega_a h^2 \times (3.0 \times 10^{-3} \text{ eV})^4$.

A relic axion population can be produced in a number of different ways. The four principle mechanisms are:

- Decay product of parent particle.
- Decay product of topological defect.
- Thermal population from the radiation bath.
- Vacuum Realignment.

I will discuss the first three briefly here, but leave most of the details to the references. Vacuum realignment is discussed in detail in Section 4.3.

3.3.1. Decay Product of Parent Particle

A massive particle, X, with $m_X > m_a$, is coupled to the axion field, and decays, producing a population of relativistic axions. If the decay occurs after the axions have decoupled from the standard model then they remain relativistic throughout the history of the Universe. In this case, axions are dark radiation (DR). In cosmology, DR is parameterised via the "effective number of relativistic neutrinos," $N_{\rm eff}$, defined as:

$$\rho_r = \rho_\gamma \left[1 + \frac{7}{8} \left(\frac{4}{11} \right)^{4/3} N_{\text{eff}} \right] \,. \tag{44}$$

Recall that three species of massless neutrinos in the standard model of particle physics contribute $N_{\text{eff}} = 3.04$, the additional 0.04 being contributed by heating after e^+e^- annihilation [98].

Assuming instantaneous decay of the parent particle when it dominates the energy density of the Universe gives:¹⁴

$$\Delta N_{\rm eff} = \frac{43}{7} \left(\frac{10.75}{g_{\star S}(T_r)} \right)^{1/3} \frac{B_a}{1 - B_a} \,, \tag{45}$$

where T_r is the reheating temperature of the decay of the parent particle, B_a is the branching ratio to axions, and $g_{\star S}(T)$ is the entropic degrees of freedom. The evolution of $g_{\star,S}(T)$ in the standard model can be computed or can be looked up, e.g. in the Review of Particle Physics [21].

DR can affect the CMB in a number of ways; for a concise description, see Ref. [103]. If we hold the angular size of the sound horizon fixed (compensating the change in matter radiation equality with a different Hubble constant or DE density), the main effect of DR is to cause additional damping of the highmultipole acoustic peaks in the CMB.¹⁵ This damping tail is well measured by *Planck*, ACT and SPT, giving $N_{\text{eff}} = 3.15 \pm 0.23$ from a representative combination of CMB data [105]. N_{eff} is also constrained by big bang nucleosynthesis (BBN, again see Ref. [105]). Whether this should be combined with the CMB constraint depends on whether the decay producing the axions occurred before or after BBN. An important point to note about neutrino constraints form the

¹⁴If the parent particle does not dominate the energy density of the Universe when it decays, then under certain circumstances it may act as a curvaton [99, 100, 101] and sources correlated isocurvature perturbations, which are also constrained by the CMB. See, e.g., Ref. [102].

¹⁵Recent constraints on N_{eff} in Ref. [104] have separated the damping tail effect from the neutrino anisotropic stress, which changes the angular scale of the higher acoustic peaks (see also constraints on neutrino viscosity in Ref. [105]).

CMB is that they do not care whether the DR is a boson or a fermion. We discuss more consequences of axionic dark radiation in Section 9.7.

A scenario in which axions are produced in this way arises in models with SUSY and extra dimensions. The DR "cosmic axion background" is thus considered a generic prediction of many string and M-theory compactifications. and it has a rich phenomenology (see e.g. Refs. [66, 106, 107, 108] and Sections 9.7 and 9.8.2 of this review). In these models, a Kähler modulus, σ , of the compact space comes to dominate the energy density of the Universe after inflation, leading to an additional matter dominated era and a non-thermal history. The modulus must decay and reheat the Universe to a temperature above $T_{\rm BBN} \sim 3$ MeV, since BBN does not occur successfully in a matter dominated universe.¹⁶ Moduli are gravitationally coupled and are therfore expected to have comparable branching ratios to hidden and visible sectors, and in particular have a large branching ratio to axions, since axions are partnered to moduli by SUSY. The modulus decay rate is given by its mass, $\Gamma_{\sigma} \sim m_{\sigma}^3/M_{pl}^2$ and it decays when $H \sim \Gamma_{\sigma}$. Decay before BBN requires $m_{\sigma} \gtrsim 10$ TeV. Moduli are thus much heavier than axions, and their decay produces a sizeable relativistic axion population, surviving from before BBN until today.

3.3.2. Decay Product of Topological Defect

The breaking of global symmetries leads to the formation of topological defects. In the case of a global U(1) symmetry, like the PQ symmetry, this means global (axionic) strings and (if $N_{\rm DW} > 1$) domain walls. In the broken PQ scenario, topological defects and their decay products are inflated away, and can be ignored, so here we focus on the unbroken PQ scenario. Axion strings decay, producing a population of cold axions, which we discuss below. The energy density in domain walls scales like $\rho_{\rm DW} \sim a^{-2}$ and can quickly dominate the energy density of the Universe, with phenomenologically disastrous results. Thus $N_{\rm DW} > 1$ models (like the DFSZ model) typically require the broken PQ

¹⁶This is the "cosmological moduli problem," see e.g. Refs. [109, 110].

scenario, or some other mechanism to remove the domain walls (see e.g. Ref. [91] and references therein). In this Section I give only the briefest overview of axion production from topological defects: see e.g. Refs. [43, 8, 111] for more details.

Let's focus on strings. Strings are formed by the "winding" of the θ angle. The value of the θ angle is set independently at each point in space when the PQ symmetry breaks. The Goldstone nature of θ homogenizes this value in each horizon volume. As the horizon grows, the homogenized area grows. However, in different horizon volumes, θ will be different. Then, if the θ angle undergoes a winding around any given point in space, the mapping between θ and the spatial co-ordinates does not allow a continuous unwinding, leading to a string-like topological defect along the length of the region enclosed by the winding. Formation of topological defects in cosmology in this manner is known as the Kibble mechanism [112].

Strings in cosmology enter into a "scaling solution," caused by strings within any horizon volume cutting themselves into loops. During the radiation dominated epoch, this requires the string energy density to scale as:

$$\rho_{\rm string} \propto \mu_{\rm string}/t^2 \,, \quad \mu_{\rm string} \sim f_a^2 \ln(f_a d) \,,$$
(46)

where μ_{string} is the energy per unit length of the axion string, and d the characteristic distance between strings. For global strings, this scaling symmetry is maintained by the continuous emission of axions. The change in the number density of axions, n_a , per entropy density, s, per Hubble time, required for this is [43]:

$$\Delta(n_a/s) \sim \frac{\mu_{\text{string}}t^2}{\omega T^3} \Delta(Ht)$$
(47)

where ω is the average energy of the radiated axion.

Recall from Eq. (38) that the axion field begins oscillating when $m_a \sim H$, which occurs at a temperature $T_{\rm osc.}$, and depends on the temperature evolution of the axion mass (we discuss this in more detail for the misalignment population of axions in Section 4). When oscillations commence, axion strings become the boundaries of domain walls connected by strings. For $N_{\rm DW} = 1$, these walls can be "unzipped" by the strings (as explained in Ref. [8]), and the decay of the topological defects is complete. Therefore, the total number of axions produced by string decay in a comoving volume is given by the integral of Eq. (47) from the time of the PQ phase transition at $T = f_a$ up to T_{osc} :

$$\frac{n_a}{s} \sim \int_{T_{\rm osc}}^{f_a} \frac{\mu_{\rm string} dT}{\omega(T) M_{pl}^2} \,. \tag{48}$$

Axions produced by string decay are dominated by the low-frequency modes, making them non-relativistic and contributing as CDM to the cosmic energy budget. Accurate computation of the relic density requires numerical simulation of the PQ phase transition and decay of axion strings in order to determine the energy spectrum, $\omega(T)$. Results of such simulations are commonly expressed as the ratio of axion energy density produced by topological defect decay compared to that produced by misalignment:

$$\Omega_a h^2 = \Omega_{a,\text{mis}} h^2 (1 + \alpha_{\text{dec.}}) \,. \tag{49}$$

For the specific case of the QCD axion, with known temperature dependence of the mass, the value of α_{dec} is calculated.¹⁷ There is a long-standing controversy over what the value of α_{dec} should be, with quoted values ranging from 0.16 to 186 [114, 115, 116, 117], with the true value possibly lying somewhere in between [111].

The uncertainty arises from the form of the spectrum ω . If the radiated axions have the longest wavelengths possible, of order the horizon, then $\omega(t) \sim t^{-1}$ [114], while if the spectrum $\sim 1/k$ (cut off at the horizon and the string size) then $\omega(t) \sim \ln(f_a t) t^{-1}$ [115]. These stem from different assumptions about simulating strings. For the QCD axion mass-temperature relation, this factor of $\ln(f_a t_{\rm osc}) \sim 70$, with the enhancement occurring for the case where $\omega \sim t^{-1}$

¹⁷As we will show shortly, the contribution from misalignment, $\Omega_{a,\text{mis}}h^2$, has a particular scaling with f_a for the QCD axion. Quoting a constant value for $\alpha_{\text{dec.}}$ in the parameterisation Eq. (49) assumes the same scaling with f_a for the population produced by topological defect decay. Ref. [113] show slightly different scalings, but argue that the uncertainty due to mass-dependence is sub-dominant to other uncertainties in the string calculation.

(accounting for the t dependence of μ with $d \sim t$). The modern direct simulation of the PQ field yields the somewhat intermediate result of Ref [111].

This is clearly a very important area of uncertainty in models of high scale inflation and intermediate scale axions that could have consequences for direct detection of the QCD axion. If decay products from topological defects can produce a relic density larger than misalignment ($\alpha_{dec.} \gg 1$), then axions with f_a as low as 10⁹ GeV could be relevant DM candidates (see Section 4.3.2 for quantitative details). Ultimately, if $\alpha_{dec.}$ were too large, then QCD axion DM would be excluded by stellar astrophysics (see Section 9.1). Direct detection of low- f_a axions is outside the reach of ADMX, but may be possible with e.g. open resonator searches (see Section 9.5.1).

Topological defects also source CMB fluctuations (e.g. Ref. [118]). A cosmic string network generates power on all sub-horizon scales [119]. Therefore, axion strings only generate power on scales of order the horizon size at string decay. This scale is small, and is not constrained by the CMB power spectrum, but axion strings may source additional power on minicluster scales.

3.3.3. Thermal Production

If axions are in thermal contact with the standard model radiation, then mutual production and annihilation can lead to a thermal relic population of axions, just as for massive standard model neutrinos and WIMPs. The couplings of an axion to the standard model are only specified in the case of the QCD axion. Furthermore, generic ALPs are often more weakly coupled to the standard model, or at least to QCD, than the QCD axion. For these reasons, we will consider only the thermal population of the QCD axion.

Axions are produced from the standard model plasma by pion scattering, and decouple when the rate for the $\pi + \pi \rightarrow \pi + a$ process drops below the Hubble rate. The thermal axion abundance is fixed by freeze-out at the decoupling temperature (see, e.g. Ref. [43]), with a larger relic density for lower decoupling temperatures. The number density in thermal axions, n_a , relative to the photon number density, n_{γ} is given by

$$n_a = \frac{n_\gamma}{2} \frac{g_{\star,S}(T_0)}{g_{\star,S}(T_D)}, \qquad (50)$$

with T_D the decoupling temperature, and T_0 the CMB temperature today. See Ref. [120] for a more complete formula and a computation involving all relevant standard model production channels. Thermal axions contribute to the effective number of neutrinos as $\Delta N_{\rm eff} \approx 0.0264 n_a/n_{a,\rm eq} \approx 10 n_a$, with $n_{a,\rm eq}$ the thermal equilibrium number density.

Since axion couplings scale inversely with f_a , only low f_a (higher mass) thermally produced axions can contribute a significant amount to the energy budget of the Universe. Thermal populations are significant for $m_a \gtrsim 0.15$ eV, when decoupling occurs after the QCD phase transition (recall that $g_{\star,S}$ reduces dramatically after the QCD phase transition, diluting the abundance of particles produced before it). For the QCD axion respecting $f_a \gtrsim 10^9$ GeV, as suggested by stellar cooling constraints (see Section 9.1), the thermal population is negligible.

Thermal axions produced in this way are relativistic as long as $T_D > m_a$. Once decoupled the axion temperature, T_a , redshifts independently from the standard model temperature, and the axions become non-relativistic when $T_a < m_a$. Thermal axions behave cosmologically in a manner similar to massive neutrinos, and contribute as hot DM, suppressing cosmological structure formation below the free-streaming scale (see Section 4.4.5). Assuming a standard thermal history, current CMB limits from *Planck* on axion hot DM constrain $m_a < 0.529 \rightarrow 0.67$ eV at 95% confidence [121, 122, 123] (for older limits from different datasets including large scale structure and WMAP, see Refs. [124, 125, 126, 127]). AFuture galaxy redshift surveys will be sensitive enough to detect a thermal axion population for all $m_a \ge 0.15$ eV [128]. Relaxing the assumption of a standard thermal history and introducing an early matter-dominated phase and low temperature reheating relaxes the bound on thermal axions, allowing masses as large as a keV [129].

3.3.4. Vacuum Realignment

The process of vacuum realignment is a model independent production mode for axions, also known as the *misalignment mechanism*. It relies only on their defining properties (being associated to spontaneous symmetry breaking, and being a pNGB), and depends only on gravitational (and to some extent self-) interactions. This production mode is our primary focus, and is discussed in detail in Section 4.3.

4. The Cosmological Axion Field

If axions are to have observable effects on cosmology, they must contribute an appreciable amount to the energy density of the Universe. Since the axion mass is so small, this implies large occupation numbers. In this case, axions can be modelled by solving the classical field equations of a condensate. This condensate can have excited states carrying energy and momentum, and indeed it will. There is nothing more mysterious here than using Maxwell's equations to describe the behaviour of electric and magnetic fields. It is also the standard way that scalar field models of inflation and DE are treated.

It is a separate question to ask whether axions form a Bose-Einstein condensate (BEC), and even to define a "BEC" in a cosmological context, where we are certainly not in the ground state. I comment briefly on this in Section 4.7. The results I present below are valid whenever the classical field equations hold, and *do not assume BEC occurs* (except to the extent that it is captured by the classical field equations). Many of the results below also apply to other models of scalar field DM at late times (when oscillations about a quadratic minimum are the only important aspect), though the early time cosmology can be markedly different (e.g. complex fields in Ref. [130], which have equation of state w = 1 at early times).

4.1. Action and Energy Momentum Tensor

The action for a minimally coupled real scalar field in General Relativity is:

$$S_{\phi} = \int d^4x \sqrt{-g} \left[-\frac{1}{2} (\partial \phi)^2 - V(\phi) \right].$$
(51)

For an axion, this action is only valid after symmetry breaking, and after nonperturbative effects have switched on. Before non-perturbative effects have switched on, the axion is massless. Non-perturbative effects do not switch on instantaneously, either, and time (temperature) dependence of the potential can be important. We discuss this shortly, in Section 4.3.

Varying the action with respect to ϕ gives the equation of motion

$$\Box \phi - \frac{\partial V}{\partial \phi} = 0, \qquad (52)$$

where the D'Alembertian is

$$\Box = \frac{1}{\sqrt{-g}} \partial_{\mu} (\sqrt{-g} g^{\mu\nu} \partial_{\nu}) \,. \tag{53}$$

Varying the action with respect to the metric gives the energy momentum tensor

$$T^{\mu}_{\ \nu} = g^{\mu\alpha}\partial_{\alpha}\phi\partial_{\nu}\phi - \frac{\delta^{\mu}_{\ \nu}}{2} [g^{\alpha\beta}\partial_{\alpha}\phi\partial_{\beta}\phi + 2V(\phi)].$$
(54)

As we will show below, there are certain limits in which the axion field behaves as a fluid. See Appendix D for useful definitions for the components of the energy momentum tensor in the fluid case.

4.2. Background Evolution

The background cosmology is defined in Appendix C. Computing the D'Alembertian for the FRW metric and taking $V = m_a^2 \phi^2/2$, the axion equation of motion is:

$$\ddot{\phi} + 3H\dot{\phi} + m_a^2\phi = 0.$$
(55)

The background energy density and pressure of the axion field are:

$$\bar{\rho}_a = \frac{1}{2}\dot{\phi}^2 + \frac{1}{2}m_a^2\phi^2\,,\tag{56}$$

$$\bar{P}_a = \frac{1}{2}\dot{\phi}^2 - \frac{1}{2}m_a^2\phi^2.$$
(57)

When the universe is matter or radiation dominated the scale factor evolves as a power law, $a \propto t^p$. In this case, Eq. (55) has an exact solution:

$$\phi = a^{-3/2} (t/t_i)^{1/2} [C_1 J_n(m_a t) + C_2 Y_n(m_a t)], \qquad (58)$$

where n = (3p - 1)/2, $J_n(x)$, $Y_n(x)$ are Bessel functions of the first and second kind, and t_i is the initial time. The dimensionful coefficients C_1 and C_2 are determined by the initial conditions. For axions in the vacuum realignment mode, the initial conditions are well defined when $H(t_i) \gg m_a$:

$$\phi(t_i) = f_a \theta_{a,i} , \quad \dot{\phi}(t_i) = 0 .$$
(59)

When matter and radiation are both important, such as near matter-radiation equality,¹⁸ or when the axion field can itself dominate the energy density, Eq. (55) must be solved either by approximation or numerically. In the case of axion DM produced by the misalignment mechanism, the most useful approximation to solve Eq. (55) is the WKB approximation.

4.3. Misalignment Production of DM Axions

The misalignment production of DM axions can be computed given the initial conditions of Eq. (59). At symmetry breaking the Hubble rate is much larger than the axion mass, and the field is overdamped. This sets $\dot{\phi} = 0$ initially. The homogeneous value of the field is specified by the scenario for when symmetry breaking occurs with respect to inflation. The term "misalignment" refers to this scenario where there is a coherent initial displacement of the axion field, and "vacuum realignment" to the process by which this value relaxes to the potential minimum.

An important buzz-word to remember about the misalignment production of DM axions is that it is *non-thermal*.

¹⁸Recall that in Λ CDM equality occurs at $z_{\rm eq} \approx 3000$, while the CMB is formed at decoupling, $z_{\rm dec} \approx 1020$. The contribution of radiation to the expansion rate at decoupling cannot be neglected.

4.3.1. Axion-Like Particles

Let's begin with the simple case of an ALP. Given ignorance of the nonperturbative physics, I will describe such an axion only by its mass, which I take to be constant in time. The general picture described here applies to the QCD axion also. The validity of the constant mass assumption will be discussed later in this subsection.

The initial condition $\dot{\phi} = 0$ fixes the relative values of C_1 and C_2 in the exact solution to the background evolution, Eq. (58). The equation of motion is linear, and so the initial field value can be scaled out. Fig. 4 shows the evolution of the axion field, Hubble rate, axion equation of state, and the axion energy density for the solution Eq. (58) in a radiation-dominated universe (p = 1/2), with arbitrary normalization of all dimensionful parameters. The scale factor is shown relative to the initial value, a_i .

At early times when $H > m_a$, the axion field is overdamped and is frozen at its initial value by Hubble friction. The equation of state at early times is $w_a = -1$, and the axion behaves as a contribution to the vacuum energy. This is why axions can serve as models for DE and inflation. All other components of the Universe scale as a to some negative power. If the axion can come to dominate the energy density while it is still overdamped with $w_a < -1/3$, it can drive a period of accelerated expansion. The length of this period depends on the ratio H/m_a when the axion comes to dominate the energy density, which is in turn fixed by the initial displacement of the field (in inflation, this fixes the values of the slow-roll parameters).

Later, when $H < m_a$, the axion field is underdamped and oscillations begin. The equation of state oscillates around $w_a = 0$, and the energy density scales as $\rho_a \propto a^{-3}$. This is the same behaviour as ordinary matter, and is why misalignment axions are a valid DM candidate. The Hubble rate at matter-radiation equality in Λ CDM is approximately $H(a_{eq}) \sim 10^{-28}$ eV. Axions heavier than this begin oscillations in the radiation dominated era and are suitable candidates to compose all the DM.



Figure 4: Evolution of various quantities in the exact solution to the background evolution of an ALP, Eq. (58), for a radiation-dominated universe (p = 1/2). Dimensionful quantities have arbitrary normalization. Vertical dashed lines show the condition defining $a_{\text{osc.}}$. Further discussion of this choice, and the approximate solution for the energy density, is given in the text.

The transition in the axion equation of state can be approximated if we define a fixed value of the scale factor, $a_{\rm osc}$, and simply fix the behaviour of $\rho_a(a)$ at late times to be

$$\rho_a(a) \approx \rho_a(a_{\rm osc})(a_{\rm osc}/a)^3; \quad (a > a_{\rm osc}).$$
(60)

Furthermore, the energy density is approximately constant up until $a_{\rm osc}$ and so we can further approximate $\rho_a(a_{\rm osc}) \approx m_a^2 \phi_i^2/2$. This gives the usual approximation used to calculate axion DM energy density. The energy density in the misalignment population is fixed by the initial field displacement and the mass alone.

How shall we define $a_{\rm osc}$? Roughly, it is when $m_a \gtrsim H$, so we can let $AH(a_{\rm osc}) = m_a$ for some constant A > 1. The larger we set A to be, the better the approximation (assuming we compute $\rho_a(a_{\rm osc})$ from the exact solution). However, we must also play this off against the expense of following oscillations in a numerical solution. The equation of motion, Eq. (38), suggests A = 3 is as a sensible-looking choice. In the example with a radiation dominated universe, I found A = 3 leads to a 40% error in the energy density at late times, with A = 2 giving a better approximation.¹⁹ The approximation Eq. (60) and the location of $a_{\rm osc}$ for A = 2 are also shown in Fig. 4.

In real-Universe examples with a matter-to-radiation transition and late time Λ domination, we found in Ref. [131] that A = 3 works well in most cases. Using the known solutions in matter and radiation domination for H(t) to fix a_{osc} in terms of other cosmological parameters, this gives the following useful approximation to the ULA fractional energy density as a function of the initial

¹⁹As already stated, the approximation in general improves as A gets larger. The poor performance at A = 3 is just because the energy density is falling rapidly at this point and errors are amplified. In this case, 3 is not a lucky number. In numerical solutions including perturbations, taking a larger A will always be better, as the improvement shown here for A = 2 applies only to the exact background solution.

displacement [132]:

$$\Omega_a \approx \begin{cases} \frac{1}{6} (9\Omega_r)^{3/4} \left(\frac{m_a}{H_0}\right)^{1/2} \left\langle \left(\frac{\phi_i}{M_{pl}}\right)^2 \right\rangle & \text{if } a_{\text{osc}} < a_{\text{eq}}, \\ \frac{9}{6} \Omega_m \left\langle \left(\frac{\phi_i}{M_{pl}}\right)^2 \right\rangle & \text{if } a_{\text{eq}} < a_{\text{osc}} \lesssim 1, \\ \frac{1}{6} \left(\frac{m_a}{H_0}\right)^2 \left\langle \left(\frac{\phi_i}{M_{pl}}\right)^2 \right\rangle & \text{if } a_{\text{osc}} \gtrsim 1, \end{cases}$$

$$(61)$$

where I have used angle brackets to denote the average homogeneous value, to remind us of the consequences when the PQ symmetry is broken or unbroken during inflation.

Let's use the WKB approximation to understand the background evolution further. The WKB approximation for $H \ll m_a$ consists of the ansatz solution

$$\phi(t) = \mathcal{A}(t)\cos(m_a t + \vartheta), \qquad (62)$$

where ϑ is an arbitrary phase, and \mathcal{A} is slowly varying such that $\dot{\mathcal{A}}/m_a \sim H/m_a \sim \epsilon \ll 1$. Plugging this into Eq. (38) and working to leading order in ϵ gives the solution $\mathcal{A}(a) \propto a^{-3/2}$. Using this solution we find that the energy density simply scales as $\rho_a \propto \mathcal{A}^2 \propto a^{-3}$, while w_a has rapid oscillations with frequency $2m_a$. Consequently, the average equation of state on time scales $t \gg 1/m_a$ is $\langle w_a \rangle_t = 0$. This gives a general proof as to why w_a oscillates around zero and $\rho_a \propto a^{-3}$ at late times when $H \ll m_a$, independent of any assumptions about the background evolution being matter or radiation dominated.²⁰

The solution for ϕ and ρ_a in the WKB approximation sheds light on the constant-mass assumption we made at the beginning of this section. The magnitude of non-perturbative effects generally varies with temperature, and so the axion mass varies with cosmological time, approaching an asymptotic value for $T \ll T_{\rm NP}$. If the asymptotic value of the mass has been reached before the axion becomes relevant in the energy density and when $a < a_{\rm osc}$ then cosmology will proceed as if we simply take $m_a = m_a(T = 0)$ everywhere. Only the quantities

²⁰This applies to fields oscillating in a harmonic potential, $V(\phi) \sim \phi^2$. Turner [133] proved the more general result for fields oscillating in an anharmonic potential, $V(\phi) \sim \phi^{\alpha}$, giving $\rho \propto a^{-6\alpha/(\alpha+2)}$.



Figure 5: ULA relic density from vacuum realignment in the broken PQ scenario with high scale inflation, $H_I \approx 10^{14}$ GeV. ULAs require $\phi_i > 10^{14}$ GeV in order to contribute more than a few percent to the DM density. Even with high scale inflation, the contribution of isocurvature backreaction is less than a percent of the total DM across the entire ULA parameter space. See Fig. 15 for more details on the allowed region at lower mass.

evaluated at $a = a_{\rm osc}$ matter. In string models, non-perturbative effects stabilise moduli and break SUSY at high energies, while ULAs oscillate in the post-BBN Universe, with $T_{\rm BBN} \ll T_{\rm SUSY}$. In that context, i.e. ULAs from string theory, constant mass is an excellent approximation.

Fig. 5 shows $\Omega_a h^2$ in the broken PQ scenario, for ULAs in the range 10^{-24} eV $\leq m_a \leq 10^{-12}$ eV (where $a_{\rm osc} < a_{\rm eq}$ and ULAs are safe from linear cosmological constraints, see Section 5), with $H_I = 7.8 \times 10^{13}$ GeV (the maximum allowed value with $r_T = 0.1$) for varying $\phi_i = f_a \theta_{a,i}$. The contribution from H_I backreaction to $\Omega_a h^2$ is less than 10^{-4} across the entire range of masses shown: backreaction of isocurvature perturbations can safely be neglected for all ULAs and $\langle \phi_i^2 \rangle \approx \phi_i^2$ can be taken as a completely free parameter. All ULAs require $\phi_i > 10^{14}$ GeV in order to contribute more than a few percent to the DM density. Since $\phi_i \lesssim f_a$ and $H_{I,\max} < 10^{14}$ GeV this implies that ULAs should

always be considered in the broken PQ scenario.

The "anthropic boundary" for ULAs in string theory is defined as the minimum mass where $\Omega_a h^2 = 0.12$ [105] can be obtained with $f_a \leq 10^{16}$ GeV [17]. Plugging $\phi_i = 10^{16}$ GeV into Eq. 61 gives:

$$m_a = 5.3 \times 10^{-19} \text{ eV} \left(\frac{f_a}{10^{16} \text{ GeV}}\right)^{-4} \quad \text{(string anthropic boundary)}, \quad (63)$$

where I have used $z_{eq} = 3400$, $\Omega_c h^2 = 0.12$, $\Omega_b h^2 = 0.022$ and h = 0.67 to fix the radiation density. ULAs heavier than this require (anthropic) tuning of ϕ_i if $f_a \sim 10^{16}$ GeV. ULAs lighter than this require larger decay constants, a large number of individual axions, or some other production mechanism, to contribute a significant amount to the DM density. Since $f_a \leq 10^{16}$ GeV is by no means a hard prediction of string theory, it is worth considering the limit of the anthropic boundary for DM-like axions with $m_a = 10^{-24}$ eV. This is visible in Fig. 5, and from the f_a scaling of Eq. (63). We find $f_a \leq 4 \times 10^{17}$ GeV: ULA DM is natural for comfortably sub-Planckian values of the decay constant.

4.3.2. The QCD Axion

QCD non-perturbative effects switch on at $T \sim \Lambda_{\rm QCD} \sim 200$ MeV, precisely when the QCD axion with intermediate f_a begins oscillations. The temperature dependence of the axion mass in QCD is given by:

$$m_a^2(T)f_a^2 = \chi_{\text{top.}}(T),$$
 (64)

where $\chi_{\text{top.}}(T)$ is the QCD topological susceptibility, which must be calculated. The original calculation is due to Ref. [41] and is reviewed in e.g. Ref. [134], while a modern calculation in the 'interacting instanton liquid model' (IILM) is given in Ref. [113]. A simple power-law dependence of the axion mass on temperature applies at high temperatures, T > 1 GeV:

$$m_a^2(T) = \alpha_a \frac{\Lambda_{\rm QCD}^3 m_u}{f_a^2} \left(\frac{T}{\Lambda_{\rm QCD}}\right)^{-n} \,. \tag{65}$$

This should be matched to the zero temperature value, Eq. (5), at low $T \lesssim \Lambda_{\rm QCD}$.

The standard [41] value for the power-law from the dilute instanton gas model (DIGM) is $n = 7 + n_f/3 + \cdots \approx 8$ (where n_f is the number of fermions active at a given temperature). The fits of Ref. [113] from the IILM give n = 6.68and $\alpha_a = 1.68 \times 10^{-7}$ (which also agrees with Ref. [135]). The temperature dependence can also be computed non-perturbatively on the lattice in the pure Yang-Mills limit (e.g. Refs. [136, 137, 138, 139]), and at low temperatures from chiral perturbation theory (for a recent calculation, see Ref. [140] and references therein). The lattice calculations of Ref. [136] find n = 5.64 (compare to the pure Yang-Mills, $n_f = 0$, DIGM). Ref. [140] consider a range between n = 2and n = 8 from lattice and instanton calculations respectively.

The temperature of the Universe in the radiation dominated era is determined by the Friedmann equation in the form

$$3H^2 M_{pl}^2 = \frac{\pi^2}{30} g_\star T^4 \,. \tag{66}$$

Taking the standard n = 8 result, using that $g_{\star} = 61.75$ for tempertures just above the QCD phase transition, and defining $3H(T_{\rm osc}) = m_a$, the QCD axion with $f_a < 2 \times 10^{15}$ GeV begins oscillating when T > 1 GeV [134]. From this point on, axion energy density scales as a^{-3} independently of the behaviour of $m_a(T)$. The relic density can thus be reliably computed from the hightemperature power-law behaviour of $m_a(T)$, scaled as a^{-3} from $T_{\rm osc}$. The relic density is fixed by the initial misalignment angle and f_a . For $f_a < 2 \times 10^{15}$ GeV it is given by [134]

$$\Omega_a h^2 \sim 2 \times 10^4 \left(\frac{f_a}{10^{16} \text{ GeV}}\right)^{7/6} \left\langle \theta_{a,i}^2 \right\rangle. \tag{67}$$

For $f_a \gtrsim 2 \times 10^{17}$ GeV oscillations begin when $T < \Lambda_{\rm QCD}$, such that the mass has reached its zero-temperature value. In this case the relic density is

$$\Omega_a h^2 \approx 5 \times 10^3 \left(\frac{f_a}{10^{16} \text{ GeV}}\right)^{3/2} \langle \theta_{a,i}^2 \rangle .$$
(68)

Note that there is not an overlapping region of validity for Eqs. (67) and (68). For 2×10^{15} GeV $\leq f_a \leq 2 \times 10^{17}$ GeV oscillations begin during the QCD epoch, the dilute instanton gas approximation breaks down and the relic density calculation is more complicated (see e.g. Refs. [134, 113, 141]). However, it is argued in Ref. [134] that Eq. (67) is a good approximation for $f_a < 6 \times 10^{17}$ GeV. For our simple purposes of illustration, we use Eq. (67) for all $f_a < M_{pl}$.

So far, we have computed the relic density using the harmonic potential, $V(\phi) = m_a^2 \phi^2/2$. For large initial displacements, $\theta_i \gtrsim 1$, anharmonic corrections caused by axion self-interactions become important. The potential becomes flatter at increased θ_{PQ} , causing the axion field to spend more time with $w_a \approx -1$, thus delaying a_{osc} and increasing the relic abundance relative to the harmonic approximation. Anharmonic effects can be taken into account with a correction factor by replacing

$$\langle \theta_{a,i}^2 \rangle \to \langle \theta_{a,i}^2 \mathcal{F}_{\text{anh.}}(\theta_{a,i}) \rangle ,$$
 (69)

where $\mathcal{F}_{anh.}(x) \to 1$ for small x and monotonically increases as $x \to \pi$. An analytic approximation to $\mathcal{F}_{anh.}(x)$ for the cosine potential is [142]

$$\mathcal{F}_{\text{anh.}}(x) = \left[\ln\left(\frac{e}{1-x^2/\pi^2}\right)\right]^{7/6}$$
. (70)

Note that the use of Eqs. (69) and (70) breaks down if the axion field comes to dominate the energy density, driving a period of inflation, since they rely on the assumption that oscillations begin in a radiation-dominated background.

A full numerical computation of the relic abundance valid for all f_a in the IILM, taking into account the temperature dependence of g_{\star} in the standard model and all anharmonic effects, is given in Ref. [113].

Axions produced by misalignment behave as DM, and we know that the DM density is $\Omega_c h^2 \approx 0.12$. Axions may not be all the DM, but they had better not produce too much of it, so we must have $\Omega_a h^2 < 0.12$.²¹ Eq. (67), and its anharmonic corrections Eqs. (69) and (70), inform the classic discussions on the QCD axion and "natural" values for f_a [25, 27, 26, 143].

²¹Violating this constraint is sometimes, misleadingly, called "overclosing the Universe," a phrase which dates from before the precision cosmology era, when one simply demanded $\rho_a < \rho_{\text{crit}}$ for some approximate value of H_0 .

First, let's just take $\langle \theta_{a,i}^2 \rangle$ to be a free parameter, and work out the consequences. High f_a axions produce too much DM unless $\theta_{a,i} \ll 1$. On the other hand, low f_a axions can only produce a fraction of the DM unless $\theta_{a,i}$ is tuned very close to π such that anharmonic corrections can boost the relic density. The "sweet spot" where $\Omega_a h^2 = 0.12$ is achieved for $\theta_{a,i} \approx 1$ is at $f_a \approx 3 \times 10^{11}$ GeV. The range of f_a where $\Omega_a h^2 \approx 0.12$ can be achieved with minimal tuning of $\theta_{a,i}$ towards zero or π is the region where broken PQ axions are "natural." It's boundaries clearly depend on taste, but allowing for tuning at the level 10^{-2} it is:

$$8 \times 10^9 \text{ GeV} \lesssim f_a \lesssim 1 \times 10^{15} \text{ GeV} \quad (\text{no tuning, broken PQ}).$$
 (71)

In the unbroken PQ scenario the relic abundance is fixed by taking $\langle \theta_{a,i}^2 \rangle = \pi^2/3$. Keeping $\Omega_a h^2 < 0.12$ and satisfying bounds from stellar cooling and supernovae defines the *classic axion window*:

$$1 \times 10^9 \text{ GeV} \lesssim f_a \lesssim 8.5 \times 10^{10} \text{ GeV}$$
 (classic axion window, unbroken PQ). (72)

Axions with $f_a \gtrsim 10^{15}$ GeV are sometimes referred to as living in the anthropic axion window [144, 145, 146]. It is so-called because although $\theta_{a,i}$ must be tuned small, if it was not small and the DM density was too large, the Universe would not be conducive to the formation of galaxies and life.²² Note that the anthropic window is automatically open to high f_a axions, since for $r_T < 1$, $f_a \gtrsim 10^{15}$ GeV is always in the broken PQ scenario where $\theta_{a,i}$ is a free parameter, although the backreaction contribution may be important depending on the value of H_I .

Let's bring together everything we know about the QCD axion DM relic

²²Refs. [147, 148] discuss the interesting case of anthropic selection with multiple axion fields. An additional fine-tuning measure is also applied based on isocurvature perturbations (see Section 5.4). However, when applied to isocurvature, the measure used in Refs. [147, 148] assumes that the inflationary parameter ϵ_{inf} has a flat prior. A least information (Jeffreys) prior on the unknown physical scale H_I would yield very different conclusions.



Figure 6: QCD axion DM relic density from vacuum realignment in the broken PQ scenario. Isocurvature constraints are ignored, see Fig. 17. Left panel: Low scale inflation, $H_I = 2\pi \times 10^9$ GeV. All of the allowed range of f_a has PQ symmetry unbroken during inflation. Large f_a requires tuning $\theta_{a,i}$ in order not to produce too much DM. Right Panel: High scale inflation, $H_I = 10^{14}$ GeV. Backreaction produces too much DM for all $f_a \leq 3 \times 10^{15}$ GeV.

density from vacuum realignment into two equations:

$$\Omega_{a}h^{2} \approx \begin{cases} 2 \times 10^{4} \left(\frac{f_{a}}{10^{16} \text{ GeV}}\right)^{7/6} \frac{\pi^{2}}{3} \mathcal{F}_{\text{anh.}}(\pi/\sqrt{3})(1+\alpha_{\text{dec}}) & (\text{unbroken PQ}), \\ 2 \times 10^{4} \left(\frac{f_{a}}{10^{16} \text{ GeV}}\right)^{7/6} (\theta_{a,i}^{2} + H_{I}^{2}/(2\pi f_{a})^{2}) \mathcal{F}_{\text{anh.}}\left(\sqrt{\theta_{a,i}^{2} + H_{I}^{2}/(2\pi f_{a})^{2}}\right) & (\text{broken PQ}). \end{cases}$$
(73)

For simplicity, as stated above, I am going to assume that Eq. (67) holds for all f_a (see the discussion below Eq. 68). See Section 3.3.2 and 4.7 for discussion on the difference between the misalignment and topological defect populations.

Fig. 6 is a contour plot of $\Omega_a h^2$ as a function of f_a and $\theta_{a,i}$ for the broken PQ scenario in two different inflation models. The first takes $H_I = 2\pi \times 10^9$ GeV, so that all of the allowed range of f_a has the PQ symmetry broken during inflation. The second scenario takes $H_I = 10^{14}$ GeV, i.e. about as large as it can be without violating current tensor constraints. In the case of low scale inflation, the entire allowed range of f_a can produce the required DM density by vacuum realignment. Large f_a requires tuning of $\theta_{a,i}$ in order to satisfy $\Omega_a h^2 < 0.12$. In the high scale inflation case, backreaction of isocurvature perturbations leads



Figure 7: QCD axion DM relic density from vacuum realignment in the unbroken PQ scenario. The fixed value of $\langle \theta_{a,i}^2 \rangle = \pi^2/3$ excludes all axions with $f_a \gtrsim 9 \times 10^{10}$ GeV for producing too much DM. The uncertainty in axion production from string decay, reflected in the range for $\alpha_{\rm dec}$, means that all axions with lower f_a can produce a significant contribution to the DM.

to too much DM production for $f_a \lesssim 5 \times 10^{16}$ GeV. Large $f_a \gtrsim 5 \times 10^{16}$ GeV anthropic axions appear compatible with high scale inflation if we allow $\theta_{a,i}$ to be tuned, however we have so far only considered constraints from the relic density, and not from the isocurvature amplitude. We will see in Section 5.4 that isocurvature constraints imply that high- f_a axions are essentially incompatible with high-scale inflation.

Fig. 7 is a contour plot of $\Omega_a h^2$ as a function of f_a and $\alpha_{\text{dec.}}$ in the unbroken PQ scenario. Based on constraints from r_T , the largest possible value of f_a in this scenario is $f_a \approx 10^{14} \text{ GeV}/2\pi$, and I allow $\alpha_{\text{dec}} \in [0.16, 186]$. In the unbroken PQ scenario, the fixed value of $\langle \theta_{a,i}^2 \rangle = \pi^2/3$ excludes all axions with $f_a \gtrsim 9 \times 10^{10} \text{ GeV}$ for producing too much DM. The possible range of α_{dec} values means that all axions with lower f_a than this have the possibility of providing the correct DM abundance. This defines the classic axion window. Note that if $\alpha_{\text{dec}} \gtrsim 200$ then the QCD axion in the unbroken PQ scenario, satisfying astrophysical constraints, would be completely excluded unless the excess DM abundance could be diluted (e.g. by late-time entropy production). This possibility is the source of the controversy over the axion abundance by string decay discussed in Section 3.3.2.

4.4. Cosmological Perturbation Theory

All specific results here assume that cosmological history begins in the radiation dominated universe after reheating. I work in two gauges: the synchronous gauge and the Newtonian gauge. These gauges, the gauge transformations between them and the equations of motion for matter and radiation, are given in the classic, and endlessly useful, Ref. [149] (see also Ref. [150]).²³ The Newtonian gauge is useful (obviously) for the Newtonian limit (discussed in more detail in the following subsection). The Newtonian potentials Ψ and Φ are

²³As usual in cosmology, note that the adage "the Russians did it first" holds very well here. If you are so inclined, you can find everything you need in Landau and Lifschitz [151]. Another useful early reference is Ref. [152]. I refer explicitly to Ref. [149] as it addresses specifically the CMB computation.

also transparently related to the gauge invariant curvature perturbation, and to the integrated Sachs-Wolfe (ISW) source terms for the CMB. The synchronous gauge, with potentials h (not to be confused with the reduced Hubble rate, also denoted h) and η , on the other hand, makes the CDM evolution particularly simple, as $\theta_c \equiv 0$. The synchronous gauge is also used by the popular CMB Boltzmann solver CAMB [153]. The full treatment of ULAs in the synchronous gauge has been implemented in AXIONCAMB, described in Ref. [131], and soon to be publicly released. Another popular Boltzmann solver is CLASS [154, 155], with a ULA model implemented in Ref. [156].

In this section I work primarily in the fluid treatment of axion perturbations. This can be derived from the perturbed field equation. In Fourier space in synchronous gauge this is

$$\delta\phi'' + 2\mathcal{H}\delta\phi' + (k^2 + m_a^2 a^2)\delta\phi = -\frac{1}{2}\phi' h', \qquad (74)$$

while in Newtonian gauge it is

$$\delta\phi'' + 2\mathcal{H}\delta\phi' + (k^2 + m_a^2 a^2)\delta\phi = (\Psi' + 3\Phi')\phi' - 2m_a^2 a^2\phi\Psi,$$
(75)

where primes denote derivatives with respect to conformal time, $ad\tau = dt$ (not to be confused with the optical depth, also denoted τ), and the conformal Hubble rate is $\mathcal{H} = aH$. The perturbed axion field is $\delta\phi$; the background field is ϕ .

4.4.1. Initial Conditions

Initial conditions are set for all modes, k, when they are super-horizon $k \ll aH$ and at early times during the radiation era. I present the simplest, zeroth order initial conditions. Corrections to these results can be derived order-by-order in the super-horizon/early-time limit. The computation is described in Ref. [157], with results specific to axions given in Ref. [131].

If all cosmological perturbations are seeded by single field inflation, the initial conditions are *adiabatic*. Radiation is the dominant component at early times, and carries the inflationary curvature perturbation. The adiabatic condition relates the overdensity in photons to the overdensity in any other fluid

component, i:

$$\delta_i = \frac{3}{4}(1+w_i)\delta_\gamma \,.$$

At early times, the axion equation of state is $w_a \approx -1$ and so $\delta_a = \delta \phi = 0$ in the adiabatic mode in the early-time, super-horizon perturbative-expansion.

This adiabatic initial condition seems very different from the standard CDM adiabatic initial condition where $\delta_c = 3\delta_{\gamma}/4$. That is because we are beginning when axions are not behaving as CDM. As the axion field rolls and begins oscillating around $w_a = 0$, the axions begin to cluster and fall into the potential wells set up by the photons. At late times, $a > a_{\rm osc}$, this evolution "locks on" to the standard CDM behaviour on large scales, as we will show from numerical results shortly.

Isocurvature initial conditions can be thought of in a number of ways. Commonly, they are thought of as entropy perturbations: i.e. perturbations in relative number density of particles of different species that leave the total curvature unperturbed. An isocurvature perturbation between two species, i and j, can be written in a gauge invariant way as (e.g. Ref. [158] and references therein)

$$S_{ij} = 3(\zeta_i - \zeta_j) \tag{77}$$

(76)

where ζ_i is the curvature perturbation due to a single species:

$$\zeta_i = -\Psi - H \frac{\delta \rho_i}{\dot{\rho}_i} \,. \tag{78}$$

The total curvature perturbation is

$$\zeta = \frac{\sum_{i} (\rho_i + P_i) \zeta_i}{\sum_{i} (\rho_i + P_i)} \,. \tag{79}$$

The most useful practical definition for all cosmological initial conditions is to think of them as simply the different normal (eigen) modes of the energy momentum tensor [157]. One then finds the early time, $\tau \ll 1$, super horizon, $k\tau \ll 1$, expansion for each mode. In the synchronous gauge each mode can be identified by the leading, zeroth order, behaviour of the fluid variables and the

metric potentials:

$\eta = 1$	(adiabatic mode),	(80)
$\delta_i = 1$	(density isocurvature in species $i)$,	(81)
$\theta_i = k$	(velocity isocurvature in species i),	(82)

with all other components unperturbed. At higher orders one then selects the growing mode for each component. The correct selection of this is crucial. For example the adiabatic mode has (e.g. Refs. [149, 157])

$$\delta_{\gamma} = -\frac{1}{3} (k\tau)^2 \,, \tag{83}$$

and from the equations of motion one finds the condition Eq. (76) relates this to the other species at each order in the perturbative-expansion, and also accounts for possible evolution of w_i (as is the case for the slowly rolling axion field at early times [131]).

In the axion iscocurvature mode, relevant for the broken PQ scenario, the initial condition is $\delta_a = 1$, with all other species unperturbed at zeroth order. The normalization and spectrum can be multiplied afterwards since the equations are linear. The spectrum is a power law, with spectral index $(1 - n_I) = 2\epsilon_{inf}$ (for inflationary slow-roll parameter ϵ_{inf} , see Section 7.2).

4.4.2. Early Time Treatment

At early times, the background equation of motion should be solved numerically to find the evolution of the axion equation of state, $w_a(\tau)$. With this in hand, the background energy density evolves as

$$\rho_a' = -3\mathcal{H}\rho_a(1+w_a)\,.\tag{84}$$

The equation of state also specifies the evolution of the adiabatic background sound speed:

$$c_{\rm ad}^2 = w_a - \frac{w_a'}{3\mathcal{H}(1+w_a)}$$
 (85)

The second order perturbed equations of motion can be rewritten as two first order equations for the axion overdensity, δ_a and dimensionless perturbed heat flux, $u_a = (1 + w_a)v_a$. The equation of state and adiabatic sound speed specify the background evolution-dependent co-efficients in the equations of motion for the fluid components. Using the result that the sound speed in perturbations, $c_s^2 = \delta P_a / \delta \rho_a = 1$ in the $\delta \phi = 0$ axion comoving gauge, the transformation to fluid variables can be performed exactly [159]. Performing a gauge transformation to the synchronous gauge, the equations of motion read [131]:

$$\delta'_{a} = -ku_{a} - (1+w_{a})h'/2 - 3\mathcal{H}(1-w_{a})\delta_{a} - 9\mathcal{H}^{2}(1-c_{\rm ad}^{2})u_{a}/k\,, \qquad (86)$$

$$u'_a = 2\mathcal{H}u_a + k\delta_a + 3\mathcal{H}(w_a - c^2_{\rm ad})u_a \,. \tag{87}$$

I stress that at this stage no approximations have been made. Given the evolution of $w_a(\tau)$ (or equivalently $\phi(\tau)$) the evolution of δ_a and u_a specify the evolution of $\delta\phi$ (with metric potentials sourced by all species).

Note that if $\phi' = 0$ then $w_a = -1$ and $w'_a = 0$. In this case, an adiabatic fluctuation with $\delta \phi = \delta \phi' = 0$ in Eq. (74) has no source and will not grow. The same holds in the fluid variables: $w_a = -1$ leads to vanishing metric source in the fluid equations, and so if $\delta_a = u_a = 0$ initially then this remains so, and no growth occurs.

In this picture, the axions source the Einstein equations with density, pressure and velocity perturbations as

$$\delta \rho_a = \rho_a \delta_a \,, \tag{88}$$

$$\delta P_a = \rho_a [\delta_a + 3\mathcal{H}(1 - c_{\rm ad}^2)(1 + w_a)u_a/k], \qquad (89)$$

$$\rho_a (1+w_a) v_a = \rho_a u_a \,. \tag{90}$$

4.4.3. The Axion Effective Sound Speed

When $a > a_{osc}$, w_a and c_{ad}^2 oscillate rapidly in time compared to the Hubble scale and all other quantities of interest (e.g. the curvature perturbation evolves on time scales of order H). The exact fluid equations now become numerically expensive to solve, and an approximation for the perturbed fluid equations, akin to the $w_a = 0$ approximation in the background equations of motion, is necessary. Consider the general equation of motion for fluids in synchronous gauge [149]:

$$\delta' = -(1+w)(\theta + h'/2) - 3\mathcal{H}(c_s^2 - w)\delta,$$

$$\theta' = -\mathcal{H}(1-3w)\theta - \frac{w'}{1+w}\theta + \frac{c_s^2}{1+w}k^2\delta,$$
 (91)

where I have only assumed the vanishing of anisotropic stress, which is valid at first order in perturbation theory for a scalar field. The evolution is specified by two quantites: the equation of state, w, and the sound speed in perturbations:²⁴

$$c_s^2 = \frac{\delta P}{\delta \rho} \,. \tag{92}$$

For an axion at late times, $a > a_{osc}$, we know how to approximate the time averaged equation of state: $\langle w_a \rangle_t = \langle w'_a \rangle_t = 0$ (see Section 4.3.1). If we can simply find a similar expression for $\langle c_s^2 \rangle_t$ evaluated in the appropriate gauge, then we can use Eqs. (91) to specify the evolution of the axion overdensity. The pressure source of the Einstein equations due to axions will then be given by $\delta P_a = \langle c_s^2 \rangle_t \rho_a \delta_a$.

Just as for the background, we can use the WKB approximation by writing the background field and field perturbation as

$$\phi = a^{-3/2} [\phi_+ \cos mt + \phi_- \sin mt], \qquad (93)$$

$$\delta\phi = \delta\phi_+ \cos mt + \delta\phi_- \sin mt \,, \tag{94}$$

where the functions $\delta \phi_{\pm}$ depend on wavenumber k as well as time. It is now possible to find the effective sound sound speed in the gauge comoving with the time-averaged axion fluid (see e.g. Refs. [160, 161] for the derivation):

$$\langle c_s^2 \rangle_t = c_{s,\text{eff}}^2 = \frac{k^2 / 4m_a^2 a^2}{1 + k^2 / 4m_a^2 a^2} \,.$$
(95)

This effective sound speed is the key to understanding the difference between ULAs and CDM in terms of structure formation.

 $^{^{24}}$ See Appendix D for discussion of different definitions of the scalar field sound speed and the relations between them.

The metric potentials in the axion comoving gauge are defined in the same way as the synchronous gauge. The gauge transformation between the two gauges induces additional terms to Eqs. (91) that decay on sub-horizon scales [131]. The axion fluid equations of motion in the synchronous gauge are:

$$\delta_a' = -ku_a - \frac{h'}{2} - 3\mathcal{H}c_{s,\text{eff}}^2\delta_a - 9\mathcal{H}^2 c_{s,\text{eff}}^2 u_a/k,\tag{96}$$

$$u'_{a} = -\mathcal{H}u_{a} + c_{s,\text{eff}}^{2}k\delta_{a} + 3c_{s,\text{eff}}^{2}\mathcal{H}^{2}u_{a}.$$
(97)

4.4.4. Growth of Perturbations and the Axion Jeans Scale

So far, we've been very precise and set up the equations of motion and initial conditions as they would be used in numerical Boltzmann equation solver to compute cosmological observables in the real Universe.

Let's take a step back for a moment to a simplified situation, and consider a Universe dominated by axion DM, and work in the Newtonian gauge. Let's take the sub-horizon limit, so that we can use the Poisson equation in its usual form:

$$k^2 \Psi^2 = -4\pi G a^2 \rho \delta \tag{98}$$

Gauge transformations on the effective sound speed between the synchronous and Newtonian gauge also vanish in this limit. Combining the equations for $\dot{\delta}_a$ and $\dot{\theta}_a$ into a single second order equation for δ_a , and using the Poisson equation to eliminate the Netwonian potential, gives the equation of motion for δ_a in physical time:

$$\ddot{\delta}_a + 2H\dot{\delta}_a + (k^2 c_{s,\text{eff}}^2 / a^2 - 4\pi G\rho_a)\delta_a = 0.$$
(99)

This is the equation for an oscillator with time-dependent mass and friction. The mass term in this equation expresses the competition between density and pressure during gravitational collapse. The origin of the effective sound speed and pressure in the axion equation of motion is scalar field gradient energy.

On large scales, $k^2 c_s^2 \to 0$, density wins and axion DM has a Jeans instability [162].²⁵ The equation of motion is exactly the same as for CDM, with the

 $^{^{25}}$ The growth of perturbations for small k, despite positive mass-squared for the perturba-



Figure 8: The exact scale-dependent linear growth for an axion DM dominated universe, Eq. (102), at three values of $\tilde{k} = k/\sqrt{m_a H_0}$, as a function of a/a_i . Normalization is arbitrary. Note that the initial scale factor in this case must obey $a_i > a_{osc}$ for the solutions to hold. *Left panel*: The growing mode, $D_+(k, a)$, Eq. (103). *Right Panel*: The decaying mode, $D_-(k, a)$, Eq. (104).

usual growing, $\delta_a \propto a$, and decaying, $\delta_a \propto a^{-3/2}$, modes. On small scales, the pressure term dominates over the density, and δ_a oscillates without growing.

The scale where density and pressure are in equilibrium and $4\pi G\rho_a = k^2 c_s^2$ is known as a the *axion Jeans scale*, and it defines a particular wavenumber, k_J . Modes with $k < k_J$ grow, while modes with $k > k_J$ oscillate. The buzz-phrase to remember referring to axion perturbations is that there is *scale-dependent* growth, and that axion DM differs from CDM on scales below the axion Jeans scale.

In the limit $k/m_a a < 1$ the sound speed has the approximate form:

$$c_{s,\text{eff}}^2 \approx \frac{k^2}{4m_a^2 a^2} \,.$$
 (100)

tions in Eqs. (74) and (75), can be understood from the rapid oscillations in ϕ' causing the system to act as a driven oscillator [159, 163].

The Jeans scale is given by

$$k_J = (16\pi G a \rho_{a,0})^{1/4} m_a^{1/2} = 66.5 a^{1/4} \left(\frac{\Omega_a h^2}{0.12}\right)^{1/4} \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{1/2} \text{ Mpc}^{-1}.$$
(101)

With $\rho_a = \rho_{\rm crit} a^{-3}$ giving the matter-dominated solution for H, and using the approximation Eq. (100) for the sound speed, there is an exact solution to Eq. (99) given by:

$$\delta_a = C_1 D_+(k, a) + C_2 D_-(k, a) \,. \tag{102}$$

The closed-form expressions for $D_{\pm}(k, a)$ are:

$$D_{+}(k,a) = \frac{3\sqrt{a}}{\tilde{k}^{2}} \sin\left(\frac{\tilde{k}^{2}}{\sqrt{a}}\right) + \left[\frac{3a}{\tilde{k}^{4}} - 1\right] \cos\left(\frac{\tilde{k}^{2}}{\sqrt{a}}\right), \quad (103)$$

$$D_{-}(k,a) = \left[\frac{3a}{\tilde{k}^4} - 1\right] \sin\left(\frac{\tilde{k}^2}{\sqrt{a}}\right) - \frac{3\sqrt{a}}{\tilde{k}^2} \cos\left(\frac{\tilde{k}^2}{\sqrt{a}}\right) \,, \tag{104}$$

where $\tilde{k} = k/\sqrt{m_a H_0} \propto k/k_J$. The solutions $D_{\pm}(k, a)$ are plotted in Fig. 8 at three different values of \tilde{k} . For low \tilde{k} , $D_{+}(k, a) \sim a$ is the usual growing mode, and $D_{-}(k, a) \sim a^{-3/2}$ is the usual decaying mode. For intermediate \tilde{k} there are some oscillations at early times while the mode is below the Jeans scale. At late times, it moves above the Jeans scale and picks up the same growing/decaying behaviour as the low \tilde{k} mode. For high \tilde{k} the mode is always below the Jeans scale, and both D_{+} and D_{-} oscillate, retaining constant amplitude.

Finally, let's return to the real Universe. Fig. 9 shows the evolution of the axion overdensity computed using AXIONCAMB, in a realistic model. The axion mass is $m_a = 10^{-26}$ eV, and axions compose all the DM (we will see shortly that this combination of mass and energy density contribution are actually ruled out precisely because of the effects shown here). During the radiation era, before $a_{\rm osc}$, the adiabatic axion perturbation is small. As the axion field begins to roll, the overdensity grows, approaching the CDM value. At low k (large scales), the overdensity locks on to the standard CDM adiabatic evolution, despite the different initial conditions between axions and CDM. This occurs before matter-radiation-equality ($a \sim 10^{-3}$), and today (a = 1) the CDM and axion models have the same amplitude of density perturbations on large scales.



Figure 9: Evolution of the axion overdensity, for a ULA mass of $m_a = 10^{-26}$ eV and a series of wave-numbers k (as shown in the figure), compared to standard CDM (dashed). Axions compose all the DM in this model. Normalization is arbitrary. All cosmological parameters take realistic values. Reproduced (with permission) from Ref. [131]. Copyright (2015) by The American Physical Society.

At intermediate k, growth is suppressed relative to CDM for some time after equality, and at a = 1 the axion amplitude is slightly suppressed relative to CDM. The highest k mode has $k > k_J$ initially, and oscillates for some time, leading to a greatly suppressed axion amplitude relative to CDM on small scales.

4.4.5. Transfer Functions: Relation to WDM and Neutrinos

Thermal DM that was relativistic at freeze-out leads to suppressed clustering power compared to CDM on scales that were sub-horizon while the particles were still relativistic. This gives rise to the free-streaming scale, $k_{\rm fs}$ [43], which is of cosmological size in models of hot dark matter (HDM, including $m_{\nu} \leq 1$ eV standard model neutrinos, see e.g. Refs. [164, 165, 166]) and warm dark matter (WDM, including sterile neutrinos and thermal gravitinos with $m_X \sim 1$ keV, see e.g. Refs. [167, 168, 169, 170]). Suppression of clustering power below the axion Jeans scale (large wavenumbers, $k > k_J$) bears a qualitative similarity to the effects of these low-mass thermal DM models [171, 172].

In linear theory, modifications to the power spectrum relative to Λ CDM can always be expressed by the use of a transfer function:

$$P_X(k,z) = T_X^2(k,z)P_{\Lambda \text{CDM}}(k,z).$$
 (105)

The function $T_X(k, z)$ accounts for both scale and redshift dependence. In Λ CDM, growth is scale-independent for $z \leq \mathcal{O}(100)$, after the baryon acoustic oscillations (BAO) have frozen-in, and radiation ceases to be relevant in the expansion rate. Therefore, the linear-theory Λ CDM power spectrum at any redshift $z \leq 100$ can be obtained from the one at z = 0 by use of the linear growth factor, D(z):²⁶

$$P_{\Lambda \text{CDM}}(k,z) = \left(\frac{D(z)}{D(0)}\right)^2 P_{\Lambda \text{CDM}}(k).$$
(106)

 $^{^{26}}$ The z = 0 power spectrum must in general be computed numerically. It is itself a product of the primordial power spectrum with some transfer function. Some useful fits for this transfer function can be found in Refs. [173, 174].

The linear growth factor is [175]:

$$D(z) = \frac{5\Omega_m}{2H(z)} \int_0^{a(z)} \frac{da'}{(a'H(a')/H_0)^3} \,. \tag{107}$$

Axions and thermal DM induce scale-dependent growth, which causes the suppression of power relative to Λ CDM. However, if this can be neglected on the scales and redshifts of interest, then a redshift-independent transfer function, T(k), can be used to describe the effects of the alternative DM model on structure formation.

Over a range of scales, the redshift-independent transfer function is a useful description of WDM, for $m_X \gtrsim 0.1$ keV, and for ULAs with $m_a \gtrsim 10^{-24}$ eV. For lighter ULAs and for HDM, scale-dependent growth remains relevant at late times and the transfer function is redshift-dependent. These lightest ULAs and HDM require their own detailed treatment, and physics other than the power suppression currently drives constraints. We will discuss them independently when the time comes.

WDM and ULAs with $m_a \gtrsim 10^{-24}$ eV can be described by the transfer functions [170, 176]:²⁷

$$T_{\rm W}(k) = (1 + (\alpha k)^{2\mu})^{-5/\mu}, \qquad (108)$$

$$T_{\rm F}(k) = \frac{\cos x_J^3(k)}{1 + x_J^8(k)},\tag{109}$$

where I have used "F" standing for "Fuzzy CDM" for ULAs described by this transfer function. These transfer functions assume that all of the DM is composed of ULAs or WDM, and cannot be used for mixed DM models. The fitting

²⁷The WDM transfer function can be computed exactly in the Boltzmann code CLASS [155].

parameters are

$$\mu = 1.12, (110)$$

$$\alpha = 0.074 \left(\frac{m_X}{\text{keV}}\right)^{-1.15} \left(\frac{0.7}{h}\right) \text{ Mpc}, \qquad (111)$$

$$x_J(k) = 1.61 \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{1/18} \frac{k}{k_{J,\text{eq}}}, \qquad (112)$$

$$k_{J,\text{eq}} = 9 \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{1/2} \text{ Mpc}^{-1}.$$
 (113)

The WDM transfer function falls off as a power-law in wavenumber. Intuitively, this is because it is caused by thermal velocities, with temperature scaling as $T \sim 1/a$, and is related to the comoving wavenumber of order the horizon size when $T \sim m_X$. This wavenumber, and the scale factor, evolve as power laws in cosmic time during matter or radiation domination. The ULA transfer function falls off more rapidly, as a cosine. Intuitively, this can be understood from the Jeans scale: solutions to a harmonic equation transition from exponential growth to harmonic oscillations when the growth exponent changes from real to imaginary.

Note that the WDM mass used here, and throughout this review, m_X , is the "thermal relic mass," which can be mapped to the larger mass of a sterile neutrino with the same free streaming scale [169, 177]:

$$m_{\nu,\text{sterile}} = 4.43 \text{ keV} \left(\frac{m_X}{\text{keV}}\right)^{4/3} \left(\frac{0.12}{\Omega_W}\right)^{1/3} . \tag{114}$$

The characteristic scale in the WDM transfer function is fixed by α^{-1} , while in the axion transfer function it is fixed by the Jeans scale at equality, $k_{J,eq}$. Note that for axions scale-dependent growth is still important on scales $k > k_J(z)$, and the transfer function Eq. (109) is only valid for smaller wavenumbers. The mild redshift dependence of $k_J \propto a^{1/4}$ means that the current Jeans scale is not so far separated from $k_{J,eq}$ (see Eq. 101).

A very rough estimate for when structure suppression is relevant on the same scales for WDM and ULAs can be obtained in the following way. For ULAs, assume that structure is suppressed for modes inside the horizon at a_{osc} , while

 for WDM assume the same for the temperature at which particles became nonrelativistic, $T_{\text{non. rel.}}$. Furthermore, assume for both that this happened during the radiation dominated epoch. If these transitions happened at the same time for WDM and ULAs, they will each suppress structure on the same scale relative to CDM. Taking $T_{\text{non. rel.}} \sim m_X$ and $H(a_{\text{osc.}}) \sim m_a$, and using that during the radiation dominated epoch $T \sim \sqrt{HM_{pl}}$ gives that WDM suppresses structure on the same scales as a ULA if:

$$m_X \sim \sqrt{m_a M_{pl}} = 0.5 \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{0.5} \text{ keV} \quad (\text{approximate match}).$$
(115)

We see that it is the large value of M_{pl} that generates the huge separation of mass scales between ULAs and WDM in their effects on structure formation.

A more precise relation between m_X and m_a can be obtained using the transfer functions Eqs. (108) and (109). The FCDM transfer function falls off more rapidly than the WDM transfer function, so first we must define a scale at which to match them. We can take this to be the half-mode, $k_{1/2}$, defined by $T(k_{1/2}) = 0.5$. For the FCDM transfer function the half-mode is [176]:²⁸

$$k_{1/2} \approx 5.1 \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{4/9} \text{ Mpc}^{-1}.$$
 (116)

Matching this to the WDM half-mode gives:

$$m_X = 0.84 \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{0.39} \text{ keV} \quad \text{(half-mode matching)}. \tag{117}$$

This agrees with the fit found using ULA transfer functions computed from AXIONCAMB in Ref. [178], and also agrees surprisingly well with the simple estimate of Eq. (115).

Transfer functions for WDM and ULAs, with the WDM mass computed using Eq. (117), are shown in Fig. 10. The lowest mass shown is $m_a = 10^{-23} \text{eV} \rightarrow m_X = 0.34 \text{ keV}$, and has $k_{1/2} = 1.6 \text{ Mpc}^{-1}$. The non-linear scale is $k_{\text{nl}} \sim 0.1 \rightarrow 1 \text{ Mpc}^{-1}$, and so we see that power suppression by $m_a \geq 10^{-23} \text{ eV}$ cannot be constrained by linear LSS observables.

²⁸We define the half mode using T(k) rather than $T^2(k)$ as Ref. [176] does, which explains the different co-efficient. Thanks to H. Y. Schive for noticing this.


Figure 10: ULA (solid) and WDM (dashed) transfer functions, Eqs. (108) and (109). WDM mass is computed to give the same value of $k_{1/2}$, using Eq. (117).

As a common reference, CDM composed of a neutralino with mass $m_X = 100 \text{ GeV}$ and decoupling temperature T = 33 MeV cuts off power due to free streaming at $k \approx 10^6 \text{ Mpc}^{-1}$ (e.g. Ref. [179]). Using Eq. (117) this is approximately the same scale as an axion with $m_a \approx 10^{-10} \text{ eV}$, and the QCD axion with $f_a \approx 10^{16} \text{ GeV}$. Thus, low- f_a QCD axions in the classic window suppress structure formation on scales *smaller* than standard WIMPs.

4.5. Non-linearities and the Schrödinger Picture

To study the clustering of axions on non-linear scales, we need to make some approximations. Axions that cluster on galactic scales began oscillating in the very early Universe, with $a_{\rm osc} \ll 1$, so we can take the WKB approximation. The virial velocity in a typical galaxy is $v_{\rm vir} \sim 100 \text{ km s}^{-1} \ll c$, and galaxies are much smaller than the horizon, so we can take the non-relativistic approximation. Overdensities in galaxies are $\delta \gtrsim \mathcal{O}(10^5)$, so perturbation theory on δ_a or ϕ is no good. However, except in the vicinity of black holes, the Newtonian potential is small, $\Psi \ll 1$. Thus the Newtonian limit is appropriate, and Ψ obeys the Poisson equation. We will also only be concerned with scales above the axion Compton wavelength (which is on relativistic scales in the Klein-Gordon equation).

To leading order in Ψ the D'Alembertian is

$$\Box = -(1 - 2\Psi)(\partial_t^2 + 3H\partial_t) + a^{-2}(1 + 2\Psi)\nabla^2 - 4\dot{\Psi}\partial_t, \qquad (118)$$

and the axion energy density is

$$\rho_a = \frac{1}{2} [(1 - 2\Psi)\dot{\phi}^2 + m_a^2 \phi^2 + a^{-2}(1 + 2\Psi)\partial^i \phi \partial_i \phi].$$
(119)

We take the WKB approximation in the form

$$\phi = (m_a \sqrt{2})^{-1} (\psi e^{-im_a t} + \psi^* e^{im_a t}), \qquad (120)$$

where ψ is a complex field, which can be written in polar co-ordinates as

$$\psi = Re^{iS} \,. \tag{121}$$

We take our limits as $\Psi \sim \epsilon_{\rm NR}^2$, and $k/m_a \sim \epsilon_{\rm NR}$ and $H/m_a \sim \epsilon_{\rm WKB}$, and work to quadratic order in $\epsilon \sim \epsilon_{\rm NR} \sim \epsilon_{\rm WKB}$. In this limit, the energy density contains the leading order piece:

$$\rho_a = |\psi|^2 = R^2 \,, \tag{122}$$

and the equation of motion for ψ is the Schrödinger equation:

$$i\dot{\psi} - 3iH\psi/2 + (2m_a a^2)^{-1}\nabla^2\psi - m_a\Psi\psi = 0.$$
 (123)

This is a non-linear Schrödinger equation, with Ψ sourced by $|\psi|^2$ via the Poisson equation. The form shown here, including the Hubble friction explicitly, can be found from the usual form by going to comoving coordinates.

While the Schrödinger equation is interesting and can provide insight into structure formation with axion DM, wave equations don't fit the bill as standard cosmologist's tools. We can make contact with standard perturbation theory [180] and non-linear simulation tools such as smoothed-particle hydrodynamics

(SPH) using, as before, a fluid description. Substituting the polar form of the wavefunction, we can find conservation and Euler equations for an effective fluid described by ψ . The fluid velocity is

$$\vec{v}_a \equiv (m_a a)^{-1} \nabla S \,. \tag{124}$$

We can now perform a background-fluctuation split and find the equations of motion in terms of the overdensity, δ_a (e.g. Refs. [181, 182]):

$$\dot{\delta}_a + a^{-1} \vec{v}_a \cdot \nabla \delta_a = -a^{-1} (1 + \delta_a) \nabla \cdot \vec{v}_a , \qquad (125)$$

$$\dot{\vec{v}}_a + a^{-1} \left(\vec{v}_a \cdot \nabla \right) \vec{v}_a = -a^{-1} \nabla (\Psi + Q) - H \vec{v} , \qquad (126)$$

$$Q \equiv -\frac{1}{2m_a^2 a^2} \frac{\nabla^2 \sqrt{1+\delta_a}}{\sqrt{1+\delta_a}} \,, \tag{127}$$

where I have defined the "quantum potential" Q^{29} The quantum potential is all we need to model the axion gradient energy and Jeans scale in the full non-linear dynamics as a simple modification to the force on a fluid element [182, 184]:

$$F = -a^{-1}\nabla(\Psi + Q).$$
(128)

Eqs. (125) and (126) can also be used as the basis for a modified perturbation theory of axion DM, which takes into account the differences to CDM near the Jeans scale. Expanding Eq. (127) to first order in δ_a and going to Fourier space provides a simple derivation of the asymptotic form of the effective sound speed, Eq. (100).

The Schrödinger form of the field equations is useful and interesting in and of itself. It is a fundamental (though approximate) equation governing axion DM on non-linear scales. We will use the Schrödinger equation to discuss axion halo density profiles in Section 6.3. Above the de-Broglie wavelength Schrödinger

²⁹We have used the Schrödinger equation as an intermediate step to get a fluid form for the axion equations without needing to perform the background-fluctuation split on ϕ first. We were thus able to retain canonical equations of motion for ρ and \vec{v} beyond linear perturbation theory. For discussion on the use of hydrodynamics to describe quantum mechanics in the "synthetic" view of Bohmian mechanics, see Ref. [183].

equation also accurately models CDM, and is an alternative to standard N-body simulation techniques [185]. The wave properties below the de Broglie scale and the introduction of the quantum force in the fluid equations are a particular regularization and softening of the Vlasov equation [186]. They also provide a setting to study modifications to the Zel'dovich approximation [187, 181], which is the basis of Lagrangian perturbation theory. Perhaps most importantly, however, the Schrödinger equation provides the best method currently available to accurately simulate axion and scalar field DM on small scales, which we will now discuss.

4.6. Simulating axion DM

A full description of DM clustering in any model can only truly be provided by non-pertrubative numerical simulations. Since the earliest days of computational cosmology, this been studied in N-body simulations, which simulate the dynamics of collisionless point particles interacting via Newton's gravitational law. The "particles" are not fundamental particles, but simulations particles, the mass of which is fixed by the simulations resolution. Newton's law is "softened" on small scales to prevent unphysical two-body pairs of these particles dominating the dynamics. These classic N-body simulations are the perfect picture of CDM, and their conceptual simplicity provides some explanation for the popularity of its study.

A simulation of CDM is defined by two properties: initial conditions, and dynamics. The initial conditions are provided by the matter power spectrum from linear theory, with higher order effects to deal with transients [180]; the dynamics is that of collisionless particles. Axions, particularly ULAs, modify both of these properties:

- *Modified initial conditions*: The initial power spectrum is suppressed relative to CDM. Modes below the Jeans scale at matter-radiation equality have the power erased.
- Modified dynamics: On scales of order the axion de Broglie wavelength,

wavelike effects must be included. The dynamics is not that of collisionless point particles.

Modified initial conditions are easily implemented in an *N*-body simulation, as long as the correct power spectrum is provided from a Boltzmann code. Such simulations provide an accurate description of axions above the de Broglie wavelength, and have been performed in Refs. [188, 189]. These simulations are very similar to those of WDM in the case that streaming velocities are irrelevant (e.g. Refs. [170, 190]). Special care must be taken, however, due to the appearance of "spurious structures" caused by discreteness effects [191]. Such spurious structures can be removed based on the shape of the protohalos [190] or on the functional shape of the halo mass function [192]. Removal of spurious structure for ULAs was carried out using the protohalo shape condition in Ref. [188]. We will discuss the halo mass function in more detail in Section 6.1.

Modified dynamics are somewhat less trivial to implement, in particular those relevant to ULAs. Modern simulations add new dynamics to the simplest CDM model such as hydrodynamics of the baryons (e.g. Ref. [193]), parameterised force law modifications for variants of SIDM [194], neutrino models with streaming velocities [195, 196], and even general relativistic effects [197] or modified gravity [198]. At their core, all these methods are based, to some degree, on the N-body paradigm.

As long as the objects to be simulated are non-relativistic (as galactic halos are), the Schrödinger equation provides the correct model of axion DM on small scales. A cosmological simulation of the Schrödinger equation is a fundamental departure from N-body simulations. The first high-resolution cosmological simulations of the Schrödinger form were recently performed in Ref. [199]. The modified dynamics caused by wavelike effects for $m_a \approx 10^{-22}$ eV appear in dwarf galaxy-sized objects on scales of order 1 kpc. The modified dynamics can be seen to introduce effects including smooth halo density profiles and interference fringes (see Section 6.3 and Fig. 20), which would be completely absent in a CDM-like N-body simulation. Resolving these features accurately in a cosmological simulation involves many computational technicalities, including e.g. the use of adaptive mesh refinement to solve the scalar field equation over a wide range of length scales.

An alternative way to model the modified dynamics of ULAs and other scalar fields in cosmological simulations, which fits more easily into the *N*-body paradigm, is suggested by the modified force law in the fluid description, Eq. 128. This modified force law could be implemented in a hydrodynamic model (as suggested in Refs [182, 184]), or indeed in any method where the local density and its derivatives can be accurately determined. This method was employed in toy models in Ref. [184], but has yet to be applied to a cosmological simulation.

The Schrödinger equation in this context models more than just axions. It is applicable to any model of scalar field DM, real or complex-valued, so long as the field is oscillating about a quadratic potential minimum, and self-interactions can be neglected. The simulations of Ref. [199] represent the state-of-the-art for simulations of these models. There is still much to be done in this area, however. For example, some of the many things not covered in Ref. [199]:

- Initial conditions. Use of full Boltzmann equation power spectra. Modified perturbation theory and Zel'dovich approximation.
- Hydrodynamics. Modelling of baryonic effects in tandem with scalar field dynamics to assess complementary roles.
- Zoom-in simulations. Dwarf galaxies and sub-structure modelled in Milky-Way and Local Group analogs from larger *N*-body simulations.

This shopping list is not meant to detract from the achievements of Ref. [199]: the field of study of such simulations is simply young compared to that of CDM N-body simulations.

4.7. My Two Cents on BEC

In this section we discuss *only* DM axions. There is some debate in the literature as to whether axion DM forms a Bose-Einstein condensate (BEC),

and over what scales such a BEC differs from CDM. For more discussion on this topic, see Refs. [200, 201, 202, 203]. The original discussions of the link between quantum theory and classical fields for the axion are in Refs. [204, 205].³⁰

Davidson $\left[202\right]$ defines a BEC as

$$BEC = condensed regime = classical field.$$
(129)

This chimes with our usual notion form undergraduate statistical mechanics: the macroscopically-occupied ground-state obeys the classical equations of motion. The important characteristic, however, is not the ground-state, which is only accessible to a homogeneous system (which cosmology certainly is not), but it is that the Fourier modes are concentrated at a particular value and that the particles in this state are coherent.

Let's define some of these notions: we will not use these formal definitions, but it helps to be precise. QFT decomposes a field operator into modes of creation, \hat{a} , and annihilation, \hat{a}^{\dagger} , operators as

$$\hat{\phi}(x) = \int \frac{d^3p}{(2\pi)^3} \frac{1}{\sqrt{2E_p}} \left(\hat{a}_p e^{-\vec{p}\cdot\vec{x}} + \hat{a}_p^{\dagger} e^{-i\vec{p}\cdot\vec{x}} \right) \,, \tag{130}$$

where \vec{p} is the three-momentum, and E_p is the energy. The ground state is defined by $\hat{a}_p |0\rangle = 0$. The classical field is defined by the coherent state [207]

$$|\phi\rangle = \frac{1}{N} \exp\left[\int \frac{d^3q}{(2\pi)^3} \tilde{\phi}(\vec{q}) \hat{a}_q^{\dagger}\right] |0\rangle, \qquad (131)$$

where $\tilde{\phi}(\vec{q})$ is the Fourier transform of the classical field, and N is a normalisation such that $\langle \phi | \phi \rangle = 1$. The expectation value of the field operator in this state is the classical field:

$$\langle \phi | \hat{\phi}(x) | \phi \rangle = \int \frac{d^3 p}{(2\pi)^3} \frac{1}{\sqrt{2E_p}} \tilde{\phi}(\vec{p}) e^{-i\vec{p}\cdot\vec{x}} = \phi(x) , \qquad (132)$$

³⁰There is a vast literature on so-called "BEC dark matter": as far as I can tell, for all practical purposes this simply maps to general scalar field models. Since the early Universe physics is often less well defined than in the case of axions, questions of condensate formation are also less clear. For a good source of references and history, see Ref. [206].

i.e. this expectation value obeys the classical equations of motion as we have been discussing in the preceding subsections, and will continue to discuss throughout this review.

The questions now are: over what timescales do axions enter the state $|\phi\rangle$, how does this state evolve, and, crucially, what is its coherence length? The "controversy" of axion BEC is over what role gravity plays in this process, particularly at late times, and over the coherence length this induces for structures with vorticity, $\vec{\nabla} \times \vec{v} \neq 0$, within galaxies.

Recall that there are two populations of DM axions: those formed from vacuum realignment, and those formed from decay of topological defects. The vacuum realignment population begins life already in the state $|\phi\rangle$. In the broken PQ scenario, the state $|\phi\rangle$ is formed by inflation, which super-cools and homgenises the axion field over the entire visible Universe. In the unbroken PQ scenario, the parent PQ field, φ , is in it is classical field state, $|\phi\rangle$, and thus the axion field created after SSB is also coherent in the state $|\phi\rangle$ over the horizon size at SSB (leading to the classical field configurations of strings, domain walls, and miniclusters, as discussed above).

Thus, for either the broken or unbroken PQ scenario, axions from the vacuum realignment mechanism are described entirely by the classical field equations, as presented in the preceding parts of this section. Thermalisation at early times is irrelevant, as coherence is established by initial conditions. The gravitational interactions lead to the usual structure formation on large scales: as perturbations grow, the field effectively loses some coherence. The Jeans scale supports the field against gravitational collapse and maintains total coherence on smaller scales. The characteristic size of collapsed objects is given by the soliton solutions to the Schrödinger-Possion equation (see Section 6.3).

For the population of cold axion particles produced by topological defect decay in the unbroken PQ scenario, axions can enter the state $|\phi\rangle$ via thermalisation. The condition for thermalisation due to any interactions is that the relaxation rate, Γ , is of order the Hubble rate.

Consider the QCD axion for concreteness. The self interactions are computed

by Taylor expanding the cosine potential, giving:

$$V_{\rm int} = \frac{\lambda}{4!} \phi^4 \,, \quad \lambda = \frac{m_a^2}{f_a^2} \frac{m_d^3 + m_u^3}{(m_d + m_u)^3} \approx 0.35 \frac{m_a^2}{f_a^2} \,. \tag{133}$$

Note that these interactions are *attractive*. The relaxation rate is [200]

$$\Gamma_{\lambda} \sim n\sigma_0 \delta v \mathcal{N} \,. \tag{134}$$

where n is the number density of particles, σ_0 is the cross section for two-to-two axion scattering in vacuum, $\sigma_0 = \lambda^2/(64\pi m_a^2)$, δv is the velocity dispersion, and \mathcal{N} is the average state occupation number. The number density is computed from the relic density, the velocity dispersion at time t is computed by redshifting the initial momentum, $p(t_{\rm osc}) \approx H(t_{\rm osc})$ (recall that topological defects decay when the classical field begins oscillating), and the occupation number is given by

$$\mathcal{N} = \frac{(2\pi)^3 n}{V_{\text{coh.}}} \,, \tag{135}$$

where $V_{\text{coh.}}$ is the spherical volume of a coherence patch: $V_{\text{coh.}} = 4\pi (m\delta v)^3/3$.

By taking $m_a t_{\rm osc} \sim 1$ we find that $\Gamma_{\lambda}(t_{\rm osc})/H(t_{\rm osc}) \sim \mathcal{O}(1)$: self interactions thermalise the cold population of axions, with an initial coherence length of order $1/H(t_{\rm osc})$.³¹ Thus, for the cold population of axions produced by topological defect decay, on all times later that $t_{\rm osc}$ we can also describe the axions as being in the state $|\phi\rangle$ obeying the classical equations of motion. This is as we expect: occupation numbers for axion DM from any production mechanism are so huge that classical field equations ought to be adequate. So far, so uncontroversial.

The question now arises as to whether axions can "re-thermalise" at later times. The two-to-two rate, Γ_{λ} , redshifts faster than H, such that at times after $t_{\rm osc}$ self-interactions are not sufficient for this purpose [200]. Now the controversial part: can gravitational interactions re-thermalise the axion condensate? If re-thermalisation at times $t > t_{\rm osc}$ occurs, then a larger coherence length will be established, and axion DM will differ from CDM on scales larger than those set by the Jeans scale and quantum pressure in the classical equations of motion.

³¹The general scalings of these arguments hold also for generic ALPs with $\lambda \sim m_a^2/f_a^2$.

Sikivie and Yang [200] propose that gravitational scattering of axions can lead to re-thermalisation of the QCD axion at a temperature $T_{re.} \sim 100 \text{ eV} (f_a/10^{12} \text{ GeV})^{1/2}$. This is argued based on the gravitational relaxation rate:

$$\Gamma_G = \frac{G_N n m_a^2}{(m \delta v)^2} \,. \tag{136}$$

If Sikivie and Yang are correct, this effect will induce a larger coherence length for the axion field, absent in the classical equations of motion. In particular, the claim is that re-thermalization due to Γ_G is *not* captured by the classical equations of motion.

However, this claim has been countered by Davidson and Elmer [201], Davidson [202], and Guth, Hertzberg and Prescod-Weinstein [203], who show that the effects of the relaxation rate Γ_G are already present in the classical equations of motion (the relevant case being the Schrödinger-Poisson equation), and thus by solving them alone on times $t > t_{\rm osc}$ we miss nothing: there is no coherence on scales larger than the Jeans scale. The rate Γ_G is the interaction rate between axions already in the condensate with one another, hence being linear in G_N . Davidson [202] also estimated the quadratic in G_N scattering between cold axion particles and the condensate, concluding that this interaction is negligible for $f_a \leq M_{pl}$.³² In the end, Davidson notes, all such questions can ultimately be answered by the Path Integral, using the Closed Time Path 2PI action in curved space. Further treatment of this is far beyond the scope of this review.

A final note here is on the possible formation of vortices in the axion field (a well-known phenomenon in BEC in the laboratory [208]), and their possible phenomenological role in galactic haloes. A net overall rotation of the axion field caused by tidal torques leading to $\nabla \times \vec{v} \neq 0$ would augment our system of classical equations due to anomalous stresses, and could lead to vortex formation. Sikivie and Yang (see also Ref. [209]) argued that this could be a distinctive feature of axion DM, and may explain the structure of caustics in DM

 $^{^{32}\}mathrm{Recall}$ that it is folk-wisdom that super-Planckian f_a violates "gravity as the weakest force"

haloes. This was explored in more detail by Rindler-Daller and Shapiro [210], who found that the axion self-interactions are of the wrong type (attractive rather than repulsive) to support vortex formation. Vortex formation depends on having self-interactions, and so goes beyond the $m_a^2 \phi^2$ simplified model we study mostly in this review. In any case, it is clearly a model-dependent effect, and one that appears not to occur for the QCD axion.

5. Constraints from the CMB and LSS

This section reviews work presented in Refs. [131, 172, 211, 212]. Bayes theorem is briefly reviewed in Appendix E. Issues related to sampling the axion parameter space are discussed in Appendix F.

5.1. The Primary CMB

The CMB temperature auto-power, C_{ℓ}^{TT} , is the data product at the disposal of the precision cosmologist. We use CMB data from *Planck* (2013 release) [213, 214] and WMAP [215], ACT [216] and SPT [217].

ULAs affect the primary (adiabatic, unlensed, no secondaries) CMB primarily via the expansion rate. The first acoustic peak of the CMB temperature power occurs at $\ell \approx 200$ and is fixed by the angular size of the BAO at recombination, $z_{\rm rec} \approx 1100$. ULAs with $z_{\rm osc} \gtrsim 1100$ affect higher acoustic peaks, while those with $z_{\rm osc} \lesssim 1100$ affect the Sachs-Wolfe (SW) plateau.

The CMB acoustic peaks constrain the relative matter-to-radiation density at different epochs, fixing the DM to baryon ratio and the redshift of matterradiation equality. Axions with $w_a \approx -1$ at any particular epoch alter the expansion rate relative to that in a pure CDM cosmology. The higher acoustic peaks probe successively higher order effects on the expansion at earlier times, however radiation is increasingly dominant at early times, and the higher acoustic peaks also Silk-damp away. Thus, there is some maximum $z_{\rm osc}$ for heavy ULAs beyond which the effects on the higher acoustic peaks vanish and ULAs become indistinguishable from CDM. If we demand that ULAs compose all the



Figure 11: Effect of ULAs on the CMB as a function of ULA mass. Here we demand that ULAs compose all the DM, with no CDM. The early-time expansion rate is altered, changing the relative heights of the higher acoustic peaks. Reproduced (with permission) from Ref. [131]. Copyright (2015) by The American Physical Society.

DM, the effects on the CMB are more dramatic for low mass ULAs, where the expansion rate is significantly altered near matter-radiation equality. These effects are illustrated in Fig. 11. The lightest ULA model shown has $m_a = 10^{-27}$ eV. The mass is just large enough that matter-radiation equality and recombination are barely changed, leaving the first peak at the same location, and the SW plateau unchanged. Higher acoustic peaks depart significantly from the CDM case. Increasingly higher masses lead to increasingly smaller effects away from CDM, with the effects moving to higher acoustic peaks. By eye, it is impossible to distinguish $m_a = 10^{-25}$ eV from CDM.

Lighter ULAs differ significantly from CDM in the post-recombination Universe. Getting matter-radiation equality right requires us to keep the CDM density at $\Omega_c h^2 = 0.12$. Introducing light ULAs at fixed H_0 thus reduces Ω_{Λ} . The Universe is now younger, with reduced distance to the CMB. This moves the first acoustic peak to lower ℓ . The ULAs have $w_a = -1$ transitioning to

ACCEPTED MANUSCRIPT



Figure 12: Effect of the lightest ULAs on the CMB. Left Panel: I hold $\Omega_c h^2 = 0.12$ fixed and introduce successively heavier axions as a fraction of the DE at fixed H_0 . The first acoustic peak moves and the ISW effect more pronounced compared to Λ CDM. Right Panel: Here we demand that the location of the first peak remains fixed, which requires reducing H_0 compared to Λ CDM, isolating the ISW effect. Reproduced (with permission) from Ref. [131]. Copyright (2015) by The American Physical Society.

 $w_a = 0$ in the late Universe, and imprint this on the low ℓ CMB via the integrated (I)SW effect. Both of these effects are shown for varying ULA masses in Fig. 12 (Left Panel). Notice that $m_a = 10^{-33}$ eV is indistinguishable from Λ CDM: axions this light have $w_a \approx -1$ today, and contribute to the effective cosmological constant and DE.

The low ℓ CMB measurement is cosmic variance limited, leading to large uncertainties, while the first acoustic peak is measured exquisitely well. We can isolate the ISW effect of ULAs by changing the value of H_0 to leave the location of the first peak unchanged. Such a cosmology is shown in Fig. 12 (Right Panel). With $\Omega_a/\Omega_d = 0.1$ and $m_a = 10^{-32}$ eV the ULA model is indistinguishable from Λ CDM (except in the quadrupole, $\ell = 2$, which is poorly measured).

5.2. The Matter Power Spectrum

The matter power spectrum, P(k, z), contains a wealth of cosmological information. The BAO imprint a fixed physical scale on the power spectrum, and this is used as a measurement of the expansion rate (e.g. Ref. [218]). The BAO measure a single number, the angular size of the sound horizon, as a function of redshift. The full shape of the matter power spectrum contains more information than just the BAO, and is our focus here. The matter power spectrum can be measured from the two-point correlation function of some tracer of the DM. Here we focus on the galaxy power spectrum, $P_{\text{gal}}(k, z) = b^2 P(k, z)$, where b is the galaxy bias. It is measured by a number of surveys, of which we choose to use the WiggleZ survey [219], which measures the galaxy power spectrum in four redshift bins centred on z = 0.22, 0.41, 0.60 and 0.78. We further restrict to only linear scales, $k \leq 0.2 h \text{Mpc}^{-1}$.

The effect of axions on the matter power spectrum probes both the expansion rate (via the BAO) and the growth of structure, via the transfer and growth functions. The most well-known effect that we have already discussed is the suppression of power caused by the existence of the axion Jeans scale. This effect is shown in Fig. 13, where the left panel shows the idealized scenario with P(k), and the right panel the effect convolved with the WiggleZ survey window function and marginalized over galaxy bias.

In the idealized case, we see how reducing the axion contribution to the DM density reduces the amount of structure suppression compared to CDM [132, 171]. For $m_a = 10^{-27}$ eV structure suppression kicks in at $k \approx 0.02 h \,\mathrm{Mpc}^{-1}$, and has a sub-percent effect on the power relative to CDM for $\Omega_a/\Omega_d = 0.01$ (ULAs contributing ~ 1% to the total DM). The galaxy bias, b, changes the character of the effect. Galaxy bias is measured by the survey by allowing b to float as a free parameter. When it varies, it can compensate, in a scale-independent manner, for suppression of power. The preferred value of b, and so the normalization of the power spectrum, is thus different for the ULA cosmologies than for Λ CDM, and this partial degeneracy reduces the constraining power of the galaxy survey.



Figure 13: Effect of the lightest ULAs on the matter power spectrum, with fixed mass and varying contribution to the DM density. *Left Panel:* The matter power spectrum. *Right Panel:* After convolution with the WiggleZ survey window function and marginalization over galaxy bias at z = 0.60. Reproduced (with permission) from Ref. [131]. Copyright (2015) by The American Physical Society.

The scale-dependent clustering of ULAs tells us that a full treatment of bias in these cosmologies should involve computing a *scale-dependent bias*, b(k), and its dependence on the ULA transfer function and growth rate. Scaledependent bias in mixed DM cosmologies is a poorly understood problem, and it has particular relevance to studies of massive neutrinos (see e.g. Ref. [220]). Scale-dependent bias can be studied through numerical simulation, or semianalytically via the halo model [221]. Ref. [131] proposed an approximate treatment of scale-dependent bias for ULAs, motivated by treatments of DE and neutrinos, and by the data, which we now outline.

Bias relates the galaxy power spectrum to the matter distribution. On scales where ULAs do not cluster (below the Jeans scale), we do not expect any correlation between the galaxies and the ULAs. Galaxy surveys only observe out to some smallest wavenumber (largest scale), k_{obs} . The scale of the observations defines an epoch, $k_{obs} = a_{bias}H(a_{bias})$: ULAs which only begin to behave like matter after this epoch will not be correlated with the galaxy distribution on observable scales. We can approximate the scale of structure suppression for ULAs as $k_{\rm osc} = a_{\rm osc} H(a_{\rm osc})$ and impose scale-dependent bias as a hard cut by excluding ULAs from the matter density if $a_{\rm osc} > a_{\rm bias}$:

$$\delta\rho_m = \Theta(a_{\rm osc} - a_{\rm bias})(\delta\rho_c + \delta\rho_b) + \Theta(a_{\rm bias} - a_{\rm osc})(\delta\rho_c + \delta\rho_b + \delta\rho_a), \quad (137)$$

where $\Theta(x)$ is the Heaviside function, and ρ_m in the overdensity is defined in the same manner. Because no current galaxy surveys observe on scales larger than the horizon size at equality, Ref. [131] made the simplification $a_{\text{bias}} = a_{\text{eq}}$, which effectively removes ULAs from the matter distribution used to compute the galaxy power spectrum for $m_a \lesssim 10^{-27}$ eV.

An unbiased tracer of the matter distribution is provided by gravitational lensing. Upcoming surveys such as *Euclid* propose to measure the galaxy shear power spectrum [222], and could improve constraints on DM models considerably [172, 223, 224] if systematics can be controlled. The forecasted sensitivity to Ω_a of the lightest ULAs for a *Euclid*-like survey is shown in Fig. 14.³³ These optimistic forecasts for weak lensing show an increase in sensitivity of around a factor of ten compared to the galaxy redshift survey alone.

The effect of axions on the expansion rate is also seen in the power spectrum, and is particularly evident if axions replace Λ (although now the issue of bias becomes more complicated [131]). This changes the age of the Universe relative to Λ CDM, with a younger Universe having less time to grow structures, reducing the amplitude of P(k). In the CMB the effect of a younger Universe could be largely compensated by reducing H_0 ; in P(k) it can be compensated by changing the amplitude of primordial fluctuations, A_s . However, as both the CMB and P(k) share common parameters, no choice of A_s and H_0 can completely remove the effects of this change, demonstrating the complementarity of CMB and LSS measurements. See Ref. [131] for further discussion.

ACCEPTED MANUSCRIPT



Figure 14: Forecasted sensitivity of a *Euclid*-like galaxy redshift (GRS) and weak lensing (WL) survey to axion DM fraction, Ω_a/Ω_d , as a function of mass. WL increases sensitivity to Ω_a by a factor of around ten compared to GRS alone. Reproduced and modified (with permission) from Ref. [172]. Copyright (2012) by The American Physical Society.

ACCEPTED MANUSCRIPT



Figure 15: Constraints to the axion dark sector energy fraction, Ω_a/Ω_d , as a function of axion mass from linear cosmological probes. *Left Panel:* Contours show 2 and 3 σ allowed regions comparing CMB and CMB+WiggleZ. *Right Panel:* CMB constraints, with sample points from chains colour-coded by axion initial displacement in Planck units. Reproduced (with permission) from Ref. [131]. Copyright (2015) by The American Physical Society.

5.3. Combined Constraints

Fig. 15 (left panel) shows the constraints on the axion dark sector density fraction, Ω_a/Ω_d , as a function of axion mass for CMB and CMB+WiggleZ data set combinations, taken from Ref. [131]. Including LSS data from WiggleZ as well as the CMB loosens constraints slightly at low mass, and tightens them slightly at high mass. The looser constraint at low mass is possibly being driven by the CMB/LSS tension in measurements of the power spectrum amplitude (commonly expressed as the " σ_8 tension"). The tighter constraint at high mass is due to the WiggleZ data points with small error bars at $k \sim 0.1 h \,\mathrm{Mpc}^{-1}$.

The normalization is $\Omega_d = \Omega_a + \Omega_c$, i.e. we consider a mixed DM model with CDM and ULAs. The allowed value at the lowest ULA masses, $m_a \approx 10^{-33}$ eV, is $\Omega_a/\Omega_d = 0.6$ implying $\Omega_a \approx 0.6$, with the CDM density held fixed at close to

 $^{^{33}}$ In this figure, neutrino parameters are included and marginalized over, lowering the CMB sensitivity compared to that found in Ref. [131] (see next section, and Appendix F).

its usual value. These ULAs are DE and drive the current period of accelerated expansion. At high mass, we see that in order for axions to be all the DM, with $\Omega_a/\Omega_d = 1$, requires $m_a \geq 10^{-24}$ eV at 95% C.L. This is the lower bound on DM particle mass from linear cosmological probes, as promised in the abstract. The constraint in the central, intermediate mass, region of 10^{-32} eV $\leq m_a \leq 10^{-25.5}$ eV is $\Omega_a/\Omega_d \leq 0.05$ and $\Omega_a h^2 \leq 0.006$ at 95%-confidence. That is, intermediate mass axions must make up less then 5% of the total DM.

It is important to note that the constraints of Ref. [131] apply to a cosmology with CDM plus a single light axion, and not to CDM plus multiple axions. It might be a good guess to assume that the constraint on the energy density in the intermediate mass regime applies to the sum total energy density for all such axions (because the constraint is independent of mass). A dedicated study is necessary, but degeneracies will be even more problematic and a prudent choice of priors and sampling will be required (see Appendix F).

Fig. 15 (right panel) shows the CMB only constraints, with sample points from MULTINEST [225] chains colour-coded by the initial axion field displacement in Planck units (and re-sampled such that point density is proportional to probability as in a Markov chain Monte Carlo, MCMC).³⁴ The field displacement is always $\phi_i < \pi M_{pl}$, and is thus consistent with a quadratic potential and sub-Planckian f_a . Axion DE requires $f_a \sim M_{pl}$. For $m_a = 10^{-22}$ eV to be all the DM requires $\phi_i \sim \mathcal{O}(\text{few}) \times 10^{16}$ GeV. This shows that a ULA with $f_a \leq 10^{16}$ GeV will satisfy all current constraints on Ω_a without fine tuning. These conclusions from numerical computation and full comparison with CMB data agree with the discussion in Section 4.3.1 based on Eq. 61.

³⁴The field displacement is found by using Eq. (61) as the initial guess in a shooting method to obtain the desired Ω_a . We solve the Klein-Gordon equation at early times, switching to $\rho_a \propto a^{-3}$ when $3H = m_a$.

ACCEPTED MANUSCRIPT



Figure 16: CMB adiabatic and isocurvature spectra. ACDM adiabatic (dashed), CDM isocurvature with $\Omega_a/\Omega_d = A_I/A_s = 1$ (dot dashed), ULA isocurvature with $\Omega_a/\Omega_d = 0.01$ and increasing m_a from left to right (solid, colour). Reproduced (with permission) from Ref. [211]. Copyright (2013) by The American Physical Society.

5.4. Isocurvature and Axions as a Probe of Inflation

Axions in the broken PQ scenario pick up isocurvature perturbations. The amplitude of these perturbations is proportional to the energy scale of inflation. The CMB places strong constraints on the allowed amplitude of such perturbations. Therefore, if axions compose the DM, constraints on isocurvature constrain the energy scale of inflation, and a detection of both would uniquely probe inflation. An independent measurement of the energy scale of inflation can be used to place strong constraints on axion cosmology.

Let's flesh these ideas out and quantify the possibilities. All of this Section assumes standard, single-field, slow-roll inflation. We'll focus on the QCD axion, which is also covered in detail in Refs. [134, 226, 141]. The case of ALPs is slightly more complicated than for the QCD axion, as the parameter space has more dimensions. ALPs are covered by Refs. [211, 212, 66].

Axion isocurvature density perturbations are of uncorrelated CDM type, as

long as the Jeans scale can be neglected, which is the case for the QCD axion. The isocurvature CMB spectrum is shown in Fig. 16, where the effect of nonnegligible ULA Jeans scale is also shown. The isocurvature power spectrum generated by Eq. (42) is:

$$P_I = A_I \left(\frac{k}{k_0}\right)^{1-n_I} , \qquad (138)$$

with amplitude

$$A_{I} = \left(\frac{\Omega_{a}}{\Omega_{d}}\right)^{2} \frac{(H_{I}/M_{pl})^{2}}{\pi^{2}(\phi_{i}/M_{pl})^{2}}.$$
(139)

The scalar power is:

$$P_{\zeta} = A_s \left(\frac{k}{k_0}\right)^{1-n_s},\tag{140}$$

with amplitude

$$A_s = \frac{1}{2\epsilon_{\inf}} \left(\frac{H_I}{2\pi M_{pl}}\right)^2 = 2.20 \times 10^{-9} \,. \tag{141}$$

The measured value of A_s is taken from *Planck* (2015), and the scalar spectral index is measured to be $n_s = 0.96$ [105]. Uncorrelated CDM isocurvature is constrained to³⁵

$$\frac{A_I}{A_s} < 0.038$$
 . (142)

The tensor-to-scalar ratio, $r_T = 16\epsilon_{inf}$, provides an independent constraint on the energy scale of inflation. *Planck* and BICEP2 [86] provide the limit $r_T < 0.12$. The projected sensitivity of CMB-S4 experiments is $r_T \sim 10^{-3}$ [229], while futuristic sensitivity from 21cm lensing could be as low as $r_T \sim 10^{-9}$ [230, 231].

All of these results are collected together for the QCD axion in Fig. 17. I plot contours for $A_I/A_s = 0.04$ and $\Omega_a h^2 = 0.12$ as functions of (f_a, H_I)

³⁵This assumes scale invariance of the isocurvature power, $\epsilon \ll 1$, which is consistent with the implied value of H_I and r_T . Compare this to the isocurvature power generated in the unbroken PQ scenario. In this case the amplitude is huge, $A_I \sim \langle (\delta\theta/\theta)^2 \rangle \sim \mathcal{O}(1) \gg A_s$, but power is only generated on very small scales, $k \gg k_0$, that are not constrained by the CMB power spectrum. Spectral distortions and miniclusters may impose interesting additional constraints [227, 228].



Figure 17: The QCD axion and CMB tensor modes. The unbroken PQ scenario produces no isocurvature, and is allowed as long as the limits on r_T and relic density (see Fig. 7) are satisfied, requiring low f_a . In the broken PQ scenario, we show various levels of tuning: $\theta_{a,i} = 1$ (solid lines), $\theta_{a,i} = 10^{-2}$ (dashed lines), $\theta_{a,i} = 10^{-4}$ (dotted lines). Constraints are shown for relic density $\Omega_d h^2 < 0.12$ (blue, lie below-left) and isocurvature amplitude $A_I/A_s < 0.04$ (red, lie below). The observable range of $10^{-9} < r_T < 0.1$ is shown in purple, with a realistic near-future limit of $r_T = 10^{-3}$ given by the solid line at $H_I \sim 10^{13}$ GeV. The allowed regime if the QCD axion in the broken PQ scenario is to be all the DM is given by the intersections of the red and blue lines (black), which always lies below a detectable tensor mode.

at fixed levels of fine tuning on $\theta_{a,i}$. Satisfying relic density and isocurvature constraints requires being below the intersections of these curves. For example, having $\theta_{a,i} = 1$ and $\Omega_a h^2 = 0.12$, requires $f_a \approx 3 \times 10^{11}$ GeV. The isocurvature constraint then enforces $H_I \leq 10^7$ GeV. The intersection of the A_I and Ω_a constraints traces out, approximately, $H_I \leq 10^{10} (f_a/M_{pl})^{1/2}$ GeV if axions are to be all the DM. The constraint quoted by *Planck* [96] in this scenario is $H_I < 0.86 \times 10^7 \text{ GeV} (f_a/10^{11} \text{ GeV})^{0.408}$ (95% C.L.), consistent with our rough estimates.

A range of measurably-large values of r_T are shown shaded purple, corresponding to a range $10^{10} \text{ GeV} \lesssim H_I \lesssim 10^{14} \text{ GeV}$. There is nowhere on the (f_a, H_I) plane where the QCD axion in the broken PQ scenario can be all of the DM, satisfy iscourvature bounds, and produce $r_T > 10^{-9}$ (a realistically observable value, shown by the dark purple line). Note that such small values of r can be obtained, consistent with A_s and n_s observations, in string inflation scenarios such as KKLT [232] or brane inflation (see Ref. [233] for details).

Relaxing the assumption that the QCD axion is all the DM, Fig. 17 shows that with $\theta_{a,i} \leq 10^{-4}$ a range of large f_a starts to become consistent with $r_T > 10^{-9}$. By trial and error, we find the maximum value of r_T consistent with isocurvature constraints and $f_a < M_{pl}$ occurs for $\theta_{a,i} \approx 10^{-7}$ where we have $\Omega_a h^2 < 10^{-6}$ and $r_T \approx 10^{-4}$. There is no amount of tuning that can make the QCD axion in the broken PQ scenario consistent with tensor modes as large as $r_T = 10^{-3}$, the CMB-S4 target.

CDM-type isocurvature modes are avoided completely in the unbroken PQ scenario. Thus, if tensor modes are observed, the QCD axion must live in the parameter space of Fig. 7 contained within the grey shaded region of Fig. 17, implying $f_a < 10^{11}$ GeV.

These conclusions can be avoided if some of our cosmological assumptions are relaxed. An example non-minimal inflation model producing $r_T > 10^{-3}$ consistent with the broken PQ scenario and high f_a , uses the radial PQ field, χ , as the inflaton, non-minimally coupled to gravity (similarly to Higgs inflation) [234]. Such a scenario can allow for simultaneous detection of DM axions by CASPEr [235] (see Section 9.5.2), and detection of r_T by, e.g., SPIDER [236]. There are many other possibilities to avoid the isocurvature problem of highscale axions by modifying inflation, particle physics, or the thermal history, for example Ref. [237], and related works.

We conclude our discussion of the QCD axion and isocurvature in summary:

- The QCD axion in the broken PQ scenario is incompatible with observablylarge tensor modes from standard inflation.³⁶
- In the broken PQ scenario with standard inflation, axion isocurvature modes could probe H_I as low as 10⁷ GeV, offering a unique probe of lowscale inflation.
- Simultaneously detecting a high $f_a \gtrsim 10^{13}$ GeV QCD axion and tensor modes at $r_T = 10^{-3}$ would falsify minimally coupled, single-field, slow-roll inflation with a standard thermal history.

6. Galaxy Formation

This section reviews work presented in Refs. [178, 238, 239].

6.1. The Halo Mass Function

The halo mass function (HMF) gives the expected number of halos per logarithmic mass bin, per unit volume, for a given cosmology. It depends fundamentally on two quantities, both of which can depend on halo mass and redshift: the variance of fluctuations, $\sigma^2(M, z)$, and the linearly extrapolated critical density required for such fluctuations to collapse, $\delta_{\rm crit}(M, z)$. The relevant standard formulae are given in Appendix G.

³⁶It is, in fact, possible to make the QCD axion in the broken PQ scenario compatible with observable tensors if we allow $f_a \gtrsim 10^{10} M_{pl}$ and tune the initial misalignment angle at a level $\theta_{a,i} \ll 10^{-10}$. I exclude such a scenario as unreasonable. The tuning is worse than the strong-*CP* problem, and the existence of a scale so much larger than the Planck scale is considered highly problematic in theories of quantum gravity.

We can compute $\sigma(M, z)$ given the linear power spectrum, P(k, z). The cutoff in power caused by the axion Jeans scale leads to a suppression of $\sigma(M, z)$ compared to CDM at low halo mass, with $\sigma(M, z)$ going to a constant as $M \to 0$. The reduced value of $\sigma(M, z)$ reduces the abundance of low mass halos.

In an Einstein-de Sitter universe (CDM with $\Lambda = 0$), spherical collapse can be solved exactly. Scale-independent growth gives a constant, mass-independent, value for δ_{crit} , which can be scaled to any redshift using the linear growth factor (the result also works well for Λ CDM on not-too-large scales):

$$\delta_{\rm crit, EdS}(z) = \frac{1.686D_0}{D(z)}.$$
 (143)

The collapse barrier is mass-independent for CDM because the growth equation is scale-invariant. In DM models with an effective pressure, the Jeans scale introduces scale-dependence into the collapse threshold. In spherical collapse simulations with WDM, where free-streaming was modelled by an effective pressure [240], a mass-dependent critical barrier is found, with $\delta_{\rm crit}$ increasing below the WDM Jeans scale. This barrier can then be used in a full excursion set model of WDM halo formation, dramatically suppressing halo formation below the effective Jeans mass [241]. Spherical collapse and the excursion set have not been studied for axion DM. Instead, Ref. [178] proposed a simple model where D(z) in Eq. (143) is simply replaced by an appropriately normalized (in both scale and redshift relative to Λ CDM) scale-dependent growth factor, \mathcal{G} . The mass can be assigned from the wavenumber using the enclosed mean density in a sphere of radius $R = \pi/k$ giving:

$$\delta_{\rm crit}(M, z) = 1.686\mathcal{G}(M, z). \tag{144}$$

We define \mathcal{G} as the relative amount of growth between axion DM and CDM, normalized to unity on large scales, k_0 , and at early times, z_{early} :

$$\mathcal{G}(k,z) = \frac{\delta_a(k_0,z)\delta_a(k,z_{\text{early}})}{\delta_a(k,z)\delta_a(k_0,z_{\text{early}})}\frac{\delta_c(k,z)\delta_c(k_0,z_{\text{early}})}{\delta_c(k_0,z)\delta_c(k,z_{\text{early}})},$$
(145)

where δ_a is computed in the axion cosmology, and δ_c is computed in the CDM cosmology, with $\Omega_a h^2 = \Omega_c h^2$. In practice, k_0 should be chosen such that



Figure 18: Right Panel: Critical overdensity for collapse, $\delta_{\rm crit}(M, z)$ for $m_a = 10^{-22}$ eV computed from scale-dependent growth using Eq. (144), normalizing for the growth in Λ CDM using D(z). Left Panel: Resultant halo mass function, compared to CDM. Modified from Ref. [238], Figs. 1 and 2.

 $k_0 < k_J(z_{\text{early}})$, but not so small such that scale dependent growth in Λ CDM due to Λ domination becomes relevant. Similarly, z_{early} should be chosen such that the power spectrum shape in Λ CDM has frozen in, i.e. after BAO formation. For DM axions in a close-to- Λ CDM cosmology, reasonable choices are $k_0 = 0.002 \,h\text{Mpc}^{-1}$ and $z_{\text{early}} \approx 300.^{37}$

The critical overdensity appears in the HMF in the argument of a Gaussian. Thus, even a modest increase in $\delta_{\rm crit}$ causes a sharp cut-off in the HMF: this is shown in Fig. 18.³⁸ The cut-off makes physical sense: there are no seed density perturbations on scales below the Jeans scale, and even if there were, growth is so suppressed there that density perturbations cannot collapse into virialized

 $^{^{37}}$ An interesting recent discussion of the relative importance of scale dependent growth to LSS simulations of axion DM is given in Ref. [188], where a similar quantity to \mathcal{G} is used to measure this.

³⁸The fact that the barrier appears in a Gaussian also renders the details of the barrier function, such as the acoustic features and smoothing scheme at masses much below the axion Jeans scale, largely irrelevant for halo statistics.

objects. At higher redshifts, when density perturbations are smaller, and the Jeans scale is larger, the effect is more pronounced. We learn that: ULAs dramatically suppress halo formation compared to CDM at low halo masses and at high redshifts.³⁹

For the QCD axion, the cut-off in the HMF induced by the Jeans scale is on extremely small scales $M < 10^{-9} M_{\odot}$ (c.f. the standard WIMP, where the smallest halos have mass $M \approx 10^{-6} M_{\odot}$ [179]). These smallest halos will certainly be tidally disrupted today, but are interesting to study the very first moments of structure formation at $z \approx 60$ in CDM models. Axion miniclusters produced in the unbroken PQ scenario for the QCD axion in the classic window have $M_{\rm mc} \approx 10^{-9} M_{\odot}$ [242]. Miniclusters of ALPs may be more, or less, massive. Being denser than ordinary halos, axion miniclusters survive to the present day and are relevant to observational searches for minihalos (e.g. Refs. [227, 243, 244]).

6.2. Constraints from High-z and the EOR

There is accumulating data about the high-z Universe. We see a number of very high redshift galaxies with Hubble Ultra Deep Field (HUDF, e.g. Ref. [245]). We also know that the intergalactic medium (IGM) is reionized by star formation. Reionization is known to be essentially complete by $z \sim 6$ (e.g. observation of Gunn-Peterson trough [246] in quasar spectra [247]). Furthermore, reionization of the IGM produces an optical depth to the CMB, which is constrained by a combination of large angle temperature and polarization correlation functions to be $\tau = 0.07$ –0.08 ± 0.02 (central value depends on dataset combinations in Ref. [105]).

³⁹There is some discussion and debate concerning the location and origin of the HMF cutoff in both WDM (filtering, spurious structure [191]) and CDM (baryonic effects) that I will not go into here. For axions, numerical simulations such as those of Ref. [199, 188], with the addition of hydrodynamics and star formation, are necessary in order to be more precise. For basic, semi-analytic results, the intuitive notion of a cut-off at the Jeans scale provided by scale-dependent growth is sufficient.

The suppression of halo formation at high-z by ULAs cannot be too severe, or else it would be inconsistent with these observations, producing too few highz galaxies to match HUDF and to efficiently reionize the IGM. Getting these things right places a lower bound on m_a if ULAs are to contribute significantly to the DM density. Ref. [238] investigated these bounds, following similar work on WDM in Ref. [248].

In order to obtain constraints from the HMF, one needs to relate the halo mass to the UV magnitude of the galaxy, $M_{\rm UV}$. This can be done by abundance matching [249, 250]. The luminosity function, $\phi_{\rm lum}(M_{\rm UV}, z)$, is fit and matched to the low-z observations. The integrated (cumulative) luminosity function is then matched by number count to the cumulative halo mass function: $\Phi_{\rm lum}(<$ $M_{\rm UV}, z) = n(> M_h, z)$. This chain of relations fixes $M_h(M_{\rm UV})$. Therefore, once the low redshift data are fixed, the high redshift value of $\Phi_{\rm lum}(M_{\rm UV}, z)$ can be predicted for a given DM model, and itself compared to observation. The cut-off in the HMF induced by the axion Jeans scale cuts off the $M_h(M_{\rm UV})$ relation at some brightest magnitude, leaving the function $\Phi_{\rm lum}(M_{\rm UV}, z)$ with no support at the faint end.

Fig. 19 (Left Panel) shows the predicted cumulative luminosity function for axion DM at z = 8. If ULAs are too light, or make up too much of the DM, it is impossible to match the observed HUDF UV luminosity. The model $m_a = 10^{-23}$ eV with $\Omega_a h^2 > 0.06$ is ruled out at $> 8\sigma$ by HUDF. The model $m_a = 10^{-22}$ eV with $\Omega_a h^2 = 0.12$ is consistent with HUDF, but only just: the UV luminosity function cuts off at $M_{\rm UV} \approx -18$, right where the constraint is. This model could be excluded by a JWST measurement of the faint-end luminosity function at $M_{\rm UV} \approx -16$ [251] if it were found to be consistent with the larger CDM value of $\Phi_{\rm lum}(M_{\rm UV}, z)$.

The UV luminosity function can also be used to predict the evolution of the ionization fraction, Q(z) (not to be confused with the quantum potential, also denoted Q). This involves a fair amount of astrophysical modelling, as described in e.g. Refs. [252, 238, 188]. The results are shown in Fig. 19 (Right Panel), with shaded regions showing modelling uncertainty. These results are broadly

ACCEPTED MANUSCRIPT



Figure 19: Left Panel: Cumulative UV luminosity at z = 8, here denoted Φ , in axion models using the abundance matching technique. Data: HUDF [245]. Dashed line: JWST reach [251]. Model numbers are different abundance matching procedures and DM composition. Models 1, 2, ULAs are all the DM. Models 3, 4, ULAs are half of the DM. Right Panel: Ionization fraction. Shaded regions cover model uncertainties. Only extreme edges shown for CDM. Reproduced from Ref. [238], Figs. 4 and 6.

consistent with the studies of Refs. [188, 189], where the underlying halo mass function was computed from N-body simulations with modified initial power spectra. Ref. [189] also used different methods to model the reionization field.

The ionization fraction gives the optical depth to redshift, $\tau(z)$, from the integral along the line of sight:

$$\tau(z) = \int_0^z dz' \frac{(1+z')^2}{H(z')} Q(z') \sigma_T \bar{n}_H (1+\eta_{\rm He} Y/4X) , \qquad (146)$$

where σ_T is the Thompson optical depth, \bar{n}_H is the mean comoving Hydrogen number density, Y = (1 - X) is the Helium fraction, X is the Hydrogen fraction, and η_{He} is the ionization state of Helium (see Ref. [238] for references and more details on these parameters). The optical depth to the CMB is $\tau(z_{\text{rec}} \approx 1100)$.

Ref. [238] found that, within the modelling uncertainty, all axion DM models with $m_a \ge 10^{-22}$ eV can reproduce a CMB optical depth consistent with observations, while $m_a = 10^{-23}$ eV cannot (though the tension for the lightest masses is slightly less with the revised, *Planck* 2015, value for the optical depth).

Thus the CMB optical depth excludes the lightest ULAs with $m_a \lesssim 10^{-23}$ eV from being all of the DM.

There is the opportunity in future to constrain axion DM with $m_a \sim 10^{-22}$ – 10^{-21} eV from the evolution of Q(z). The cut off in the HMF delays the formation of the first galaxies, and thus reionization occurs at lower redshift than in CDM. Once collapse has begun, structure builds up more rapidly for ULAs, and reionization completes in a smaller redshift window. These different reionization histories distinguish ULAs and CDM. For example, the amplitude of the kinetic Sunyaev-Zel'dovich effect [253] in the CMB is sensitive to the duration of reionization (e.g. Ref. [254]). This will be measured in the near future by Advanced ACTPol [255] and could distinguish $m_a \leq 10^{-21}$ eV from CDM [238].

The bottom line is that high-z constraints currently exclude $m_a = 10^{-23}$ eV from being all of the DM at high confidence, and $m_a = 10^{-22}$ eV is right on the edge of acceptability. The bounds are only approximate, as a lot of uncertain astrophysics is involved, but Ref. [238] covered a range of models and the lower limit on $m_a \gtrsim 10^{-22}$ eV is reliable by order of magnitude. Similar results were also found by Ref. [188], giving $m_a \ge 1.2 \times 10^{-22}$ eV (2σ). This is the current lower limit on DM particle mass from non-linear clustering. Future constraints on high-z galaxies, and on the mean redshift and duration of reionization, could improve this limit by some two or more orders of magnitude. A measurement of the large scale 21cm power spectrum could constrain ULA mass as high as $m_a \approx 10^{-18}$ eV [182].

6.3. Halo Density Profiles

N-body simulations of pure CDM indicate that halo density profiles have a universal shape, known as the Navarro-Frenk-White (NFW) profile [256]:

$$\frac{\rho_{\rm NFW}(r)}{\rho_{\rm crit.}} = \frac{\delta_{\rm NFW}}{r/r_s(1+r/r_s)^2} \,, \tag{147}$$

where δ_{NFW} is a function of the "halo concentration," commonly denoted as c, and r_s is the scale radius. The concentration is defined such that the virial

radius is $r_{\rm vir} = cr_s.^{40}$ Notice that the NFW halo is a smoothly varying power law, with $\rho \sim r^{-1}$ in the centre: the so-called 'cusp.'

A dwarf galaxy in Λ CDM with $M \sim 10^{10} M_{\odot}$ has peak circular velocity on the order of 50 km s⁻¹ at a radius of around 10 kpc. The de Broglie wavelength, $\lambda_{\rm dB} = 1/mv$, of a particle inside such galaxy is then

$$\lambda_{\rm dB} \ge 4 \times 10^{-2} \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{-1} \text{ kpc},$$
(148)

and for a ULA is non-negligible in terms of the galaxy size. Using that $v \sim M/r$ and $M \sim \rho r^3$, setting $\lambda_{\rm dB} = r$ we find that $\lambda_{\rm dB} \sim m_a^{-1/2} \rho^{-1/4} \sim r_J$ where r_J is the Jeans scale.

Let's work directly with the Jeans scale. Taking $r_J = 2\pi/k_J$ and simply scaling Eq. (101) to the halo density gives

$$r_J = 94.5 \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{-1/2} \left(\frac{\rho(r_J)}{\rho_{\text{crit.}}}\right)^{-1/4} \left(\frac{\Omega_a h^2}{0.12}\right)^{-1/4} \text{ kpc} \,. \tag{149}$$

This is a polynomial equation to be solved for r_J . Plugging in a typical overdensity of 10⁶ with $m_a = 10^{-22}$ eV gives $r_J \sim 3$ kpc. The ULA Jeans scale inside a dwarf halo can be very large.

The wavelike effects of ULAs (the de Broglie and Jeans scales) affect the halo density profile, and it cannot be completely described by the CDM result. How is the NFW profile modified by the presence of a ULA and what forms on small scales? Clearly there should be some granularity and a smoothing of the central cusp, each caused by the wave-mechanical uncertainty principle. When the density is smoothed over many Jeans scales, the profile should return to being NFW-like. These effects are observed in simple one-dimensional [176] and full cosmological [199] simulations. Both the core and the granularity [257] can be understood by considering a certain class of soliton solution [258, 259] of the

⁴⁰The virial radius is taken to be the radius where the density is 200 times the critical density, and the virial velocity is the circular velocity at this radius. The mass of a halo is often defined as $M_{200} = M(\langle r_{\rm vir} \rangle)$. One can use this to derive $\delta_{\rm NFW}(c)$. A typical concentration is $c \sim 10$.

axion equations of motion.⁴¹

We work in the non-relativistic Schrödinger picture of Section 4.5. Stationary wave, constant energy solutions take the form

$$\psi = \mathcal{X}(r)e^{-i\gamma t}, \qquad (150)$$

where γ is the energy eigenvalue. The system possesses a very useful *scaling* symmetry [258]:

$$(r, \mathcal{X}, \Psi, \gamma, M(< r), \rho) \to (r/\lambda, \lambda^2 \mathcal{X}, \lambda^2 \Psi, \lambda^2 \gamma, \lambda M(< r), \lambda^4 \rho), \qquad (151)$$

where the scale factor is λ , $\rho = \mathcal{X}^2$ is the soliton density, and $M(\langle r)$ is the soliton mass enclosed within radius r. Imposing the correct boundary conditions [239, 262] one can numerically solve the resulting system of ordinary differential equations to find $\mathcal{X}(r)$ and γ . Thanks to the scaling symmetry, this solution need only be found once. The solution with $\mathcal{X}(0) = 1$ gives $\gamma = -0.692$ for the zero node groundstate. The ground state solution for an isolated soliton is reached rapidly by a process of "gravitational cooling" [263, 262]. The ground state also provides a good description of the cores in virialised DM halos found in the simulations of Ref. [199].

The groundstate soliton solution possess a single characteristic radius, $r_{\rm sol}$, fixed entirely by the choice of units, which in turn is fixed by the axion mass. The scaling symmetry then uniquely fixes the relationship between the central density, $\rho_{\rm sol}$, and the characteristic radius:

$$r_{\rm sol} \propto m_a^{-1/2} \rho_{\rm sol}^{-1/4}$$
 (152)

The soliton characteristic radius has the same scaling properties as the Jeans scale! This is no surprise: the scalings are derived on dimensional grounds in

⁴¹Technically, these solutions are pseudo-solitons since the field is time-dependent, and they are not absolutely stable. This is a distinct difference between axions, which are real-valued fields, and complex scalar field DM. Complex fields have a conserved U(1) charge and true soliton solutions known as boson stars [260]. See e.g. Ref. [261], the Appendix of Ref. [239], and references therein, for more discussion.

ACCEPTED MANUSCRIPT



Figure 20: Left Panel: Halo density profiles from cosmological simulations of structure formation with a non-relativistic scalar field of mass $m_a = 8.1 \times 10^{-23}$ eV (equivalent to a ULA). There is a central soliton core, transitioning to an NFW profile at large radius, as Eq. (155). Right Panel: Understanding halo formation from soliton collision. The solitons virialize and leave behind a small, dense core, and a granular outer halo: (d) is a close up of (c) detailing this. Reproduced (with permission) from Ref. [257]. Copyright (2015) by The American Physical Society.

the non-relativistic limit. The Jeans scale is found from Eq. (100), which as we showed can be derived from perturbation theory on the Schrödinger equation via the quantum potential.

A good fit to the soliton density profile is provided by:

$$\rho_{\rm sol}(r) = \frac{\rho_{\rm sol}(0)}{(1 + (r/r_{\rm sol})^2)^8}, \qquad (153)$$

with

$$r_{\rm sol} = 22 \left(\frac{\rho_{\rm sol}(0)}{\rho_{\rm crit}}\right)^{-1/4} \left(\frac{m_a}{10^{-22} \text{ eV}}\right)^{-1/2} \text{ kpc} \,. \tag{154}$$

The soliton density has dropped to $\rho_{\rm sol}(0)/2$ at $r_{1/2} \approx 0.3 r_{\rm sol}$, which might be said to be the 'core radius.' For a central overdensity of 10^6 and $m_a = 10^{-22}$ eV we have $r_{1/2} = 0.2$ kpc, which is smaller than the naive halo Jeans scale, but is of order the de Broglie scale solved for via the circular velocity in the soliton profile [239].

A complete model for the axion halo density profile must match the soliton and NFW profiles continuously at some radius. An exact description of the matching is currently lacking (though of course, by order of magnitude it must be at the Jeans/ de Broglie scale), so we can simply parameterize it to occur at r_{ϵ} and write

$$\rho(r) = \Theta(r_{\epsilon} - r)\rho_{\rm sol}(r) + \Theta(r - r_{\epsilon})\rho_{\rm NFW}(r).$$
(155)

This profile can be used to compare to galactic rotation curves and stellar kinematical data, either to fix the ULA mass, or to make predictions for a given mass. Similar profiles occur in other models of scalar field DM, such as selfinteracting real or complex fields, and can also be used to fit density cores (see Section 6.4) and constrain the parameters of these models [264, 265].

Fig. 20 shows results from numerical simulation of structure formation with a massive scalar field in the non-relativistic regime, taken from Ref. [257], and discussed in Section 4.6. The left panel shows density profiles taken from a full cosmological simulation at various redshifts, for $m_a = 8.1 \times 10^{-23}$ eV [199]. The profiles show a central soliton matching to NFW when the density has dropped to $\mathcal{O}(10^{-2})$ of the central density. The soliton profile is well fit by Eq. (153). The right panel shows a numerical experiment of halo formation from collision of multiple solitons. The solitons virialize and leave behind a dense core, with a granular structure in the outer halo on the scale of the core size. The density profile from the soliton collision experiments is also shown in the left panel (arbitrarily normalized to show on the cosmological scale), and also has the same general form as Eq. (155). The formation of solitons during structure formation with ULAs seems an established numerical fact, but many consequences of this have yet to be fully explored.

6.4. ULAs and the CDM Small Scale Crises

The main CDM "small scale crises" are [266]:

- The missing satellites problem [267, 268]: CDM predicts more small Milky Way satellites than are observed.
- The too-big-to-fail problem [269]: CDM predicts more massive satellites that should contain stars than are observed.

• The cusp-core problem [270]: many observed low-mass systems contain flat central density profiles, not NFW cusps.

All of these problems, and variants of them, are essentially related to the overabundance of structure on small-scales in CDM, which itself is caused by the cold, collisionless, scale-free nature of CDM clustering.

Methods to address the small-scale problems come in two varieties: baryonic/astrophysical solutions, and dark matter solutions. A recent set of stateof-the-art simulations discussing the baryonic solutions based on feedback from star formation is Ref. [271], while a review of the relevant issues if Ref. [272].

Dark matter based solutions are interesting, as they attempt to solve the problems by the introduction of a small number of universal parameters. The extent to which these models offer a solution can in principle point to specific values of these parameters. Because of this, we should not only demand solutions to the small-scale crises, but also a complete and consistent cosmological history, which gives the models some predictive power. They also offer us a framework for parameterizing our uncertainty about DM. In the absence of a fundamental theory of DM, as Bayesians we should allow for varying DM properties at the same time as we vary the baryonic physics. Moving away from CDM in this way may allow for a mixed baryon-DM solution with more reasonable priors on astrophysical parameters. Finally, a range of parameters will also be excluded, e.g. providing too few satellites, and independent of offering a solution to the small-scale crises we have learned something new about DM.

So what do DM solutions to the small-scale crises look like? Two popular models are self-interacting (SI)DM [273], and WDM [170]. I will only discuss WDM in detail, as it is interesting to contrast with ULAs. For further discussion of SIDM and other interacting models with relation to the small-scale crises and other areas of galaxy formation, see e.g. Refs. [274, 275, 276, 277].

WDM suppresses structure formation by free-streaming and a cut-off in the matter power, as we discussed in Section 4.4.5. This has the ability to address the missing satellites and too-big-to-fail problems for 1.5 keV $\lesssim m_X \lesssim$

ACCEPTED MANUSCRIPT



Figure 21: Core size in a WDM halo of mass $M = 5 \times 10^8 M_{\odot}$ as a function of WDM thermal relic mass, with uncertainties given by the shaded region. A representative constraint of $m_X > 2$ keV is shown by the vertical dashed line, which leads to small, $\mathcal{O}(10 \text{ pc})$ cores and imposes the WDM *Catch 22*. Reproduced (with permission) from Ref. [278, 279], Fig. 2.
2.3 keV [190], while still producing enough satellites and passing constraints on phase space density [280]. Fermion degeneracy pressure and thermal velocities also allow WDM to form density cores [281]. The core-size-WDM mass relation is plotted in Fig. 21, with $r_c \sim m_X^{-1/2}$. Herein lies a problem known as the *Catch 22* of WDM [278]: core sizes in dwarf galaxies are too small if constraints from satellite abundance and LSS are accounted for. Specifically, the N-body simulations of Ref. [278] found that masses $m_X \sim 1-2$ keV gives a core of size $r_c \sim 10(20)$ pc in a dwarf galaxy of mass $10^{10(8)} M_{\odot}$, far smaller than the $\mathcal{O}(\text{kpc})$ cores required in e.g. Fornax and Sculptor [282]. Ref. [283] computed the WDM phase space density from N-body simulations and used this to derive the core size expected from free-streaming. A mass $m_X \approx 0.5$ keV can provide cores to the Milky Way dSphs, which is too light to be consistent with structure formation.

That an ultralight scalar field, such as an axion, could potentially also resolve the small-scale crises has been known for some time [284, 285, 176, 286]: the Jeans scale suppresses the formation of low mass halos, and at the same time leads to density cores in the form of solitons, as we have already discussed. Here we will address one issue: do ULAs suffer a *Catch 22* like WDM does? The answer, in short, is "no," or more accurately "not as severely."

Fig. 22 shows the one dimensional likelihood for ULA mass from fitting stellar velocity dispersion data of Ref. [282]. This simplified data uses two stellar populations and measures only the slopes of the density profiles within a given radius, in principle allowing an arbitrarily large core outside of this (and hence arbitrarily low axion mass). However, this would allow arbitrarily large dSph mass, while masses $M \gtrsim \mathcal{O}(\text{few}) \times 10^{10} M_{\odot}$ are forbidden by their long dynamical friction time scales [287].⁴² In Fig. 22 the dynamical friction constraint is imposed as a hard prior, supplementing the density profile slope analysis [282] of Ref. [239].

⁴²I compute the maximum mass for each dSph individually from the formula in Ref. [287] using their co-ordinates [288] and an approximate circular velocity $v_c \approx 200 \text{ km s}^{-1}$.



Figure 22: One dimensional posterior of ULA mass required to provide soliton cores to Fornax and Sculptor velocity dispersion data [282, 239], including a hard prior $M_{\rm vir} < M_{\rm fric}$ [287]. The 95% C.L. limit is 0.1×10^{-22} eV $< m_a < 1.4 \times 10^{-22}$ eV, the upper half of which is consistent with dedicated studies of structure formation and reionization with ULAs [238, 188]. Also shown is the 95% C.L. limit for a Jeans analysis of Fornax [199, 188], and the range required for Ursa Minor (UMi) cold clump longevity and long Fornax globular cluster (GC) orbital decay times [289].

Matching the Fornax and Sculptor data with ULAs alone, i.e. with the halo profile Eq. (155), requires 0.1×10^{-22} eV $< m_a < 1.4 \times 10^{-22}$ eV at 95% C.L. The best fit using a simplified Jeans analysis on Fornax alone is $m_a = 8.1^{+1.6}_{-1.7} \times 10^{-23}$ eV [199] (1 σ errors). Ref. [289] found that a range 0.3×10^{-22} eV $< m_a < 1 \times 10^{-22}$ eV can explain the cold clump longevity in Ursa Minor, and the distribution of globular clusters in Fornax, while respecting some constraints on the maximum dSph mass. All of these limits hint at a mass $m_a \sim 10^{-22}$ eV to solve CDM small-scale problems. Recall that this mass is allowed by constraints from halo formation and reionization [238, 188], reviewed in Section 6.2, i.e. ULAs do not suffer from the *Catch 22* like WDM does.

Eq. (117) translates the lower bound on ULA mass from high-z galaxies, $m_a \gtrsim 10^{-22}$ eV, into an equivalent WDM mass of $m_X \gtrsim 0.8$ keV, which from Fig. 21 gives a minuscule core size of $\mathcal{O}(30 \text{ pc})$. A harder constraint on $m_X > 2 \text{ keV}$ implies, by scaling of the half-mode, $m_a > 10^{-21} \text{ eV}$. Scaling the core size (from the 1 kpc core in Fornax with $m_a = 10^{-22} \text{ eV}$) as $m_a^{-1/2}$ still provides a significant $\mathcal{O}(300 \text{ pc})$ core even for this hypothetically stronger constraint.

Translating bounds from WDM to ULAs using Eq. (117) is good for orderof-magnitude estimates only. The exact constraints from structure formation depend sensitively on the slope of the transfer function and mass function near the cut off (e.g. Ref. [188]), which distinguishes WDM and ULAs, such that dedicated studies are necessary. There are tantalizing hints for $m_a = 10^{-22}$ eV as a solution to the small-scale crises. It is on the edge of current constraints, and of detectability in the EOR. Dedicated studies of this model, including full simulations with star formation and feedback (such as those comparing WDM and CDM including feedback in Ref. [290]), are necessary to explore this further.

7. Axions and Accelerated Expansion

7.1. Axions and the Cosmological Constant Problem

Our discussion in this review began with one of the greatest unsolved problems in modern physics: the cosmological constant (c.c.) problem [1], one of the most notoriously hard problems to solve in high energy physics [291]. One particularly attractive solution to this problem is anthropic tuning, which can be realized by eternal inflation populating a large number of vacua in the string landscape [292, 293] (the original idea dates back to Ref. [294]). In this picture, four-form fluxes and topologically complex compact spaces with $\mathcal{O}(100)$ or more cycles both play important roles.⁴³ Recall from Section 2.4 that axions arise from the wrapping of such fluxes on cycles. Furthermore, the canonical axion potential $V(\phi) \propto \cos \phi/f_a$ can provide positive and negative contributions to

 $^{^{43}}$ This "100" is one origin of the famous statement that the string theory landscape contains 10⁵⁰⁰ vacua. In this context it arises from demanding that the number of vacua is densely enough distributed near the observed value of the c.c. to make a universe in this region sufficiently likely.

ACCEPTED MANUSCRIPT

the vacuum energy, allowing axions to cancel contributions to the c.c. from other sources in a cosmologically dynamical manner.

The above observations suggest that:

- Axions may play a central role in the solution of the c.c. problem.
- The anthropic solution of the c.c. problem in the string landscape provides good motivation for the existence of the axiverse.

In this section we will briefly discuss a few ideas relating axions to the c.c. problem.

We begin with the simplest model of axion quintessence. As we already saw in Section 5, ULAs with $m_a \sim H_0 \sim 10^{-33}$ eV can act as DE, with the axion potential energy providing an effective cosmological constant and driving accelerated expansion as a form of quintessence. Since the axion mass is protected by a shift symmetry and can easily remain so light, the idea of axion and general pNGB [295] quintessence is natural, and has a long history [296].⁴⁴

The model is specified by the potential

$$V(\phi) = \Lambda_a^4 \left[1 + \cos\left(\frac{\phi}{f_a}\right) \right] \,, \tag{156}$$

(note the phase shift from our previous definition). The most recent constraints on this scenario using *Planck* data can be found in Ref. [298] and are summarized in Fig. 23. Since the vacuum in this potential has zero energy, the quintessence contribution to the energy budget, Ω_{ϕ} , is controlled by the initial field displacement, ϕ_i . The value of $\Omega_{\phi} \approx 0.69$ is well constrained by the requirement of driving accelerated expansion, and just as we saw in Fig. 15 (right panel) large field displacements and decay constants are required to achieve this. There is a degeneracy between the energy density and the decay constant caused by the requirement of keeping the potential roughly flat compared to H_0 : increasing Λ_a requires increasing f_a to retain flatness.

⁴⁴For a review of DE and quintessence models, see Ref. [297].



Figure 23: Constraints on axion DE from *Planck. Left Panel:* Potential mass scale in units of the critical density, versus decay constant. Note that here M rather than Λ_a is used. *Right Panel:* Field displacement versus density fraction. The density fraction is well constrained by the demand that the axion cause accelerated expansion with zero overall vacuum energy. Reproduced (with permission) from Ref. [298].

A simple generalization of this quintessence model goes along the lines of N-flation (see Section 7.2.1), and was discussed in Ref. [58]. Taking the string theory-inspired potential in Eq. (34) for N axions of almost degenerate mass, and assuming a fixed decay constant:

$$f_a = \frac{M_{pl}}{S_{\text{inst.}}}, \qquad (157)$$

it can be shown that axion quintessence requires

$$S_{\text{inst.}} \sim 200 - 300 \text{ and } N \gtrsim S_{\text{inst.}}^2$$
, (158)

if the axion contribution to DE is to be non-negligible.

Alternatively, successful quintessence can occur for sub-Planckian decay constants if the initial displacement $\phi_i/f_a \sim \pi$. This idea was considered in Ref. [299] for the case of multiple axions. Taking constant $f_a \approx 10^{17}$ GeV, potential energy scale $\Lambda_a = 10^{12}$ GeV $e^{-S_{\text{inst}}}$ and assuming that the instanton action changes by $\mathcal{O}(10)$ for each axion, then with 24 axions the probability that one axion is close enough to the top of the cosine potential to drive successful quintessence occurs in approximately 1% of cases. This relatively modest number of axions can achieve successful quintessence with sub-Planckian f_a and minimal fine-tuning. However, the limiting case in this study was the assumption of constant f_a , rather than considering the variation of f_a with $S_{\text{inst.}}$. The heavier axions in this scenario will be subject to all the phenomenology and constraints discussed elsewhere in this review. In Ref. [299] it was proposed to avoid unwanted impacts on cosmology by having the heavy axions decay, or evolve in a modified potential.

The models of Refs. [298, 58, 299] simply require that axions provide successful quintessence, and assume that the bare c.c. is of an acceptably small value, due to some unknown physical mechanism, or due to anthropics. This is a solution to the "new c.c." or "why now?" problem of obtaining small masses and potential energies of order the present critical density. Let us now turn to the role of axions in solving the "old c.c." problem, i.e. the much more taxing problem that

$$\rho_{\Lambda,\text{obs.}} \sim 10^{-120} M_{pl}^4, \text{ while } \rho_{\Lambda,\text{theory}} \sim M_{pl}^4.$$
(159)

Ref. [300] considered the possibility of using subleading instanton corrections in a multi-axion model to generate a field space with an exponentially large number of vacua. The potential for the N axion fields θ_i charged under instantons labelled by j with charge Q_i^j has the form

$$V(\vec{\theta}) = \sum_{j} \Lambda_j^4 \left[1 - \cos(2\pi \mathcal{Q}_i^j \theta^i + \delta^j) \right] + V_0 \,, \tag{160}$$

where where δ^j is an arbitrary phase. The leading potential is split into bands of width $\Lambda_{\text{sub.}}^4$ by the subleading pieces, with each band containing $\mathcal{N}_{\text{sub.}}$ different vacua. This splitting leads to vacua within $\Lambda_{\text{sub.}}^4/\mathcal{N}_{\text{sub.}}$ of zero, as illustrated in Fig. 24. Therefore, if we take $\Lambda_{\text{sub.}}^4 \sim M_{pl}^4$ one requires $\mathcal{N}_{\text{sub.}} \sim 10^{120}$ distinct vacua to solve the c.c. problem.

For a random matrix model of the instanton charges, Ref. [300] showed that

ACCEPTED MANUSCRIPT



Figure 24: The axionic band structure of the cosmological constant. A multi-axion theory with sub-leading instanton contributions can give rise to an exponentially large number of vacua, with energy splittings inversely proportional to the number of vacua. This mechanism may provide a solution to the cosmological constant problem. Reproduced, with permission, from Ref. [300]. Copyright (2016) by The American Physical Society.



Figure 25: Constraints on inflationary models from *Planck* [96], showing 1 and 2σ marginalized confidence regions. Note that the potentials $\sim \phi^{2/3}$, $\sim \phi$, and $\sim \phi^{4/3}$ are the approximate predictions of axion monodromy models if power spectrum oscillations are ignored.

that expected number of vacua in a theory with N axions obeys the bound

$$\langle \mathcal{N}_{\text{sub.}}^2 \rangle \gtrsim \sqrt{2\pi N} \left(\frac{3}{e}\right)^N .$$
 (161)

Thus there is an exponentially large number of vacua. An example with 500 axions suffices to obtain the desired factor if 10^{120} . In this model, the expected mass distribution of the axions was not computed, but the logarithmic distribution of Λ_j was invoked. It is thus not clear at this stage what the role of these axions would be in terms of a DM model. Some evidence suggests that this model could incorporate successful axion inflation, a topic to which we now turn.

7.2. Axion Inflation

In Section 3.2 we discussed the role of stable axion DM fields as spectators during inflation. Here, we discuss the scenario where an unstable axion field itself drives inflation. Inflation [93, 94, 95] is a hypothetical period of accelerated expansion in the early Universe, invoked to explain certain cosmological puzzles relating to initial conditions.⁴⁵ The simplest inflationary models involve a single, minimally coupled, scalar field ("the inflaton"), driving the expansion by the existence of a potential, $V(\phi)$, on which the field is slowly rolling. Inflation ends when this field reaches the minimum of its potential, oscillates, and decays into radiation: a process known as "reheating." This reheating must occur in order to produce a hot big bang cosmology and all its successful predictions, from BBN to the CMB.

The inflaton potential must be very flat compared to the other scales in play, namely the Hubble scale. The expansion is driven by the potential, and so $3H_I^2 M_{pl}^2 \approx V(\phi)$. This defines the inflationary "slow roll parameters," which depend on the flatness of the potential. The first two slow roll parameters are:

$$\epsilon_{\rm inf} = \frac{M_{pl}^2}{2} \left(\frac{V'}{V}\right)^2, \quad \eta_{\rm inf} = M_{pl}^2 \frac{V''}{V}, \quad (162)$$

and inflation requires each of these be very much less than unity over a large, relative to H_I , field range. Axions are extraordinarily good inflaton candidates because the shift symmetry protects the flatness of the potential from quantum corrections. It is important to note that, because the inflaton must decay, the axion driving inflation is not a dark matter (or dark energy) axion. In particular, therefore, the inflaton is not the QCD axion!

The standard view of constraints on inflationary models is shown in Fig. 25, taken from Ref. [96]. These simple constraints allow the cosmological initial conditions two degrees of freedom after normalization by A_s . These are the tilt, n_s , and the tensor-to-scalar ratio, r_T . These numbers are determined by the parameters of the inflaton potential. Additional freedom is afforded to the

⁴⁵It is not my purpose here to give a review of inflation, and I defer all detailed calculations and notation. For a general review of inflation, see Ref. [301], for inflation in string theory, see Ref. [302], and for specifics of axion inflation, see Ref. [11]. The state of the art in constraints on inflation can be found in Refs. [96, 303], while an exhaustive list of single-field-slow-roll models can be found in Ref. [233].

model by the number of e-folds of observable inflation, N_* , which takes into account uncertainty about the reheating epoch [304, 305, 306] and the initial conditions of the inflaton itself [307, 308]. The constraints shown assume that the primordial power spectra are described by power laws. We will briefly discuss spectra with features later.

7.2.1. Natural Inflation and Variants

So-called "Natural Inflation" [309] is the simplest example of inflation with an axion. It simply takes our usual potential

$$V(\phi) = \Lambda_a^4 \left[1 \pm \cos\left(\frac{\phi}{f_a}\right) \right] \,. \tag{163}$$

Natural Inflation is a standard single field slow roll model, giving power law scalar and tensor power spectra.

In its original incarnation, Natural Inflation takes $\Lambda_a \sim m_{\rm GUT}$ and $f_a \sim M_{pl}$. One combination of these parameters is fixed by normalizing A_s , and so, including N_* , the model has two additional parameters specifying its location on the (n_s, r_T) plane. Thus, in Fig. 25, Natural Inflation sweeps out a broad region, a portion of which is consistent with the observational constraints. In the limit $f_a \to \infty$ with Λ_a^2/f_a held fixed, Natural Inflation approaches $m^2\phi^2$ "chaotic" inflation. Furthermore, we see that Natural Inflation consistent with the observed value of n_s predicts a measurably large value of $r_T \gtrsim 10^{-2}$. This is a reasonable sensitivity to expect for near-future CMB experiments [236, 229], and so Natural Inflation makes testable predictions.⁴⁶

The value of the tensor-to-scalar ratio in single field slow roll inflation is closely tied to the field range, $\Delta \phi$, over which the potential is flat, and for which inflation occurs. The "Lyth bound" [87] states:

$$\Delta \phi = 0.46 M_{pl} (r_T / 0.07)^{1/2} \,. \tag{164}$$

It is generally held that over such large field excursions one loses perturbative control over quantum mechanical corrections to the potential (in particular,

⁴⁶Up to the usual caveats made by notable detractors.

those of quantum gravity ⁴⁷), and so achieving large amplitude tensor modes is hard to achieve in a theoretically consistent manner.

The natural field range in the potential Eq. 163 is f_a , and so for Natural Inflation the Lyth bound implies that f_a must be of order the Planck scale. The potential is protected from other corrections by the axion shift symmetry, which is restored in the limit $\Lambda_a \to 0$, making the the theory technically natural. This is where the "natural" in Natural Inflation comes from: the axion potential is flat over scales $\Delta \phi \sim f_a$, and is immune to radiative corrections. "Standard" inflation at the GUT scale, with observably large r_T , can be achieved with a Planckian decay constant.

As we have already mentioned, however, the weak gravity conjecture [59] places some constraints on $f_a \gtrsim M_{pl}$ in theories of quantum gravity, in particular forbidding it in the case of a single canonically normalised axion field. We have also seen that in string theory one finds $f_a < M_{pl}$ in our simple example. One should therefore worry about embedding Natural Inflation in a UV complete theory. The simplest models, which remain quasi-single field and produce power-law initial power spectra, are based on the general idea of "Assisted Inflation" [75] (or even more generally, on "kinetic alignment" [62]).

In Assisted Inflation, one uses the frictional coupling of multiple fields induced by the Hubble expansion to provide extra damping to the collective motion in field space. This slows the collective motion down, effectively flattening the potential of the quasi-single field trajectory. A simple example of Assisted Inflation applied to axion models is "N-flation" [71]. N-flation takes N axions with identical potentials:

$$V(\vec{\phi}) = \sum_{n=1}^{N} V_n(\phi_n) , \qquad (165)$$

where $V_n = \Lambda_n^4 \cos(\phi_n/f_n)$ is the familiar cosine potential.⁴⁸ One now simply

⁴⁷See also Ref. [310], which suggests that large field inflation in general might be forbidden by entropy bounds in quantum gravity.

⁴⁸I drop the higher order instanton corrections discussed in Ref. [71]. The radiative stability of N-flation in field theory and in string theory was also established in Ref. [71], and so it fits

applies Pythagoras theorem to the N-dimensional field space.

For simplicity, consider the case of all equal decay constants, $f_n = f_a$, and scales $\Lambda_n = \Lambda_a$. Now displace each field from the origin by an equal amount,⁴⁹ $\phi_n = \alpha M_{pl}$, with $\alpha^2 < 2\pi f_a^2/M_{pl}^2$. The total radial displacement is $\phi_r = \sqrt{N}\alpha M_{pl}$ and the mass of the radial field is $m = \Lambda_a^2/f_a$. It is clear that we can arrange for super-Planckian displacement of ϕ_r , with $f_a < M_{pl}$ and $\alpha^2 \ll 1$, if N is large enough. As in Assisted Inflation, each individual ϕ_n feels the friction of all its brothers and sisters, and it is the collective radial motion in field space that acts as the inflaton.

Finally, the Kim-Nilles-Peloso model [61] generalizes the multi-axion potential allowing rotations between the fields. This occurs if multiple axions, i, each obtain potentials from multiple non-perturbative sources, j, but with different strengths, f_{ij} . "Decay constant alignment" then allows to create a flat-direction on the potential with a large effective value of $f_{a,\text{eff}} > M_{pl}$ even is each individual $f_{ij} < M_{pl}$, so long as sufficient degeneracy between the decay constants occurs.

7.2.2. Axion Monodromy

Axion Monodromy $[312, 313]^{50}$ is another model within the pantheon of UV completions of axion inflation allowing for large field excursions, and thus measurably-large r_T . It differs from the examples discussed above, however, in that it *does not produce power-law initial power spectra*, but instead modulates the power law spectra with periodic features.

In string theory, a monodromy occurs when an axion field winds around a particular location in moduli space, like the Riemann sheets of $\log z$ winding around the origin in the complex plane. The monodromy provides an explicit

the maxims of a natural theory.

⁴⁹The equal displacement trajectory is an attractor of Assisted Inflation [75]. N-flation also takes initial conditions with zero angular momentum in field space. For a discussion of the dynamics with angular motion, see Ref. [311].

 $^{^{50}}$ For some possible issues in explicit realisations of this model, see e.g. Refs. [314, 315].

breaking of the periodicity of the axion potential, and lifts it at large field values. The extra potential energy is supplied by the wrapping of branes around compact dimensions. It has been described colloquially as a "wind up toy."

Over large field excursions $\Delta \phi \gg f_a$ the potential is on average described as $V \propto \phi^p$ for some p, while on small scales the potential is modulated by the usual, instanton-induced, axion cosine. The potential is of the form

$$V(\phi) = \mu^{4-p}\phi^p + \Lambda_a^4 \left[1 - \cos\left(\frac{\phi}{f_a}\right)\right].$$
(166)

As inflation proceeds along the ϕ direction, one has slow roll on the ϕ^p piece. From Fig. 25 we see that the predictions of large-field ϕ^p models of inflation, with p = 2/3, 1, 4/3, motivated by axion monodromy, are consistent with the observations, and predict measurably large tensor modes.

The cosine part of the potential, however, modulates the slow-roll trajectory with oscillations. This leads to an oscillatory power spectrum for the primordial curvature perturbations of the form [316]

$$P_{\zeta}(k) = A_s \left(\frac{k}{k_0}\right)^{n_s - 1 + \frac{\delta n_s}{\ln k/k_0} \cos(\phi_k/f_a)}, \qquad (167)$$

with $\phi_k = \sqrt{\phi_0^2 - 2\ln(k/k_0)}$, ϕ_0 the value of the field at horizon crossing of the pivot scale, and $\delta n_s \propto \Lambda_a^4/\mu^3 f_a$ for p = 1.

The axion monodromy power spectrum undergoes rapid oscillations in log k, and constraining it properly using CMB data requires special care (e.g. Refs. [317, 318]). The latest *Planck* data show no statistically significant evidence for the presence of power spectrum oscillations, though there are various lowsignificance hints [96]. Axion monodromy also predicts "resonant non-Gaussianity" [316]. Current data cannot reach the sensitivity to confirm hints of oscillations in the power spectrum through resonant non-Gaussianity in the bispectrum, though this may be possible in future.

8. Gravitational Interactions with Black Holes and Pulsars

In this section we consider two astrophysical probes of axion DM that arise purely from gravitational interactions, and are quite distinct from any signatures we have considered so far.

8.1. Black Hole Superradiance

BHSR is a very general way to search for light bosonic fields. It relies only on their gravitational interaction and assumes nothing about couplings to the standard model or their cosmological energy density.

Massive bosonic fields can form bound states around astrophysical black holes (BHs), just like the energy levels of electrons in the hydrogen atom. Infalling scalar waves extract energy and angular momentum from a spinning Kerr BH and emerge with more energy than they went in with; this is known as the Penrose process [319]. Being bosons, the energy levels in the "gravitational atom" can be filled exponentially via this superradiant instability (see Ref. [320] for a review). The boson mass leads to the existence of stable orbits, like the energy levels of an atom. These stable orbits lead to a barrier in the effective potential, and act like the mirror in Press and Teukolsky's "black hole bomb" [321, 322]. The energy levels then fill up via the superrandiant instability until they eventually radiate away the extracted energy, for example as gravitational waves. The bosons do not even need to be present initially (i.e. they do not have to be the DM) for this process to occur: superradiance can start from a quantum mechanical fluctuation. It is thus a completely generic feature of massive bosonic fields in astrophysics, and turns astrophysical BHs into sensitive detectors of bosons in the mass range 10^{-20} to 10^{-10} eV [74, 17, 323, 324, 325].

The instability leads to the spin down of BHs. The spin-down rate is controlled by the effective coupling of the gravitational atom:

$$\alpha_G = r_G m_a \,, \quad r_G \equiv GM \,, \tag{168}$$

where M is the BH mass. The size of the "cloud" formed around the BH is fixed by the orbital velocity $v \sim \alpha_G/\ell$ to be $r_c \sim n^2 r_G/\alpha_G^2$ (where ℓ is the orbital quantum number and n is the energy level). This is approximately the



Figure 26: Constraints on axions from BHSR. Left Panel: Solar mass black holes, 2σ . Right Panel: Supermassive black holes, 1σ . Reproduced (with permission) from Ref. [74]. Copyright (2015) by The American Physical Society.

de Broglie scale for a circular orbit of radius r_c , and we observe the link to our previous discussions of density profiles and the Jeans scale.

With $\alpha_G = 0.3$ the superradiance time-scale is short (~ years) for both stellar mass ($M = 10M_{\odot}$) and super-massive ($M = 10^7 M_{\odot}$) BHs, which sets the characteristic axion mass for spin-down. A number of BHs are observed, and their masses and spins have been measured (data are given with citations in Ref. [74]). Since the spinning BHs would be spun-down in the presence of a light boson, these observations can be used to exclude various axion masses.

The exclusions are shown in Fig. 26. Stellar mass BHs exclude a range of masses 6×10^{-13} eV $< m_a < 2 \times 10^{-11}$ eV at 2σ , which for the QCD axion excludes 3×10^{17} GeV $< f_a < 1 \times 10^{19}$ GeV. The supermassive BH measurements are more uncertain: there are fewer measurements excluding a narrower range of masses at 1σ only. The range probed is roughly 10^{-18} eV $< m_a < 10^{-16}$ eV. Higher precision measurements in future could improve these bounds.

Finally, transitions and annihilations within the axion cloud predict the emission of monochromatic gravitational waves. The detection prospects for such a signal with advanced LIGO [326] and eLISA [327] are discussed in Ref. [74]. Advanced LIGO has the potential to discover evidence for the QCD axion with $m_a \sim 10^{-10}$ eV in the not-too-distant future. Further in the future, eLISA may be sensitive to the lower-frequency emission for ULAs with $m_a \sim 10^{-17}$ eV, with the possibility to detect ~ 10's of events from axion annihilations out to a radius of ~ 100 Mpc.

8.2. Pressure Oscillations and Pulsar Timing

The pressure, $P_a = w_a \rho_a$, in the axion energy momentum tensor undergoes rapid oscillations as $\cos 2m_a t$, leading to the $\langle w_a \rangle = 0$ DM-like properties of the axion. Local pressure perturbations, δP_a , also undergo such oscillations. Such pressure oscillations induce oscillations of the gravitational potential, which in turn induce a time-dependent frequency shift and a time delay for any propagating signal. If the DM in the Milky Way is composed of ULAs, then the amplitude of the signal is fixed by the local DM abundance, $\rho_{\rm DM} \approx 0.3$ GeV cm⁻³. Ref. [328] considered the effect of such oscillations on pulsar timing experiments.

Consider the energy momentum tensor, Eq. (54). The local axion field can be described as

$$\phi(\vec{x}, t) = \phi_0(\vec{x}) \cos[mt + \xi(\vec{x})], \qquad (169)$$

where ϕ_0 is the local amplitude and ξ is a local phase. To leading order, the energy density is static, but the pressure oscillates. The local amplitude is fixed by the DM density as:

$$\phi_0(\vec{x}) = \frac{\sqrt{2\rho_{\rm DM}}}{m_a} \,, \tag{170}$$

which in turn fixes the local pressure:

$$P(\vec{x},t) = -\frac{1}{2}m_a^2\phi_0^2\cos(2m_a t + 2\xi).$$
(171)

The Newtonian potentials, Ψ and Φ , are sourced by the density and the pressure. They have dominant time-independent pieces, and sub-dominant oscillating pieces, found from the Einstein equations.

The oscillating potential induces an oscillating delay in arrival time of pulsar signals, with frequency $2m_a$ and amplitude [328]:

$$\Delta t_{\phi} = \frac{\pi G_N \rho_{\rm DM}}{m_a^3} \sin\left[m_a D + \xi(\vec{x}_0) - \xi(\vec{x}_p)\right], \qquad (172)$$



Figure 27: Current and forecasted constraints on axion/scalar DM from the effect of pressure oscillations on pulsar timing. The green line shows the expected level of signal if axions compose the local DM. SKA will be sensitive to masses $m_a \leq 2.3 \times 10^{-23}$ eV. Reproduced (with permission) from Ref. [328].

where D is the distance to the pulsar, \vec{x}_p is the pulsar location, and \vec{x}_0 is the position of the Earth. In the variance of this signal the unknown local phases, ξ , and the pulsar distance, D, drop out. The amplitude of the signal decreases for heavier axions, and has a maximum at a given mass set by the DM density.

Ref. [328] considered the sensitivity of pulsar timing arrays to this signal by comparing the amplitude Δt_{ϕ} to the corresponding time delay from a stochastic gravitational wave background. Fig. 27 shows the current constraints from Parkes Pulsar Timing Array (PPTA) [329], forecasts for a 5 year observation with PPTA, and forecasts for ten years with the Square Kilometre Array (SKA). Current limits do not reach the level of the expected signal from ULAs, however SKA will be sensitive to masses $m_a \leq 2.3 \times 10^{-23}$ eV and DM fractions as low as one percent. This is a powerful probe complementary to the constraints from structure formation discussed in Sections 5 and 6.

The best current limits from pulsar timing come from the analysis of Ref. [330]

from the NANOGrav PTA. The limits are an order of magnitude higher than the expected signal at $m_a = 10^{-23}$ eV, consistent with the rougher bounds shown in Fig. 27. Uncertainties in the analysis of PTA data relevant for constraining pressure oscillations include characteristics of the partner in binary pulsars, and modelling of radio wave propagation through the ionized interstellar medium. In the Bayesian analysis of Ref. [330], the unknown pulsar parameters were marginalized over, following Ref. [331].

As already mentioned, the pulsar timing signal from pressure oscillations depends only on gravitational interactions. Recently, Ref. [332] considered the pulsar timing signal from interactions between scalar DM and the standard model. For typical coupling strengths, these model-dependent signals are much stronger than the pressure oscillation signal. For $m \leq 10^{-22}$ eV the PTA limits from interactions can be stronger than e.g. torsion balance or atom interferometry constraints.

9. Non-Gravitational Interactions

Two classic methods for detecting the QCD axion were proposed by Sikivie in Ref. [333] and are known as *haloscopes* and *helioscopes*. Another archetypal axion experiment is "light shining through a wall" (LSW) [334]. In recent years there has been a flurry of new ideas in axion (and scalar) direct detection (see, for example, Refs. [47, 15]). Some of the most important bounds on axions, in particular establishing the lower limit on $f_a \gtrsim 10^9$ GeV for the QCD axion, come from considering stellar processes (e.g. Ref. [10]). Many bounds on axions from their interactions exploit the two-photon coupling in the presence of magnetic fields (the Primakoff [335] process, see Fig. 28), though we will also discuss the fermion and $G\tilde{G}$ couplings. A recent review of constraints on the axion-photon coupling is given in Ref. [16], and shown in Fig. 29. We do not discuss collider signatures of axions in any detail. A recent discussion of existing constraints and future prospects is given in Ref. [35].



Figure 28: Axion-photon interaction via the Primakoff process. In the presence of an external magnetic field, B, axions can convert into photons, and vice versa. This basic process, arising from the electromagnetic anomaly and expressed in the effective interaction with co-efficient $g_{\phi\gamma}$ in Eq. (23), underpins many constraints on axions and efforts to detect them.

9.1. Stellar Astrophysics

Axion emission is an energy-loss channel for stars and supernovae. The observed properties of stars can be used to limit the existence of such a channel, and the emitted stellar axions can be searched for. The stellar astrophysics limits *apply regardless of whether the axion is DM*, because we are producing axions directly, and not relying on a cosmic population.

The solar luminosity in axions is

$$L_a = 1.85 \times 10^{-3} \left(\frac{g_{\phi\gamma}}{10^{10} \text{ GeV}}\right)^2 L_{\odot} \,, \tag{173}$$

where L_{\odot} is the photon luminosity. The maximum luminosity is at 3 keV, and the average is 4.2 keV [10]. Axion production occurs as long as m_a is less than the cental temperature of the sun, $T_{\odot} \approx 1$ keV and leads the sun to consume nuclear fuel faster. A very crude bound can be found by demanding that the axion luminosity is less than the photon luminosity. Equating $g_{\phi\gamma} \sim$ $(\alpha_{\rm EM}/2\pi f_a)$ for the QCD axion gives $f_{\rm QCD} \gtrsim 5 \times 10^5$ GeV.

The strongest bound on solar axions can be derived from direct searches for them. The helioscope converts solar axions back to photons in a macroscopic

ACCEPTED MANUSCRIPT



Figure 29: Summary of constraints on the axion-photon coupling $g_{\phi\gamma}$, Eq. (23) (here labelled $g_{a\gamma}$) as a function of axion mass. The line "ALP CDM" corresponds to setting $g_{\phi\gamma} = \alpha_{\rm EM}/2\pi f_a$ and requiring f_a to be large enough such that $\Omega_a h^2 \approx 0.12$ (c.f. Fig. 5). Reproduced (with permission) from Ref. [16].

B field on earth, and observes the photons in the X-ray. The state-of-the-art helioscope is the CERN Axion Solar Telescope (CAST) [336, 337, 338]. The 95% C.L. bounds are:

$$g_{\phi\gamma} < 8.8 \times 10^{-11} \text{ GeV}^{-1} \quad (m_a \lesssim 0.02 \text{ eV}),$$
 (174)

$$g_{\phi\gamma} < 3.3 \times 10^{-10} \text{ GeV}^{-1} \quad (m_a \lesssim 1.17 \text{ eV}),$$
 (175)

where the two bounds refer to two different experimental configurations (low mass, vacuum; high mass, ³He). The proposed International AXion Observatory (IAXO) [339] could improve the bound on $g_{\phi\gamma}$ by an order of magnitude (see Fig. 29).

The ratio of horizontal branch (HB) stars to red giants in galactic globular clusters is altered by axion-photon conversion inside stars, and places a competitive constraint on $g_{\phi\gamma}$ for axions with masses less than the stellar internal temperatures, $T \leq 100$ keV. In Fig. 29, this is shown as $g_{\phi\gamma} < 1 \times 10^{-10}$ GeV⁻¹ (this constraint is also shown in terms of the axion lifetime in Fig. 32.). The most up-to-date constraint using 39 galactic globular clusters and state-of-theart stellar modelling is that of Ref. [340], which gives:

$$g_{\phi\gamma} < 6.6 \times 10^{-11} \text{ GeV}^{-1} \quad (95\% C.L.) \quad (m_a \lesssim 100 \text{ keV}).$$
 (176)

Supernova SN1987a places the strongest limit on $g_{\phi\gamma}$ for low mass axions from the lack of observation of a gamma ray signal coincident with the neutrino burst due to axion-photon conversion within the Milky Way. The most up-todate limit from Ref. [341] is

$$g_{\phi\gamma} < 5.3 \times 10^{-12} \text{ GeV}^{-1} \quad (m_a < 4.4 \times 10^{-10} \text{ eV}).$$
 (177)

Note that this limit is not shown on Fig. 29, which does not extend to such low mass axions. SN1987A also places bounds on heavier axions with masses less than the SNe internal temperature, $T \approx 50$ MeV, where axion emission leads to additional cooling. An approximate bound is (e.g. Ref [342]):

$$g_{\phi\gamma} < 10^{-9} \text{ GeV}^{-1} \quad (m_a < 50 \text{ MeV}).$$
 (178)

Energy loss in globular cluster stars and white dwarfs sets limits on the axionelectron coupling, $g_{\phi e}$. The strongest constraint comes from axion bremsstrahlung in globular cluster red giants [343]:

$$g_{\phi e} < 3.3 \times 10^{-13} \,. \tag{179}$$

Finally, the duration of the neutrino burst from SN1987a can be used to constrain the axion-nucleon interaction, $g_{\phi N}$. If axions interact strongly enough with nuclei, then axion emission via nuclear bremsstrahlung, $N+N \rightarrow N+N+\phi$, is a more efficient energy-loss channel than neutrino emission, shortening the observed neutrino burst [344]. The theoretical calculation of supernova energy loss involves many uncertainties, but approximate bounds can be obtained. For a KSVZ type axion with no tree-level fermion couplings the bound is (see Ref. [10] for discussion)

$$f_a \gtrsim 4 \times 10^8 \text{ GeV} \quad (\text{KSVZ}).$$
 (180)

9.2. "Light Shining Through a Wall"

LSW is based on a very simple idea: shine a laser beam at a wall; apply a magnetic field so that it converts into axions, which travel freely through the wall; on the other side of the wall apply another magnetic field to convert the axions back to observable photons (for a review, see Ref. [13]). Just like the stellar astrophysics limits, this is direct axion production and *applies regardless* of whether the axion is DM.

The conversion probability, $P(\gamma \to \phi)$, for photons of energy ω into axions in the presence of a coherent magnetic field, B, of length L is

$$P(\gamma \to \phi) = 4 \frac{g_{\phi\gamma}^2 B^2 \omega^2}{m_a^4} \sin^2 \left(\frac{m_a^2 L}{4\omega}\right) \,. \tag{181}$$

The conversion probability can also be affected by using a medium with a refractive index $n_r \neq 1$, and by use of resonant cavities to enhance conversion and reconversion on either side of the wall.

The constraints from current LSW experiments are not particularly strong compared to astrophysical constraints, and do not appear on the scale of Fig. 29.

The strongest bounds come from the Any Light Particle Search (ALPS) experiment [345] and are roughly

$$g_{\phi\gamma} \lesssim 7 \times 10^{-8} \text{ GeV}^{-1} \quad (m_a \lesssim 10^{-3} \text{ eV}).$$
 (182)

The planned experiment ALPS-II [346] will improve these limits by more than three orders of magnitude, sensitive to $g_{\phi\gamma} \sim 2 \times 10^{-11} \text{ GeV}^{-1}$ over a similar range of masses. The projected reach is shown in Fig. 29 and will be competitive with astrophysical and helioscope limits discussed in Sec. 9.1.

9.3. Vacuum Birefringence and Dichroism

In the presence of a magnetic field, the Primakoff interaction between axions and photons allows for the vacuum to become birefringent and dichroic [333]. These effects cause the polarization plane of linearly polarized light to be rotated as it propagates. With no external magnetic field, we simply have birefringence (rotation with no absorption, we consider this effect in a cosmological context in Section 9.9), while in the presence of a magnetic field, there is absorption of one polarization state, i.e. dichroism. The amplitude of the dichroism is given by [347]

$$\varepsilon = \sin 2\theta \left(\frac{BLg_{\phi\gamma}}{4}\right)^2 \left[\frac{\sin(m_a^2 L/4\omega)}{m_a^2 L/4\omega}\right]^2, \qquad (183)$$

where θ is the angle between the magnetic field, B, and the polarization direction, L is the length of the magnetic region, and ω is the photon energy. The effect can be enhanced in a Fabrey-Perot cavity by increasing the number of passes the light makes in the cavity. Measuring the dichroism of the vacuum in the presence of a *B*-field can thus be used to place constraints on the existence of axions possessing the two-photon coupling.

Using this technique, in 2006 PVLAS reported evidence for a polarization rotation in the presence in a $B \approx 5$ T field of $\alpha = (3.9 \pm 0.5) \times 10^{-12}$ rad/pass (3σ uncertainties). This was interpreted as evidence for an axion with $m_a \approx 1$ meV and $g_{\phi\gamma} \approx 10^{-5}$ GeV⁻¹ [347]. Although this signal was already in tension with results from helioscopes, considerable interest was generated. The relevant parameter space was later directly excluded by the LSW experiment, GammeV [348]. Furthermore, reruns of PVLAS at different field strengths [349] showed that the signal of Ref. [347] was in fact due to instrumental artefacts. Nevertheless, this remains an interesting part of the story of axion constraints.

9.4. Axion Mediated Forces

The couplings $g_{\phi e}$ and $g_{\phi N}$ of Eq. (23) cause the axion to mediate spindependent forces. Such force exists *independently of whether the axion is DM*. The resulting dipole-dipole interaction in the non-relativistic limit gives rise to the following potential [350]:

$$V(r) = \frac{g_{\phi i}g_{\phi j}}{16\pi M_i M_j} \left[(\hat{\sigma}_i \cdot \hat{\sigma}_j) \left(\frac{m_a}{r^2} + \frac{1}{r^3} \right) - (\hat{\sigma}_i \cdot \hat{r}) (\hat{\sigma}_j \cdot \hat{r}) \left(\frac{m_a^2}{r} + \frac{3m_a}{r^2} + \frac{3}{r^3} \right) \right] e^{-m_a r},$$
(184)

where i, j labels the electron or nucleon with mass M, $\hat{\sigma}$ is a unit vector in the direction of the spin, and \hat{r} is a unit vector along the line of centres.

The interaction is of Yukawa-type and its range is suppressed by $e^{-m_a r}$. Even though this force can be long-range for ULAs, they are not subject to standard fifth-force constraints since the macropscopic sources must be spinpolarized. The dipole-dipole interactions between nucleons and electrons are only weakly constrained by current experiments, and the resulting bounds on $g_{\phi e}$ and $g_{\phi N}$ are not as strong as those from stellar astrophysics. They are [351]

$$g_{\phi N} < 0.85 \times 10^{-4} \quad (m_a \lesssim 10^{-7} \text{ eV}),$$
 (185)

$$g_{\phi e} < 3 \times 10^{-8} \quad (m_a \lesssim 10^{-6} \text{ eV}).$$
 (186)

(187)

If the axion also has scalar interactions of the form $g_s \phi \bar{\psi} \psi$, then monopolemonopole and monopole-dipole potentials are induced [350]. For a general ALP, g_s should be very small on symmetry grounds. The limits on the scalar interaction strength for the QCD axion are given by the limits on d_n and by the amount of CP violation in the standard model. Current bounds are weaker than the astrophysical limits and do not reach the level of sensitivity to constrain the QCD

axion-induced nucleon-nucelon monopole-dipole and monopole-monopole interactions. However, the proposed method of Ref. [352] using Nuclear Magnetic Resonance to probe the monopole-dipole interaction could cover a wide range corresponding to the entire classic axion window, $10^9 \text{ GeV} \lesssim f_a \lesssim 10^{12} \text{ GeV}$. Despite its tiny value, the scalar coupling of the QCD axion offers a very promising avenue for discovery.

9.5. Direct Detection of Axion DM

9.5.1. Haloscopes and ADMX

Let's begin with the classic haloscope experiments [333], which search for DM axions using the $g_{\phi\gamma}$ coupling. A haloscope currently in operation is the Axion Dark Matter eXperiment (ADMX) [88].

A DM axion enters a microwave cavity, where it interacts with an applied magnetic field, converting into a photon which is then detected. The cavity geometry is tuned such that this conversion is resonant, enhancing the conversion rate. The power generated in the cavity is

$$P = g_{\phi\gamma}^2 \frac{V B_0 \rho_a C}{m_a} \min\left(Q, Q_a\right), \qquad (188)$$

where ρ_a is the local DM density in axions, V is the cavity volume, B_0 is the applied magnetic field strength, Q is the quality factor of the cavity, Q_a is the ratio of the halo axion energy to energy spread, and C is a mode dependent form factor for the cavity. For approximate ADMX parameters V = 500 L, $B_0 = 7$ T, $Q = 10^5$, in the classic QCD axion window with $f_a \approx 10^{12}$ GeV, the power is $P \approx 10^{-21}$ W.

Since ADMX is a DM detector, it also relies on ρ_a being large, and quoted constraints assume that axions in its sensitivity range compose all the DM. Because of the resonant tuning required, ADMX is very precise, but is only able to probe a narrow range in the mass-coupling plane (see Fig. 29). ADMX is sensitive to axions with $m_a \approx 10^{-6}$ eV. Current constraints exclude ALPs of this mass more strongly coupled to photons than the QCD axion. In the near



Figure 30: Future reach of axion DM direct detection experiments using the two-photon coupling. The classic window of the QCD axion can be completely covered. Reproduced, with permission, from "Resonant Dark Matter Detectors Beyond 10 GHz," Gray Rybka, PATRAS10 (2014).

future ADMX will able to probe most of the model space (KSVZ and DFSZ) for the QCD axion with $10^{-6} \text{ eV} \lesssim m_a \lesssim 10^{-5} \text{ eV}$, i.e. $f_a \sim 10^{12} \text{ GeV}$.

Other upgrades and new proposals for axion DM direct detection experiments in the classic QCD axion window using the two-photon coupling include the use of open resonators (the ORPHEUS experiment) [353], LC-circuits [354] and broadband searches with SQUIDs [355]. Projections for some of these techniques are shown in Fig. 30, and could cover the mass range 10^{-8} eV $\leq m_a \leq$ 10^{-2} eV of the QCD axion.

9.5.2. Nuclear Magnetic Resonance and CASPEr

The Cosmic Axion Spin Precession Experiment (CASPEr) [235], comes in two varieties. Both strategies are novel, as they do not rely on the "standard" two-photon coupling. Each CASPEr experiment uses the property that the axion couplings to nucleons are spin dependent. The interactions can be detected by spin-polarizing a sample in an applied magnetic field, and searching

ACCEPTED MANUSCRIPT

 $\begin{array}{r} 45\\ 46\\ 47\\ 48\\ 50\\ 51\\ 52\\ 53\\ 54\\ 55\\ 56\\ 57\\ 58\\ 59\\ \end{array}$



Figure 31: Sensitivity of the CASPEr experiments, assuming the DM is contained exclusively in a single ALP. CASPEr is a resonant experiment and sensitivity assumes a 3 year operation of scanning. *Left Panel:* CASPEr-Electric and the nucleon EDM coupling. Orange shaded: phase 1. Red shaded: phase 2. Dashed red: magnetometer noise limit in phase 2. *Right Panel:* CASPEr-Wind and the axial nucleon moment (note their g_N is our \tilde{g}_N). Red: Xe sample. Blue: ³He sample. Dashed lines: magnetization noise limits. Reproduced (with permission) from Refs. [235, 47]. Copyright (2014,2013) by The American Physical Society.

for spin-precession using nuclear magnetic resonance techniques. The induced magnetization is resonant at the Larmour frequency of the applied magnetic field, $2\mu_m B_{\text{ext}} = m_a$ (where μ_m is the nuclear magnetic dipole moment) and is detected using a SQUID magnetometer. For reasons that will become apparent, we refer to the two distinct experiments as "CASPEr-Electric" [235] and "CASPEr-Wind" [47]. Just like with ADMX, CASPEr is a DM detector and the sensitivity to axions scales with the DM abundance. CASPEr has not yet been constructed, and we discuss projected sensitivities.

CASPEr-Electric exploits the axion coupling to (ϕ/f_a) Tr $G\tilde{G}$, which gives rise to the EDM coupling, g_d . CASPEr-Electric thus explores the defining property of the QCD axion. The dipole moment induced by an axion is $d_n = g_d \phi$. Recall that the QCD axion solves the strong-CP problem by setting the *timeaverage* of the nucleon EDM to zero, as required by experiments constraining the static EDM [20]. The same oscillations in the axion field that allow it to function as a DM candidate, however, lead to *EDM oscillations*, $d_n \sim 10^{-16} (\phi/f_a) \cos(m_a t) e \,\mathrm{cm}$, where ϕ is the *local* value of the axion field amplitude. CASPEr-Electric applies an electric field to a spin-polarized sample and detects the precession of the nuclear spins about the \vec{E} field axis caused by the non-zero EDM.

The projected sensitivity of CASPEr-Electric is shown in Fig. 31, Left Panel. In phase 2 CASPEr-Electric will be able to detect the QCD axion for $f_a \gtrsim 10^{16}$ GeV, with ultimate limits from magnetization noise able to reach $f_a \gtrsim 3 \times 10^{13}$ GeV. CASPEr-Electric is thus highly complementary to ADMX and astrophysical bounds.

CASPEr-Wind exploits the axion coupling to the axial nuclear current, $g_{\phi N}$, and the induced spin-dependent force. As the earth moves relative to the DM halo of our galaxy, so we experience a "DM wind" of axions. The effective coupling in the nucleon Hamiltonian is $H_N \supset \tilde{g}_{\phi N} m_a \phi \cos(m_a t) \vec{v} \cdot \vec{\sigma}$, where $\vec{\sigma}$ is the nuclear spin, and \vec{v} is the DM wind velocity. The spin-dependent force creates a torque around the direction of the DM wind and leads to spin precession of nuclei without the need for an applied electric field. CASPEr-Wind

is thus somewhat simpler to implement than CASPEr-Electric.

The projected sensitivity of CASPEr-Wind is shown in Fig. 31, Right Panel. While CASPEr-Wind is not sensitive to the QCD axion (except in the noiselimited regime), it is sensitive to the ULA model of Ref. [46], and is complementary to cosmological axion searches.

9.6. Heavy Axions and Axion Decays

In this section we consider constraints on axions with masses $m_a \gg 1$ eV. Note that the constraints summarised in Fig. 29 (and much of the phenomenology discussed elsewhere in this review) typically do not apply to such high masses, as they rely on the coherence of the axion field. The bounds from stellar astrophysics in Section 9.1 can apply for m_a as large as 1 keV. We consider primarily the astrophysical and cosmological consequences of axion decay, but mention some other constraints in passing.

Consider the axion-photon coupling, $g_{\phi\gamma}$, defined in Eq. (23), which we recall has mass-dimension -1, and is in general a free parameter for ALP models, with approximate scale $1/f_a$. This coupling allows axions to decay into two photons, with a lifetime:

$$\tau_{\phi\gamma} = \frac{64\pi}{m_a^3 g_{\phi\gamma}^2} \approx 130 \text{ s} \left(\frac{\text{GeV}}{m_a}\right)^3 \left(\frac{10^{-12} \text{ GeV}^{-1}}{g_{\phi\gamma}}\right)^2.$$
(189)

Consider the KSVZ axion, with the photon coupling fixed by Eq. (27). Taking the age of the Universe to be $\tau_{\rm univ.} \approx 10^{10}$ years we find that the QCD axion is stable on the lifetime of the Universe for $f_a \gtrsim 1.9 \times 10^6$ GeV. Thus, the QCD in the allowed range of f_a is stable on the lifetime of the Universe, and hence is a DM candidate.

ALPs, on the other hand, may decay on much shorter time scales. The coupling of ALPs is in general proportional to the mass, since couplings go as $1/f_a$ and $m_a = \Lambda_a^2/f_a$. Thus heavier ALPs can be unstable on cosmological timescales and will decay to standard model particles (or light dark sector particles). The decay of such a population of ALPs injects additional relativistic energy density into the Universe, which is constrained by a number of probes.

We will closely follow the recent compilation of constraints in Ref. [356], as shown in Fig. 32. Some early constraints on ALPs from decays can be found in Refs. [342, 357], while further reading can be found in Ref. [358] (for general physics and consequences of decaying particles, see Ref. [359]).

The presence and later decay of ALPs in the early Universe can change the effective number of relativistic species, N_{eff} (Eq. 44), and the baryon-to-photon ratio, $\eta_b \equiv n_b/n_\gamma$, at different times in cosmological history. A lower value of $N_{\text{eff}}^{\text{CMB}}$ affects the CMB power spectrum, as discussed in Section 3.3.1. The baryon ratio at the CMB is well measured, fixind $\eta_b^{\text{CMB}} = 2.74 \times 10^{-8} \Omega_b h^2$. The photon energy density is also fixed by the equally well measured T_{CMB} . Therefore ALP decays can actually *reduce* N_{eff} and *increase* η_b . An ALP decaying between BBN and the CMB reduces $N_{\text{eff}}^{\text{CMB}}$ if the decay occurs after neutrino decoupling, by heating of the plasma.⁵¹ Decay before BBN also reduces $N_{\text{eff}}^{\text{BBN}}$ is increased. ALP decay between BBN and the CMB reduces are themselves relativistic at BBN, $N_{\text{eff}}^{\text{BBN}}$ is increased. ALP decay between BBN and the CMB reduces $N_{\text{eff}}^{\text{BBN}}$ and the CMB reduces η_b^{BBN} is a compared to η_b^{CMB} .

Changes of the expansion rate , via $N_{\rm eff}$, and baryon abundance during BBN affect the light element abundances. The standard model predictions of the BBN light element abundances are extremely well verified (with the famous exception of Lithium): see Refs. [42, 360] for reviews. The helium abundance, Y_p and the deuterium-to-hydrogen ratio, D/H, place strong constraints on ALPs, both from decays and from the contribution of thermally produced axions with $m_a \lesssim 1$ MeV to the radiation density at BBN.

Energy injections at different epochs can also change the shape of the CMB frequency power spectrum, such that it is no longer a perfect black body. Such effects are known as *CMB spectral distortions*, and are strongly constrained by

 $^{^{51}}$ It is interesting to note the opposite effects of different ALPs on $N_{\rm eff}$: decay of a heavy particle to an ALP leads to an increase, while decay of a heavy ALP to photons leads to a decrease. The effects of light and heavy ALPs and moduli could conspire to hide them from our view.

ACCEPTED MANUSCRIPT



Figure 32: Constraints on heavy ALPs from decays, in the mass-lifetime plane. The axion mass is here labelled m_{ϕ} . The CMB, D/H, and Yp regions are excluded at 3σ , the Collider and Beam Dump regions are excluded at 2σ , and the SN1987a and HB Stars regions are less formal. Reproduced (with permission) from Ref. [356].

the COBE-FIRAS measurements (for a review, see Ref. [228]). Early decays of axions heat the plasma leading to distortions of " μ -type" (chemical potential) for decays between $10^5 \leq z \leq 10^6$, or "y-type" (Compton scattering) for decays between $1100 \leq z \leq 10^5$. These effects are computed in e.g. Refs. [356, 358, 357].

In the life-time range of relevance to cosmological axion decays, the axionphoton coupling also has collider signatures, allowing, for example, single-photon final states in electron-positron colliders. The constraint from LEP [361, 362, 363] is [35]

$$g_{\phi\gamma} < 4.5 \times 10^{-4} \text{ GeV}^{-1}$$
 (LEP: $m_a \lesssim \text{ GeV}$). (190)

In fact, a stronger bound due to the single photon final state was derived much earlier, using ASP data [364] in Ref. [342]: $g_{\phi\gamma} \leq 5.5 \times 10^{-4} \text{ GeV}^{-1}$ for $m_a \gg 29 \text{ GeV}$. Anomalous decays of heavy quark states lead to similar bounds.

The summary of these constraints is shown in Fig. 32. The DFSZ and KSVZ axion models are excluded for m_a in the keV to MeV range, as are most ALPs with

$$1 \text{ keV} \lesssim m_a \lesssim 1 \text{ GeV},$$
 (191)

$$10^{-4} \text{ s} \lesssim \tau_{\phi\gamma} \lesssim 10^6 \text{ s}.$$
 (192)

There is an open window for short-lived, $\tau_{\phi\gamma} < 0.01$ s, heavy, $m_a \gtrsim 1$ GeV, ALPs that decay early enough and are sufficiently non-relativistic at BBN to not alter the light element abundances.

9.7. Axion Dark Radiation

We discussed in Section 3.3.1 how a population of relativistic axions can be created by decay of a modulus. The CMB power spectrum and other cosmological observables constrain the simplest consequence of this: the relativistic axion energy density, parameterized by $\Delta N_{\rm eff}$. This population of axions, if coupled to the standard model, can also be probed by axion scattering.

If the modulus decay that produced the axion DR also reheats the Universe, then the axion energy is $E \sim m_{\sigma} \sim T_{\gamma} \sqrt{M_{pl}/m_{\sigma}} \gg T_{\gamma}$. Because the energy is



Figure 33: Constraints on axion DR from the primordial helium abundance [365]. The fermion interaction is taken to have strength $c_f = 1$, and here m_{ϕ} is the modulus mass. Final states to $b\bar{b}$ (solid), $c\bar{c}$ (dashed) and $s\bar{s}$ (dot-dashed) are considered, with varying amounts of DR, $\Delta N_{\rm eff} = 0.1, 0.5, 1$ (green, black, red; corrected labelling from typo in original). Areas below curves are excluded. Reproduced (with permission) from Ref. [108].

much higher than the plasma temperature, this gives access to processes that are otherwise kinematically forbidden. This leads to interesting constraints and phenomenology despite the f_a -suppressed axion couplings. Ref. [108] discussed the phenomenology in detail.

An axion-fermion coupling of the form $\mathcal{L}_f = c_f m_f \phi \bar{\psi} \gamma^5 \psi / f_a$ (this form can be obtained from the axial current interaction in Eq. 23 by use of the equations of motion) allows for production of heavy fermions via the process $a + \gamma \rightarrow f + \bar{f}$. The secondary decay of the fermions can alter the proton to neutron ratio during BBN, and thus the primordial helium abundance. Each axion scattering process can be mapped onto an "effective decay process" [108] for which constraints can readily be found in the literature (e.g. Ref. [365]). The constraints are shown in Fig. 33. Taking $c_f = 1$, BBN constraints rule out values of $f_a \leq 10^9$ GeV over a wide range of modulus masses.

Axion DR also has a flux at Earth and, if the axion-photon coupling is non-

vanishing, could be detected by helioscopes like CAST. The axion DR flux is distinct from the solar flux in two important ways: firstly, because of its cosmological origin, it is isotropic; secondly, the DR flux is not suppressed by as many powers of $g_{\phi\gamma}$, due to the different production mechanism compared to solar axions. Taking $g_{\phi\gamma} \sim f_a^{-1}$, the DR signal in a heliscope is thus suppressed as only f_a^{-2} , compared to the f_a^{-4} suppression for solar axions. For a modulus mass of $m_{\sigma} = 5 \times 10^6$ GeV and $\Delta N_{\text{eff}} \approx 0.6$ the flux is $\Phi_a \approx 1.09 \times 10^6$ cm⁻²s⁻¹ [108], which is of order the solar QCD axion flux for $f_a = 10^{10}$ GeV. The DR background in this model is thus in reach of IAXO. For these same parameters, the energy spectrum peaks in the keV range, and has a form characteristic of the axion DR background from modulus decay.

9.8. Axions and Astrophysical Magnetic Fields

Let's further consider the Primakoff process, but now for the case of ULAs in the presence of astrophysical magnetic fields. Gamma rays from blazars suggest that the cosmic background field exceeds $B \sim 10^{-16}$ G in large voids [366, 367], while it could be large as nG, with Mpc coherence length. Larger magnetic fields are present in clusters of galaxies, with strength $B \sim \mu$ G and coherence length of order kpc.

9.8.1. CMB Spectral Distortions

In the presence of a background magnetic field axion photon mixing occurs and, just like in the case of massive neutrinos, propagation and interaction eigenstates are not the same. Furthermore, plasma effects lead to an effective photon mass:

$$m_{\gamma}^2 = \omega_p^2(z) - 2\omega^2(n_H - 1), \qquad (193)$$

where ω is the photon frequency, and the refractive index of neutral hydrogen is n_H . The plasma frequency, ω_p , depends on the free electron density, and is thus a function of redshift determined by recombination and reionization. At $\omega = T_{\rm CMB}$ the photon plasma mass at z = 0 is $m_{\gamma} \sim 10^{-14}$ eV. Resonant axionphoton conversion occurs when $m_{\gamma} = m_a$. Since for high frequency photons m_{γ}^2



Figure 34: Constraints on ULAs from CMB spectral distortions from FIRAS, and projected for PIXIE/PRISM. The axion mass is labelled m_{ϕ} in this plot. The dark band shows masses where multiple resonant conversions effectively exclude such axions entirely, for $g_{\phi\gamma} \neq 0$. Reproduced (with permission) from Ref. [371]. Copyright (2013) by The American Physical Society.

passes through zero, resonant conversion can occur for arbitrarily low axion mass, and can occur multiple times as m_{γ}^2 changes sign.

The frequency dependence of the resonant conversion epoch leads to a spectral distortion [368]. COBE-FIRAS [369, 370] measured the CMB to be a black body to high precision. This constrains the resonant conversion probability, which in turn leads to a constraint on the product $g_{\phi\gamma}B_0$, where B_0 is the spatially averaged magnetic field strength today.

The constraints have been addressed in detail in Refs. [372, 371]. Fig. 34 shows the constraints on ULAs from FIRAS, and projected constraints from a PIXIE [373]/PRISM [374]-like mission. Multiple resonant conversions occur for $10^{-14} \text{ eV} \leq m_a \leq 10^{-12} \text{ eV}$, effectively excluding any $g_{\phi\gamma} \neq 0$ for this mass range. While constraints are only on the product $g_{\phi\gamma}B_0$, they are stronger than the product of current upper limits on $g_{\phi\gamma}$ and B_0 individually.

9.8.2. X-ray Production

As discussed a number of times, axion DR can be produced by the decay of a modulus, and the axion DR energy today is $E_0 \sim T_{\rm CMB} \sqrt{M_{pl}/m_{\sigma}}$. For a modulus mass $m_{\sigma} \sim 10^6$ GeV (suggested by string theory solutions to the EW hierarchy problem) this gives rise to a cosmic axion background (CAB) with energy $E \sim 0.1 - 1$ keV. The energy density in the CAB is

$$\rho_{\rm CAB} = 1.6 \times 10^{60} \text{ erg Mpc}^{-3} \left(\frac{\Delta N_{\rm eff}}{0.57}\right),$$
(194)

Conversion of the CAB to photons in the presence of magnetic fields leads to production of X-rays.

Clusters of galaxies are permeated by magnetic fields with $B \sim \mu G$ and coherence lengths $L \sim \text{kpc.}$ Axion-photon conversion in this environment is predicted to lead to excess X-ray emission from clusters [108, 375]. The Xray luminosity of a typical Mpc sized cluster is $\mathcal{L}_{\text{cluster}} \sim 10^{44} \text{ erg s}^{-1}$. The excess soft X-ray luminosity in Coma is $1.6 \times 10^{42} \text{ erg s}^{-1}$ [376], which could plausibly be explained with an axion-photon coupling $g_{\phi\gamma} \sim 10^{-14} \text{ GeV}^{-1}$ [375], depending on the axion mass and the photon plasma mass in the intra-cluster medium. This emission has fixed redshift scalings, since the CAB is cosmological in origin. It is also predicted to correlate with cluster magnetic fields, unlike an annihilating DM signal.

Ref. [377] considered X-ray production within galactic magnetic fields. For the strength of coupling required to explain the soft X-ray excess in Coma, conversion within the Milky Way is negligible. Star burst galaxies, with larger magnetic fields, may produce an observable signal, in particular if the inhomogeneous free electron density is accounted for in modelling the emission.

Conversion in cosmological magnetic fields could contribute to an unresolved cosmic X-ray background. This is essentially the inverse of the spectral distortion effect discussed in the previous subsection, with a different energy spectrum. A diffuse cosmic X-ray background in the keV energy range is observed [378], with diffuse intensity that could be explained by the CAB with $g_{\phi\gamma} \sim 10^{-13} \text{ GeV}^{-1}$, assuming nG strength cosmological magnetic fields [375].
From Fig. 34 we see that this explanation for the X-ray background will in addition produce a CMB spectral distortion close to the FIRAS bound, and observable with PIXIE/PRISM.

9.9. Cosmological Birefringence

CMB polarization comes in E-modes and B-modes. E-modes are generated from temperature fluctuations at last scattering by the quadrupole anisotropy, and the E spectrum can be predicted from the measurement of the adiabatic temperature fluctuations. B-modes can be generated in three ways: primordially, by tensor fluctuations with relative amplitude r_T ; by gravitational lensing along the line of sight; and finally by the birefringent effect, rotating of E into B.

In the presence of the axion-photon coupling in Eq. (23), the fields satisfying free wave equations are $\vec{D} = \vec{E} + \frac{g_{\phi\gamma}}{2}\phi\vec{B}$ and $\vec{H} = \vec{B} - \frac{g_{\phi\gamma}}{2}\phi\vec{E}$ [380] (note \vec{E} and \vec{B} are the fields of electromagnetism, and are not the same as E and B mode polarization). Therefore, if the axion field ϕ varies in time or space it can cause rotation of the plane of polarization of the CMB [381]:

$$\Delta\beta = \frac{g_{\phi\gamma}}{2} \int d\tau \phi' \,, \tag{195}$$

where it is reminded that τ is conformal time, and primes denote derivatives with respect to this. The integral is performed along the line of sight from the surface of last scattering at $z_{\rm dec}$ to today. When the axion is oscillating, the integral vanishes. Therefore, significant rotation only occurs for ULAs that begin oscillations after photon decoupling. Using $z_{\rm dec} = 1020$, $\Omega_m = 0.31$, $\Omega_{\Lambda} = 0.69$, h = 0.67, we find that ULAs with a mass $m_a \lesssim 3H_{\rm dec} = 1 \times 10^{-28}$ eV can cause significant cosmological birefringence.

The uniform misalignment of ULAs in the broken PQ scenario (see Section 3.2.2) leads to a uniform rotation of the plane of CMB polarization. Such a uniform rotation is constrained to be $|\Delta\beta| < 1.4 \times 10^{-2}$ [382]. If we assume $\phi(\tau_0) = 0$, this gives the approximate constraint $\phi_i g_{\phi\gamma} < 2.8 \times 10^{-3}$. Taking $g_{\phi\gamma} \sim \alpha_{\rm EM} f_a^{-1}$, CMB polarization rotation imposes a constraint on the (ϕ_i, f_a)



Figure 35: CMB B-mode power from birefringence caused by ULAs coupled to magnetic fields with $(H_I g_{\phi\gamma})^2 \approx 0.17$ (red, solid). The large angle signal can mimic tensor modes with $r \sim 0.1$ (blue, short dashed), while the small angle signal contains distinctive BAO from the E-modes (green, dot-dashed) and, for this choice of parameters, dominates over lensing power (cyan, long dashed). Reproduced (with permission) from Ref. [379] (where the data are described). Copyright (2009) by The American Physical Society.

plane. For ULAs, and using Eq. (61) for $a_{\rm eq} < a_{\rm osc}$, the birefringence constraint is of order the constraint on the DM abundance from temperature anisotropies (Fig. 15, right panel), assuming $f_a < M_{pl}$ and $m_a \lesssim 1 \times 10^{-28}$ eV (excluding also the lightest DE like axions where $\phi(\tau_0) \neq 0$). Thus, if a sub-dominant population of such ULAs is detected in LSS in future, e.g. by *Euclid* (Fig. 14), then this may well be accompanied by birefringence in the CMB.

Anisotropies in the axion field cause anisotropic rotation. This leads to generation of BB anisotropy power from EE, and thus EB cross-correlations, and can be significantly sourced by ULA isocurvature pertrubations: see e.g. Refs. [383, 382, 379, 384, 385]. The resulting CMB power spectra are shown in Fig. 35. The amplitude of the power spectrum scales as $(H_I g_{\phi\gamma})^2$. This effect is particularly interesting as it can generate B-modes that dominate over those produced by tensor perturbations. This could source large angle B-mode power in low-scale inflation if $H_I g_{\phi\gamma} \sim 0.1$. Since the power is generated from the E-modes, there is also oscillating, large amplitude, small-angle B-power in this scenario. This would be present even after de-lensing and is distinct from the tensor mode power, which falls rapidly on small angular scales.

The most recent constraints on anisotropic birefringence come from the Bmode power and 4-point function measured by POLARBEAR [386]. These constraints are consistent with zero signal.

10. Concluding Remarks

In this review we have presented the vast cornucopia of axion physics. We have considered the motivations and models for axions coming from particle physics and string theory. We have seen how axions can be produced in the early Universe by a variety of mechanisms. Axions can play important roles in all of the unsolved mysteries of cosmology: inflation, dark matter, and dark energy. They also lead to novel phenomena, such as fuzzy dark matter, and dark radiation. Axion couplings to the standard model are fixed by symmetry considerations, and can be computed in specific models. We studied the tailored direct and indirect searches for axions, which are quite different to more "standard" searches for new particle physics.

I hope, dear reader, that you have come away from this review with a sense for the fascinating progress that has been made in axion physics over the last years and decades. I also hope that you can see the places on the horizon where new opportunities are arising. Let me briefly reiterate some of these:

- The dark sector and large scale structure: Soon, large scale structure measurements will reach the precision to test in detail aspects of standard neutrino physics, such as the neutrino mass, and number of neutrinos. Axions share many degeneracies with the neutrino sector. Misalignment-produced ULAs suppress structure formation on cluster scales; hot axions contribute to dark radiation either via thermal production or via modulus decay. Improved measurements and studies of CMB polarization and gravitational lensing of galaxies could easily discover these effects at the same time as testing neutrino physics. Breaking degeneracies via multiple probes is an important endeavour for both axion and neutrino physics.
- Axions with $m_a \sim 10^{-22}$ eV and the CDM small-scale crises: The CDM small-scale crises, if they are indeed crises, can be solved by ULAs. Observational and simulation techniques on these scales are always improving, and axion physics must keep up. There are some simulations on the market, but the field has not been studied in anywhere near as much depth as competing models, such as WDM. The tantalizing prospect to see evidence for axions on these scales, in galactic dynamics and in the epoch of reionization, must not be overlooked, and much work is necessary to exploit this opportunity.
- Progress in string theory model building and the axiverse: A large part of the motivation to study axions comes from their apparent prevalence in string theory. In principle, therefore, constraints on axions can be interpreted as constraints on string theory. There is already a large program

of model building in this direction. The focus has largely been on inflation, but extensions to other parts of cosmology are slowly being made. This model building should also be done holistically, with emphasis on the many different facets of axion physics that combine and provide the opportunity to make unique and verifiable predictions.

• Novel experiments for axion direct detection: Axion direct detection has, for many years, focused on the $\vec{E} \cdot \vec{B}$ coupling and the QCD axion. Recent years have seen an upsurge in interest in searching for the other possible axion couplings in terrestrial experiments. These searches are more generally applicable to ALPs, which may only possess a fraction of the couplings allowed by symmetry, for example having no coupling to photons. All direct searches for axions provide vital information to cosmology, not least by limiting the decay constant in specific models, but also by allowing the possibility to actually identify the DM as axion-like by the form of its couplings.

This summary is not the end. Axion physics is alive and well, and growing: long may it be so.

Acknowledgments. This review was prepared in part for a lecture presented at the mini-workshop "Axion Theory and Searches" at IPhT CEA/Saclay. I would like to thank the organisers of the workshop, Marco Cirelli, Bradley Kavanagh and Filippo Sala, for inviting me to lecture, and the other lecturers, Joerg Jaeckel and Pierre Sikivie, for stimulating discussion. I thank all my collaborators on the work presented here: Brandon Bozek, Malcolm Fairbairn, Pedro Ferreira, Daniel Grin, Renée Hlozek, Robert Hogan, Luca Iliesiu, Edward Macaulay, Kavilan Moodley, Ana Pop, Joseph Silk, Hiroyuki Tashiro, Maxime Trebistch, Scott Watson and Rosemary Wyse. Special thanks to the authors of Refs. [74, 328, 108, 16, 257, 278, 356, 379, 298, 235, 47, 300] for permission to reproduce their figures. I thank Jihn Kim and Maxim Pospelov for discussions on the particle theory of axions, Thomas Bachlechner for discussion on the weak gravity conjecture, Sacha Davidson for discussion of BEC's, and Renée Hlozek and Cliff Burgess for reading parts of the manuscript. Finally, I thank the anonymous referee, whose comments helped greatly improve the breadth of this work. This work was supported at Perimeter Institute by the Government of Canada through Industry Canada and by the Province of Ontario through the Ministry of Research and Innovation; and by a Royal Astronomical Society research fellowship, hosted at King's College London.

A. Theta Vacua of Gauge Theories

I will simply state some relevant results to give you a feel for this topic: see the wonderful book by Coleman, Ref. [18], for the gory details. I follow Coleman's notation and normalisation in this discussion.

Quantum theory depends on the Euclidean functional integral, with the path integral being dominated by field configurations of finite Euclidean action. These dominant contributions to the semi-classical approximation are known as *instantons*. The action for a gauge field theory with gauge group G (for definiteness, take G = SU(N)) and gauge coupling g_G in 4 flat Euclidean dimensions is

$$S = \frac{1}{4g_G^2} \int d^4 x (F_{\mu\nu}, F_{\mu\nu})$$
(A1)

A field configuration of finite action must have $F \sim O(1/r^3)$ as $r \to \infty$ and so the gauge field must be of the form

$$A_{\mu} = g \partial_{\mu} g^{-1} + O(1/r^2) \,, \tag{A2}$$

for some gauge transformation g(x), which is a function mapping G to the variables of Euclidean 4-space. In order not to alter the asymptotic baheviour in r we must have that g(x) maps G to only the angular variables. That is, the field configurations are defined up to a mapping of G to the space-time boundary, which in this case is topologically the three-dimensional hypersphere, S^3 .

How many different mappings are there, and how can we classify them? Firstly, we can always make a gauge transformation by some other element h, which is a continuous function, and continuously deform one into another. That means that all *homotopoically equivalent* mappings are equivalent field configurations.⁵² We now need to classify the homotopically distinct mappings.

A theorem [387] states that we need only consider the SU(2) subgroups of our group G. SU(2) is topologically S^3 , and so one such mapping is the trivial mapping

$$g^{(1)}(x) = (x_4 + i\vec{x} \cdot \vec{\sigma})/r$$
, (A3)

where $\sigma_{1,2,3}$ are the Pauli matrices. It is then also possible to prove (Coleman does not prove it, and I certainly won't) that all mappings from S^3 to S^3 are homotopic to a family of mappings

$$g^{(\nu)}(x) = [g^{(1)}(x)]^{\nu},$$
 (A4)

where ν is an integer called the *winding number*. For the simple example of wrapping U(1) round a circle, this is easy to visualise, and ν labels the representations of U(1) as $e^{i\nu\theta}$, with θ the angle on S^1 .

Finally, it is possible to show that the winding number of a field configuration is given by the integral

$$\nu = \frac{1}{32\pi^2} \int d^4 x(F, \tilde{F}) \,, \tag{A5}$$

where \tilde{F} is the dual field strength as defined below Eq. (2). The winding number is a topological invariant of the field configuration, providing a finite contribution to the Euclidean action proportional to the integral of Eq. (2).

The winding number describes the boundary conditions of the gauge fields with $\nu = n$ in some state $|n\rangle$. The vacuum of the theory is given by a superposition of states

$$|\theta\rangle = \sum_{n} e^{in\theta} |n\rangle \,. \tag{A6}$$

such that

$$\langle \theta | e^{-HT} | \theta \rangle \propto \int [dA] e^{-S} e^{i\nu\theta} \,.$$
 (A7)

⁵²An important consequence of this is the fact that U(1) gauge theory has no instantons in 3+1 dimensions. U(1) is topologically the circle, S^1 , which, when wrapped around S^3 , can be continuously deformed to a single point: the trivial mapping.

A $\theta\text{-vacuum}$ is thus described by a term in the action

$$S_{\theta} = \frac{\theta}{32\pi^2} \int d^4 x(F, \tilde{F}) \,. \tag{A8}$$

All the θ -vacua are topologically distinct, and transitions between them are forbidden as they involve discontinuous changes in the gauge field boundary conditions.

By considering a gas of n instantons and \bar{n} anti-instantons, such that $\nu = n - \bar{n}$, Coleman goes on to show that

$$\langle \theta | e^{-HT} | \theta \rangle \propto \exp[e^{-S_0} \cos \theta],$$
 (A9)

so that the energy of the $\theta\text{-vacuum}$ is

$$E(\theta) \propto e^{-S_0} \cos \theta$$
, (A10)

with the one-instanton action

$$S_0 = \frac{8\pi^2}{g_G^2} \,. \tag{A11}$$

B. EFT for Cosmologists

This is an extremely heuristic description of EFT. For a rigorous treatment, see e.g. Ref. [388].

The general notion of EFT is based on the idea that at low energies, q, we can replace a "fundamental" action, S, with an effective action, $S_{\text{eff}}(q)$. In the jargon, this is thought of in terms of the Wilsonian picture of the renormalization group: we define an action in the UV at a scale Λ_{UV} and then use the renormalization group equations to "run" down to $q < \Lambda_{\text{UV}}$. This is referred to as "integrating out" fields with masses m > q. Quantum field theory (e.g. Refs. [389, 390]) allows for interactions mediated by virtual particles, and when these particles are integrated out this leads to effective interactions in the low-energy theory that were not present in the UV theory.

Consider the case of the Fermi interaction, represented in Fig. B.36 for muon decay. In the EW theory we know that, at a fundamental level, charged lepton-neutrino interactions are governed by a term in the action $S \supset ig_2 W_\mu \bar{\ell}_i \gamma^\mu \nu_i +$



Figure B.36: Muon decay and the Fermi interaction as an example of EFT. The fundamental theory involves exchange of virtual W bosons with momentum q^{μ} . At low-momentum transfer, $q^2 \ll m_W^2$, the interaction can be replaced with the effective 4-fermion interaction proportional to G_F .

h.c., where g_2 is the EW coupling constant, ℓ_i is the charged lepton field, ν_i its corresponding neutrino, and W_{μ} is a charged W boson. This allows for W^{\pm} particles to mediate muon decay (recall that a similar process involving quarks and the CKM matrix elements mediates nuclear β -decay, and was the original use of the Fermi interaction). The exchanged 4-momentum is q^{μ} , and the Wboson propagator is proportional to $1/(q^2 + m_W^2)$, where $m_W = 80.4$ GeV [42] is the mass of the W. At small momentum transfer, $q^2 \ll m_W^2$ (corresponding via the uncertainty principle to large distances) the propagator can be replaced by an effective 4-fermion interaction proportional to g_2^2/m_W^2 . Higher order interactions come suppressed by higher powers of m_W . In the low-energy EFT we replace the EW gauge invariant interaction with the Fermi interaction using $G_F = \sqrt{2}g_2^2/8m_W^2$. For muon decay, the low energy theory has a term in the effective action $S_{\text{eff}}(q < m_W) \supset G_F(\bar{e}\nu_e)(\bar{\nu}_{\mu}\mu) + h.c.$

The situation with axions and the chiral anomaly is more complicated to compute, but is easy to represent in pictures. The case of the KSVZ axion model is shown in Fig. B.37. The fundamental action contains Yukawa interactions



Figure B.37: The colour anomaly in the KSVZ axion model. Heavy quarks, Q, run in a loop with momentum q^{μ} . At low-momentum transfer, $q^2 \ll m_Q^2$, the interaction can be replaced with the effective $\phi G \tilde{G}/f_a$ interaction.

between the axion and the heavy quark fields, Q. The Q fields also interact with gluons. Virtual Q-particles then induce an effective axion-gluon interaction at loop-level. At low momentum transfer, $q^2 \ll m_Q^2$, the heavy quarks can be integrated out and the effective action has a term $S_{\text{eff}}(q < m_Q) \supset \phi G \tilde{G}/32\pi^2 f_a$. This is the dominant term in the expansion in powers of $1/m_Q$. It gives the largest contribution to the explicit breaking of $U(1)_{\text{PQ}}$, and thus the axion potential, and also generates the necessary $G \tilde{G}$ interaction required for a solution to the strong-CP problem. EFT can also be applied to light quarks after chiral symmetry breaking. This gives rise to the second term in Eq. (25), which gives a contribution to the axion-photon coupling from the colour anomaly.

C. Friedmann Equations

Consider the line element for the flat Friedmann-Robertson-Walker (FRW) Universe:

$$ds^2 = -dt^2 + a(t)^2 d\vec{x}^2 \,, \tag{B1}$$

where a(t) is the cosmic scale factor. The scale factor obeys the Friedmann equations:

$$\begin{aligned} 3H^2 M_{pl}^2 &= \bar{\rho} \,, \\ 6(\dot{H} + H^2) M_{pl}^2 &= \bar{\rho} + 3\bar{P} \,, \end{aligned} \tag{B2}$$

where $H = \dot{a}/a$ is the Hubble rate, $\bar{\rho}$ and \bar{P} are the homogeneous background values of the components of the energy momentum tensor as defined in Eqs. (C1), and homogeneity and isotropy of the FRW metric demand the vanishing of velocity and anisotropic stress at the background level. The current cosmic time is $t = t_0$, and the current Hubble rate is $H(t_0) \equiv H_0 =$ $100h \text{ km s}^{-1} \text{ Mpc}^{-1} = 2.13h \times 10^{-33} \text{ eV} = hM_H$. Normalising $a(t_0) = 1$, the redshift is given by z = 1/a - 1. The scale factor and redshift can both serve as useful time co-ordinates.

Cold (C)DM, baryons and non-relativistic massive-neutrinos have zero pressure, and the energy density in matter scales as $\bar{\rho}_m = \bar{\rho}_{m,0}a^{-3}$. Radiation, including photons and relativistic neutrinos, has pressure $\bar{P}_r = \bar{\rho}_r/3$ and the energy density scales as $\bar{\rho}_r = \bar{\rho}_{r,0}a^{-4}$

The first of Eqs. (B2) is commonly known as the Friedmann equation, while the second is known as the Raychaudhuri equation. The Friedmann equation is a first order constraint, and is sufficient to solve the background evolution in the case of a flat or open universe with positive energy density. The Raychaudhuri equation is only necessary to solve for collapsing universes (closed, or an AdS scalar field potential), although there are occasions when it is more numerically stable than the Friedmann equation.

D. Cosmological Fluids

Useful references for this section include Refs. [149, 131, 159, 150]. The components of the energy momentum tensor can be identified with the energy-

density, ρ , pressure, P, velocity, v_i , and anisotropic stress, Σ_{ij} of a perfect fluid:

$$T^{0}_{\ 0} = -\rho ,$$

$$T^{0}_{\ i} = (\rho + P)v_{i} ,$$

$$T^{i}_{\ j} = P\delta^{i}_{\ j} + \Sigma^{i}_{\ j} .$$
(C1)

In full General Relativity this decomposition holds for linear perturbations, where $T = \overline{T} + \delta T$, and helps identify the physical meaning of the sources of the Einstein equation. Perturbations are defined such that \overline{T} has the symmetries of the FRW metric. Perturbations in fluid components are defined as $\rho = \overline{\rho} + \delta \rho =$ $\overline{\rho}(1+\delta)$, $P = \overline{P} + \delta P$. Homogeneity and isotropy at the background level imply that v_i and Σ_{ij} are (at least) first order. The related variables θ and σ and are defined by

$$\theta = ik^j v_j \,, \tag{C2}$$

$$(\bar{\rho} + \bar{P})\sigma = -\left(\hat{k}^{j}\hat{k}_{i} - \frac{1}{3}\delta^{j}_{\ i}\right)\Sigma^{i}_{\ j}, \qquad (C3)$$

where \hat{k} is a unit vector in Fourier space.

The continuity equation for the energy density is

$$\dot{\bar{\rho}} = -3H(1+w)\bar{\rho}\,,\tag{C4}$$

where the equation of state is $w = \bar{P}/\bar{\rho}$. Matter and radiation have constant equations of state, $w_m = 0$, $w_r = 1/3$. The cosmological constant has equation of state $w_{\Lambda} = -1$. In the general, the equation of state can evolve in time. It's equation of motion is

$$\dot{w} = -3H(1+w)(w-c_{\rm ad}^2),$$
 (C5)

where the adiabatic (background) sound speed is

$$c_{\rm ad}^2 = \frac{\bar{P}}{\dot{\rho}} = c_s^2 - \frac{w}{\delta}\Gamma.$$
 (C6)

The sound speed in fluctuations is

$$c_s^2 = \frac{\delta P}{\delta \rho} \,, \tag{C7}$$

and Γ is the non-adiabatic pressure perturbation.

It is important to note that definitions of "sound speed" are not universal, and that the sound speed itself is not gauge invariant. I adopt the definitions above, and apply them in whatever gauge we happen to be working in (synchronous or Newtonian). This is in keeping with the treatment of Ref. [149], and is convenient and intuitive for standard cosmological perturbation theory as applied to the post-inflationary universe.

Some authors define the sound speed as the co-efficient in the equation of motion of the gauge invariant "Mukhanov-Sasaki" variable, ν . This is common in inflationary theory, and among relativists. For a scalar field, let's denote this particular sound speed c_{ϕ}^2 . One can prove that $c_{\phi}^2 = 1$: i.e. it is the sound speed in the gauge in which $\delta \phi = 0$ (flat scalar field slicing). The nontrivial growth and scalar field Jeans scale in this formulation can be understood from the behaviour of the background (anti-)friction terms induced by gauge transformations from, e.g., the Newtonian gauge to the $\delta \phi = 0$ gauge [163]. This is consistent with the time-averaged effective sound-speed we employed in Section 4.4.3, and the driven nature of Eqs. (74) and Eqs. (75) in the oscillating regime [159].

E. Bayes Theorem and Priors

All cosmologists worth their salt are Bayesians. This happy state of affairs is forced upon us by the unavoidably one-shot nature of observing the cosmos. An introduction to Bayesian methods in cosmology can be found in Ref. [391], with a more advanced specific treatment in Ref. [392].

We are interested in the probability of our theory, specifed by a vector of parameters $\vec{\theta}$, given the data D: $P(\vec{\theta}|D)$. What we have access to is the *like-lihood*, \mathcal{L} , i.e. the probability of the data given the theory: $P(D|\vec{\theta}) = \mathcal{L}(D, \vec{\theta})$. Bayes theorem relates these for us:

$$P(\vec{\theta}|D) = \frac{P(D|\vec{\theta})P(\vec{\theta})}{P(D)}; \quad \text{posterior} = \frac{\text{likelihood} \times \text{prior}}{\text{evidence}}. \tag{D1}$$

The probability of the theory, $P(\vec{\theta})$, is the all-important *prior*. In an MCMC setting, the prior can be thought of as the distribution from which we draw sample theory curves to compare to the data (although it can also be imposed later on top of uniform sampling). The probability of the data, P(D), can be computed as a normalization. It can often be ignored, since we are interested in ratios of probabilities, although it is important for model comparison and Bayesian evidence.

The likelihood reflects our uncertainty on the data. A very simple assumption is to weight data points individually, and assume Gaussian errors, so that a model has a likelihood as a product of Gaussians given by the distance of the theory curve from each data point. In many real-world examples, the likelihood is much more complicated. For example, the *Planck* likelihood is discussed in Ref. [214].

The prior reflects our degree of belief in a model, and is often where physics can be put in. See Ref. [393] for an example in dark energy theory, and the formalism for treating information gain over the prior in a Bayesian context.

An "uniformative" prior is the Jeffreys prior, which for most practical purposes is flat in log space. It is a suitable prior for unknown energy scales, for example the axion mass and decay constant. The log-flat prior on axion mass is also physically motivated: in string theory the mass scales exponentially with some modulus, σ , of the compact space: $m_{a,i} \propto e^{-c\sigma_i}$, where *i* labels the axion species. The moduli are expected to have a uniform distribution in real space (since the scale is set by the compactification volume), leading to a log-flat axion mass distribution. String theory predicitions for the f_a distribution are in general not log-flat, since $f_{a,i} \propto M_{pl}/\sigma_i$ [5]. The distribution can be calculated from random matrix theory, which selects some preferred scale somewhat below the Planck scale (e.g. Refs. [66, 394]).

The axion initial misalignment angle, on the other hand, is a compact variable, and so the natural prior is a uniform prior. For the QCD axion, holding f_a fixed and using that $\Omega_a h^2 \propto \theta_{a,i}^2$ this gives the prior distribution for the relic density (e.g. Ref. [145]):

$$P(\Omega_a h^2) \propto rac{1}{\sqrt{\Omega_a h^2}} \, .$$

(D2)

This fixed prior from theory makes axions uniquely predictive in landscape and multiverse scenarios (e.g. Refs. [144, 395, 396]). Incorporating additional information such as the prior on f_a for the QCD axion, or on m_a for ALPs, has not yet been fully explored in the literature.

F. Degeneracies and Sampling with ULAs

On scales much larger than the Jeans scale, axion DM is degenerate with CDM. For very low mass axions with $m_a \sim H_0$, the axion equation of state is $w_a \approx -1$ even today, and axions are degenerate with the cosmological constant and DE. Our goal is to use precision cosmology to map out the range of axion masses in between, i.e. those masses constrained by cosmology because such axions are neither equivalent to CDM nor DE. This leads to a very challenging degeneracy structure for $\Omega_a h^2$ as a function of m_a , which is illustrated in Fig. F.38.

Standard cosmological parameter estimation is carried out using MCMC analysis (the industry standard used by *Planck* is COSMOMC [397]; see e.g. Ref. [398] for a description of the methodology). The chain is begun at some location close to the maximum likelihood, and then randomly (and ergodically) explores this likelihood, with the density of samples reflecting the value of the likelihood. With infinite computing time, the process is guaranteed to explore the entire likelihood. Allowing for a wide prior on m_a makes the convergence of this process very slow, and the chain can get "stuck" in particular regions (modes) of the likelihood. For example, we might get stuck in a high-likelihood region with large m_a , and $\Omega_a h^2 \approx \Omega_c h^2$. What we really want to know is the constraint on $\Omega_a h^2$ at intermediate masses, and what the range of "intermediate" really is for a given observable.

Working around this bottleneck requires using different tools to estimate the likelihood than a standard "out-of-the-box" MCMC. The method employed in



Figure F.38: Degeneracy of $\Omega_a h^2$ and $\Omega_c h^2$. Sample points for an MCMC chain are shown, binned by axion mass, m_a . High axion mass leads to a one-to-one degeneracy, with $\Omega_a h^2 + \Omega_c h^2 \approx 0.12$. Low mass axions behave as DE, allowing for large $\Omega_a h^2$ and fixing $\Omega_c h^2 = 0.12$. Intermediate masses are constrained to have $\Omega_a h^2 < 0.12$. Reproduced (with permission) from Ref. [131]. Copyright (2015) by The American Physical Society.

Ref. [131] used nested sampling with MULTINEST [225], an algorithm designed for multi-modal likelihoods, instead of MCMC. However, it still proved prohibitively expensive to have enough sample points to achieve accurate limits on $\Omega_a h^2$ across the full range of m_a in the two dimensional (m_a, Ω_a) plane. A twostep procedure was used to overcome this. Three separate mass ranges ran independently. A more coarse global chain was then ran, and the information from this was used to importance-sample the individual chains together on the (m_a, Ω_a) plane.⁵³

The procedure described above was able to deal with the degeneracies between CDM, DE and axions that occur for high and low ULA masses respectively. A separate issue that has yet to be addressed fully is the degeneracy between ULAs and neutrinos at intermediate ULA mass. Cosmology is approaching the required precision to detect the effects of $\sum m_{\nu} = 0.06$ eV, the minimum consistent with oscillation experiments. It is crucially important to address all possible degeneracies so that a future detection can be considered robust. Ref. [171] used a grid-based likelihood, where convergence is not an issue, but only constrained m_a and m_{ν} independently. Grids scale poorly for large numbers of parameters, and are unsuitable for precision analysis. Ref. [172] performed a preliminary investigation using a Fisher matrix formalism to perform forecasts. At the level of the study, degeneracies were not too severe: the difference in behaviour between axions and neutrinos during the radiation era breaks the degeneracy in the effect on structure formation. However, Ref. [172] looked at individual ULA masses independently, and did not study the degeneracies as a function of ULA mass. Including ULAs, CDM and neutrinos in a full parameter estimation pipeline will likely require further tricks like those described here to be employed when sampling the likelihood.

In general, when considering degeneracies, it is important to break the ef-

 $^{^{53}}$ A similar procedure using a 'hot' MCMC chain as the global sample could also have been used, but MULTINEST was found to be more efficient. Another alternative would be to use an ensemble sampler, such as EMCEE [399].

fects of axion DM up into two parts: effects on the background expansion, and effects on the perturbations. Axion cosmology coming purely from the misalignment production is a well defined model where all effects on the expansion rate, clustering and initial conditions come packaged together. As we saw in Section 4.4.5, and has been discussed extensively elsewhere in the literature, the axion transfer function is similar to the WDM and neutrino transfer functions. However, these thermal and non-thermal components behave quite differently in their effects on the background expansion, leading to, for example, very different CMB signatures for similar transfer functions. It also might naively appear that any effect on the transfer function can be mimicked by a change in the primordial power. However, the primordial power affects radiation and DM, and so its effects show up in the CMB as well as in the matter power spectrum. The DM transfer function will only show up at leading order in the matter power spectrum. Multiple measurements can thus break that possible degeneracy. Similarly, axion effects on the background expansion could be mimicked by some particular model for the DE equation of state or modified gravity (MG). However, the particular physical DE/MG model may have different clustering or early Universe behaviour from the corresponding axion model, allowing the two to be distinguished.

G. Sheth-Tormen Halo Mass Function

The HMF is given by

$$\frac{dn}{d\ln M} = -\frac{1}{2} \frac{\rho_m}{M} f(\nu) \frac{d\ln \sigma^2}{d\ln M}, \qquad (F1)$$

$$\nu \equiv \frac{\delta_{\rm crit}}{\sigma} \,. \tag{F2}$$

For $f(\nu)$ we use the Sheth-Tormen function [400]:

$$f(\nu) = A \sqrt{\frac{2}{\pi}} \sqrt{q} \nu (1 + (\sqrt{q}\nu)^{-2p}) \exp\left[-\frac{q\nu^2}{2}\right],$$
 (F3)

with parameters $\{A = 0.3222, p = 0.3, q = 0.707\}$. This is a semi-analytic result for the HMF derived in ellipsoidal collapse, which fits results from CDM N-body

simulations reasonably well. Other fits for $f(\nu)$ can be found by fitting directly to N-body simulations, but the Sheth-Tormen result will do for us.

The variance is defined by smoothing the power spectrum with some window function, W(k|R), of radius R and assigning a mass using the enclosed matter density:

$$\sigma^{2}(M,z) = \frac{1}{2\pi^{2}} \int_{0}^{\infty} \frac{dk}{k} \Delta^{2}(k,z) W^{2}(k|R(M)) , \qquad (F4)$$

where $\Delta^2(k, z) = k^3 P(k, z)$. A real-space spherical top-hat window function assigns mass unambiguously:

$$W(k|R) = \frac{3}{(kR)^3} (\sin kR - kR \cos kR),$$
 (F5)

$$M = \frac{4}{3}\pi\rho_m R^3.$$
(F6)

- S. Weinberg, The cosmological constant problem, Reviews of Modern Physics 61 (1989) 1-23. doi:10.1103/RevModPhys.61.1.
- F. Wilczek, Problem of Strong p and t Invariance in the Presence of Instantons, Phys. Rev. Lett.40 (1978) 279–282. doi:10.1103/PhysRevLett.40.279.
- [3] R. Peccei, H. R. Quinn, CP Conservation in the Presence of Instantons, Phys. Rev. Lett.38 (1977) 1440-1443. doi:10.1103/PhysRevLett.38. 1440.
- [4] F. Wilczek, Current Contents Citation Classics 16 (1991) 8-9.
- [5] P. Svrcek, E. Witten, Axions in string theory, JHEP6 (2006) 51. arXiv: hep-th/0605206, doi:10.1088/1126-6708/2006/06/051.
- [6] R. D. Peccei, The Strong CP Problem and Axions, in: M. Kuster, G. Raffelt, B. Beltrán (Eds.), Axions, Vol. 741 of Lecture Notes in Physics, Berlin Springer Verlag, 2008, p. 3. arXiv:hep-ph/0607268.
- J. E. Kim, G. Carosi, Axions and the strong CP problem, Reviews of Modern Physics 82 (2010) 557-601. arXiv:0807.3125, doi:10.1103/ RevModPhys.82.557.

- [8] P. Sikivie, Axion Cosmology, in: M. Kuster, G. Raffelt, B. Beltrán (Eds.), Axions, Vol. 741 of Lecture Notes in Physics, Berlin Springer Verlag, 2008, p. 19. arXiv:astro-ph/0610440.
- [9] J. E. Kim, Light Pseudoscalars, Particle Physics and Cosmology, Phys. Rept. 150 (1987) 1–177. doi:10.1016/0370-1573(87)90017-2.
- [10] G. G. Raffelt, Astrophysical Axion Bounds, in: M. Kuster, G. Raffelt,
 B. Beltrán (Eds.), Axions, Vol. 741 of Lecture Notes in Physics, Berlin Springer Verlag, 2008, p. 51. arXiv:hep-ph/0611350.
- [11] E. Pajer, M. Peloso, A review of axion inflation in the era of Planck, Classical and Quantum Gravity 30 (21) (2013) 214002. arXiv:1305.3557, doi:10.1088/0264-9381/30/21/214002.
- [12] A. Ringwald, Exploring the role of axions and other WISPs in the dark universe, Physics of the Dark Universe 1 (2012) 116-135. arXiv:1210.
 5081, doi:10.1016/j.dark.2012.10.008.
- [13] J. Redondo, A. Ringwald, Light shining through walls, Contemporary Physics 52 (2011) 211-236. arXiv:1011.3741, doi:10.1080/00107514. 2011.563516.
- [14] A. Ringwald, Searching for axions and ALPs from string theory, ArXiv e-printsarXiv:1209.2299.
- [15] Y. V. Stadnik, V. V. Flambaum, New generation low-energy probes for ultralight axion and scalar dark matter, ArXiv e-printsarXiv:1506.08364.
- [16] G. Carosi, A. Friedland, M. Giannotti, M. J. Pivovaroff, J. Ruz, J. K. Vogel, Probing the axion-photon coupling: phenomenological and experimental perspectives. A snowmass white paper, ArXiv e-printsarXiv: 1309.7035.
- [17] A. Arvanitaki, S. Dimopoulos, S. Dubovsky, N. Kaloper, J. March-Russell, String axiverse, Phys. Rev. D81 (12) (2010) 123530. arXiv:0905.4720, doi:10.1103/PhysRevD.81.123530.

- [18] S. Coleman, Aspects of Symmetry, Cambridge University Press, 1988.
- [19] R. J. Crewther, P. di Vecchia, G. Veneziano, E. Witten, Chiral estimate of the electric dipole moment of the neutron in quantum chromodynamics, Phys. Lett. B 88 (1979) 123–127. doi:10.1016/0370-2693(79)90128-X.
- [20] C. A. Baker, D. D. Doyle, P. Geltenbort, K. Green, M. G. D. van der Grinten, P. G. Harris, P. Iaydjiev, S. N. Ivanov, D. J. R. May, J. M. Pendlebury, J. D. Richardson, D. Shiers, K. F. Smith, Improved Experimental Limit on the Electric Dipole Moment of the Neutron, Phys. Rev. Lett.97 (13) (2006) 131801. arXiv:hep-ex/0602020, doi:10.1103/PhysRevLett.97. 131801.
- [21] K. A. Olive, Particle Data Group, Review of Particle Physics, Chinese Physics C 38 (9) (2014) 090001. arXiv:1412.1408, doi:10.1088/1674-1137/38/9/090001.
- M. Pospelov, A. Ritz, Electric dipole moments as probes of new physics, Annals of Physics 318 (2005) 119-169. arXiv:hep-ph/0504231, doi: 10.1016/j.aop.2005.04.002.
- [23] C. Vafa, E. Witten, Parity Conservation in QCD, Phys. Rev. Lett.53 (1984) 535. doi:10.1103/PhysRevLett.53.535.
- [24] S. Weinberg, A New Light Boson?, Phys. Rev. Lett.40 (1978) 223-226. doi:10.1103/PhysRevLett.40.223.
- [25] J. Preskill, M. B. Wise, F. Wilczek, Cosmology of the invisible axion, Phys. Lett. B 120 (1983) 127–132. doi:10.1016/0370-2693(83)90637-8.
- [26] L. F. Abbott, P. Sikivie, A cosmological bound on the invisible axion, Phys. Lett. B 120 (1983) 133–136. doi:10.1016/0370-2693(83)90638-X.
- [27] M. Dine, W. Fischler, The not-so-harmless axion, Phys. Lett. B 120 (1983)
 137–141. doi:10.1016/0370-2693(83)90639-1.

- [28] J. E. Kim, Weak-interaction singlet and strong CP invariance, Phys. Rev. Lett.43 (1979) 103–107. doi:10.1103/PhysRevLett.43.103.
- [29] M. A. Shifman, A. I. Vainshtein, V. I. Zakharov, Can confinement ensure natural CP invariance of strong interactions?, Nuclear Physics B 166 (1980) 493–506. doi:10.1016/0550-3213(80)90209-6.
- [30] M. Dine, W. Fischler, M. Srednicki, A simple solution to the strong CP problem with a harmless axion, Phys. Lett. B 104 (1981) 199–202. doi: 10.1016/0370-2693(81)90590-6.
- [31] A. Zhitnitsky, On Possible Suppression of the Axion Hadron Interactions. (In Russian), Sov.J. Nucl. Phys. 31 (1980) 260.
- [32] G. Aad, et al., Observation of a new particle in the search for the Standard Model Higgs boson with the ATLAS detector at the LHC, Phys. Lett. B 716 (2012) 1-29. arXiv:1207.7214, doi:10.1016/j.physletb.2012.08.
 020.
- [33] S. Chatrchyan, et al., Observation of a new boson at a mass of 125 GeV with the CMS experiment at the LHC, Phys. Lett. B 716 (2012) 30-61. arXiv:1207.7235, doi:10.1016/j.physletb.2012.08.021.
- [34] G. 't Hooft, Computation of the Quantum Effects Due to a Four-Dimensional Pseudoparticle, Phys.Rev. D14 (1976) 3432-3450. doi: 10.1103/PhysRevD.18.2199.3, 10.1103/PhysRevD.14.3432.
- [35] K. Mimasu, V. Sanz, ALPs at colliders, Journal of High Energy Physics
 6 (2015) 173. arXiv:1409.4792, doi:10.1007/JHEP06(2015)173.
- [36] J. E. Kim, Constraints on very light axions from cavity experiments, Phys. Rev. D58 (5) (1998) 055006. arXiv:hep-ph/9802220, doi:10. 1103/PhysRevD.58.055006.
- [37] A. Djouadi, L. Maiani, A. Polosa, J. Quevillon, V. Riquer, Fully covering the MSSM Higgs sector at the LHC, JHEP 06 (2015) 168. arXiv:1502. 05653, doi:10.1007/JHEP06(2015)168.

- [38] G. Aad, et al., Constraints on new phenomena via Higgs boson couplings and invisible decays with the ATLAS detectorarXiv:1509.00672.
- [39] M. Srednicki, Axion Couplings to Matter. 1. CP Conserving Parts, Nucl. Phys. B260 (1985) 689. doi:10.1016/0550-3213(85)90054-9.
- [40] C. G. Callan, Jr., R. Dashen, D. J. Gross, Toward a theory of the strong interactions, Phys. Rev. D17 (1978) 2717-2763. doi:10.1103/PhysRevD. 17.2717.
- [41] D. J. Gross, R. D. Pisarski, L. G. Yaffe, QCD and instantons at finite temperature, Reviews of Modern Physics 53 (1981) 43-80. doi:10.1103/ RevModPhys.53.43.
- [42] K. Olive, et al., Review of Particle Physics, Chin. Phys. C38 (2014)
 090001. doi:10.1088/1674-1137/38/9/090001.
- [43] E. W. Kolb, M. S. Turner, The early universe., Addison-Wesley, 1990.
- [44] K. Blum, R. T. D'Agnolo, M. Lisanti, B. R. Safdi, Constraining axion dark matter with Big Bang Nucleosynthesis, Phys. Lett. B 737 (2014) 30-33. arXiv:1401.6460, doi:10.1016/j.physletb.2014.07.059.
- [45] A. G. Dias, A. C. B. Machado, C. C. Nishi, A. Ringwald, P. Vaudrevange, The quest for an intermediate-scale accidental axion and further ALPs, Journal of High Energy Physics 6 (2014) 37. arXiv:1403.5760, doi: 10.1007/JHEP06(2014)037.
- [46] J. E. Kim, D. J. E. Marsh, An ultralight pseudoscalar boson, Phys. Rev. D93. arXiv:1510.01701.
- [47] P. W. Graham, S. Rajendran, New observables for direct detection of axion dark matter, Phys. Rev. D88 (3) (2013) 035023. arXiv:1306.6088, doi:10.1103/PhysRevD.88.035023.
- [48] M. B. Green, J. H. Schwarz, E. Witten, Superstring theory. Volume 1 -Introduction, Cambridge University Press, 1987.

- [49] M. B. Green, J. H. Schwarz, E. Witten, Superstring theory. Volume 2 -Loop amplitudes, anomalies and phenomenology, Cambridge University Press, 1987.
- [50] K. Becker, M. Becker, J. H. Schwarz, String Theory and M-Theory, Cambridge University Press, 2007. doi:10.2277/0521860695.
- [51] P. Candelas, G. T. Horowitz, A. Strominger, E. Witten, Vacuum configurations for superstrings, Nuclear Physics B 258 (1985) 46–74. doi: 10.1016/0550-3213(85)90602-9.
- [52] E. Witten, Some properties of O(32) superstrings, Physics Letters B 149 (1984) 351–356. doi:10.1016/0370-2693(84)90422-2.
- Y.-H. He, Calabi-Yau Geometries: Algorithms, Databases and Physics, International Journal of Modern Physics A 28 (2013) 30032. arXiv:1308.
 0186, doi:10.1142/S0217751X13300329.
- [54] M. Kreuzer, H. Skarke, Reflexive polyhedra, weights and toric Calabi-Yau fibrations, ArXiv Mathematics e-printsarXiv:math/0001106.
- [55] S. M. Carroll, Spacetime and geometry. An introduction to general relativity, Addison Wesley, 2004.
- [56] J. Maldacena, The Large-N Limit of Superconformal Field Theories and Supergravity, International Journal of Theoretical Physics 38 (1999) 1113– 1133. arXiv:hep-th/9711200, doi:10.1023/A:1026654312961.
- [57] E. Witten, Anti-de Sitter space and holography, Advances in Theoretical and Mathematical Physics 2 (1998) 253-291. arXiv:hep-th/9802150.
- [58] P. Svrcek, Cosmological Constant and Axions in String Theory, ArXiv High Energy Physics - Theory e-printsarXiv:hep-th/0607086.
- [59] N. Arkani-Hamed, L. Motl, A. Nicolis, C. Vafa, The string landscape, black holes and gravity as the weakest force, JHEP6 (2007) 60. arXiv: hep-th/0601001, doi:10.1088/1126-6708/2007/06/060.

- [60] J. P. Conlon, S. Krippendorf, Axion decay constants away from the lamppost, ArXiv e-printsarXiv:1601.00647.
- [61] J. E. Kim, H. P. Nilles, M. Peloso, Completing natural inflation, JCAP1 (2005) 5. arXiv:hep-ph/0409138, doi:10.1088/1475-7516/2005/01/005.
- [62] T. C. Bachlechner, M. Dias, J. Frazer, L. McAllister, Chaotic inflation with kinetic alignment of axion fields, Phys. Rev. D91 (2) (2015) 023520. arXiv:1404.7496, doi:10.1103/PhysRevD.91.023520.
- S. B. Giddings, A. Strominger, Axion-induced topology change in quantum gravity and string theory, Nuclear Physics B 306 (1988) 890-907. doi:10.1016/0550-3213(88)90446-4.
- [64] D. J. E. Marsh, Axiverse extended: Vacuum destabilization, early dark energy, and cosmological collapse, Phys. Rev. D83 (12) (2011) 123526. arXiv:1102.4851, doi:10.1103/PhysRevD.83.123526.
- [65] D. J. E. Marsh, E. R. M. Tarrant, E. J. Copeland, P. G. Ferreira, Cosmology of axions and moduli: A dynamical systems approach, Phys. Rev. D86 (2) (2012) 023508. arXiv:1204.3632, doi:10.1103/PhysRevD.86. 023508.
- [66] B. S. Acharya, K. Bobkov, P. Kumar, An M Theory Solution to the Strong CP Problem and Constraints on the Axiverse, JHEP 1011 (2010) 105. arXiv:1004.5138, doi:10.1007/JHEP11(2010)105.
- [67] M. Cicoli, M. D. Goodsell, A. Ringwald, The type IIB string axiverse and its low-energy phenomenology, JHEP10 (2012) 146. arXiv:1206.0819, doi:10.1007/JHEP10(2012)146.
- [68] V. Balasubramanian, P. Berglund, J. P. Conlon, F. Quevedo, Systematics of Moduli Stabilisation in Calabi-Yau Flux Compactifications, Journal of High Energy Physics 3 (2005) 7. arXiv:hep-th/0502058, doi:10.1088/ 1126-6708/2005/03/007.

- [69] J. P. Conlon, F. Quevedo, K. Suruliz, Large-volume flux compactifications: moduli spectrum and D3/D7 soft supersymmetry breaking, Journal of High Energy Physics 8 (2005) 7. arXiv:hep-th/0505076, doi:10.1088/ 1126-6708/2005/08/007.
- M. Cicoli, K. Dutta, A. Maharana, N-flation with hierarchically light axions in string compactifications, JCAP8 (2014) 12. arXiv:1401.2579, doi:10.1088/1475-7516/2014/08/012.
- [71] S. Dimopoulos, S. Kachru, J. McGreevy, J. G. Wacker, N-flation, JCAP8 (2008) 3. arXiv:hep-th/0507205, doi:10.1088/1475-7516/2008/08/003.
- [72] J. A. Grifols, E. Massó, R. Toldrà, Gamma Rays from SN 1987A due to Pseudoscalar Conversion, Phys. Rev. Lett.77 (1996) 2372-2375. arXiv: astro-ph/9606028, doi:10.1103/PhysRevLett.77.2372.
- [73] J. W. Brockway, E. D. Carlson, G. G. Raffelt, SN 1987A gamma-ray limits on the conversion of pseudoscalars, Phys. Lett. B 383 (1996) 439– 443. arXiv:astro-ph/9605197, doi:10.1016/0370-2693(96)00778-2.
- [74] A. Arvanitaki, M. Baryakhtar, X. Huang, Discovering the QCD axion with black holes and gravitational waves, Phys. Rev. D91 (8) (2015) 084011. arXiv:1411.2263, doi:10.1103/PhysRevD.91.084011.
- [75] A. R. Liddle, A. Mazumdar, F. E. Schunck, Assisted inflation, Phys. Rev. D58 (6) (1998) 061301. arXiv:astro-ph/9804177, doi:10.1103/ PhysRevD.58.061301.
- T. Banks, M. Dine, P. J. Fox, E. Gorbatov, On the possibility of large axion decay constants, JCAP6 (2003) 1. arXiv:hep-th/0303252, doi: 10.1088/1475-7516/2003/06/001.
- [77] K. Kooner, S. Parameswaran, I. Zavala, Warping the Weak Gravity Conjecture, ArXiv e-printsarXiv:1509.07049.

- [78] J. Brown, W. Cottrell, G. Shiu, P. Soler, On Axionic Field Ranges, Loopholes and the Weak Gravity Conjecture, ArXiv e-printsarXiv:1504. 00659.
- [79] B. Heidenreich, M. Reece, T. Rudelius, Weak Gravity Strongly Constrains Large-Field Axion Inflation, ArXiv e-printsarXiv:1506.03447.
- [80] J. Brown, W. Cottrell, G. Shiu, P. Soler, Fencing in the Swampland: Quantum Gravity Constraints on Large Field Inflation, ArXiv eprintsarXiv:1503.04783.
- [81] G. W. Gibbons, S. W. Hawking, Cosmological event horizons, thermodynamics, and particle creation, Phys. Rev. D15 (1977) 2738-2751. doi:10.1103/PhysRevD.15.2738.
- [82] A. D. Miller, R. Caldwell, M. J. Devlin, W. B. Dorwart, T. Herbig, M. R. Nolta, L. A. Page, J. Puchalla, E. Torbet, H. T. Tran, A Measurement of the Angular Power Spectrum of the Cosmic Microwave Background from L = 100 to 400, ApJLett524 (1999) L1–L4. arXiv:astro-ph/9906421, doi:10.1086/312293.
- [83] P. de Bernardis, P. A. R. Ade, J. J. Bock, J. R. Bond, J. Borrill, A. Boscaleri, K. Coble, B. P. Crill, G. De Gasperis, P. C. Farese, P. G. Ferreira, K. Ganga, M. Giacometti, E. Hivon, V. V. Hristov, A. Iacoangeli, A. H. Jaffe, A. E. Lange, L. Martinis, S. Masi, P. V. Mason, P. D. Mauskopf, A. Melchiorri, L. Miglio, T. Montroy, C. B. Netterfield, E. Pascale, F. Piacentini, D. Pogosyan, S. Prunet, S. Rao, G. Romeo, J. E. Ruhl, F. Scaramuzzi, D. Sforna, N. Vittorio, A flat Universe from high-resolution maps of the cosmic microwave background radiation, Nature404 (2000) 955–959. arXiv:astro-ph/0004404.
- [84] S. Hanany, P. Ade, A. Balbi, J. Bock, J. Borrill, A. Boscaleri, P. de Bernardis, P. G. Ferreira, V. V. Hristov, A. H. Jaffe, A. E. Lange, A. T. Lee, P. D. Mauskopf, C. B. Netterfield, S. Oh, E. Pascale, B. Rabii, P. L. Richards, G. F. Smoot, R. Stompor, C. D. Winant, J. H. P.

Wu, MAXIMA-1: A Measurement of the Cosmic Microwave Background Anisotropy on Angular Scales of 10' - 5deg, ApJLett545 (2000) L5-L9. arXiv:astro-ph/0005123, doi:10.1086/317322.

- [85] G. F. Smoot, C. L. Bennett, A. Kogut, E. L. Wright, J. Aymon, N. W. Boggess, E. S. Cheng, G. de Amici, S. Gulkis, M. G. Hauser, G. Hin-shaw, P. D. Jackson, M. Janssen, E. Kaita, T. Kelsall, P. Keegstra, C. Lineweaver, K. Loewenstein, P. Lubin, J. Mather, S. S. Meyer, S. H. Moseley, T. Murdock, L. Rokke, R. F. Silverberg, L. Tenorio, R. Weiss, D. T. Wilkinson, Structure in the COBE differential microwave radiometer first-year maps, ApJLett396 (1992) L1–L5. doi:10.1086/186504.
- [86] BICEP2/Keck and Planck Collaborations, P. A. R. Ade, N. Aghanim, Z. Ahmed, R. W. Aikin, K. D. Alexander, M. Arnaud, J. Aumont, C. Baccigalupi, A. J. Banday, et al., Joint Analysis of BICEP2/Keck Array and Planck Data, Phys. Rev. Lett.114 (10) (2015) 101301. arXiv:1502.00612, doi:10.1103/PhysRevLett.114.101301.
- [87] D. H. Lyth, What Would We Learn by Detecting a Gravitational Wave Signal in the Cosmic Microwave Background Anisotropy?, Physical Review Letters 78 (1997) 1861-1863. arXiv:hep-ph/9606387, doi:10. 1103/PhysRevLett.78.1861.
- [88] S. J. Asztalos, G. Carosi, C. Hagmann, D. Kinion, K. van Bibber, M. Hotz, L. J. Rosenberg, G. Rybka, J. Hoskins, J. Hwang, P. Sikivie, D. B. Tanner, R. Bradley, J. Clarke, ADMX Collaboration, SQUID-Based Microwave Cavity Search for Dark-Matter Axions, Phys. Rev. Lett.104 (4) (2010) 041301. arXiv:0910.5914, doi:10.1103/PhysRevLett.104.041301.
- [89] C. J. Hogan, M. J. Rees, Axion miniclusters, Phys. Lett. B 205 (1988) 228–230. doi:10.1016/0370-2693(88)91655-3.
- [90] P. Sikivie, Of Axions, Domain Walls and the Early Universe, Phys. Rev. Lett.48 (1982) 1156-1159. doi:10.1103/PhysRevLett.48.1156.

- [91] S. M. Barr, J. E. Kim, New Confining Force Solution of the QCD Axion Domain-Wall Problem, Phys. Rev. Lett.113 (24) (2014) 241301. arXiv: 1407.4311, doi:10.1103/PhysRevLett.113.241301.
- [92] F. W. Stecker, Q. Shafi, Axions and the evolution of structure in the universe, Physical Review Letters 50 (1983) 928-931. doi:10.1103/ PhysRevLett.50.928.
- [93] A. H. Guth, Inflationary universe: A possible solution to the horizon and flatness problems, Phys. Rev. D23 (1981) 347-356. doi:10.1103/ PhysRevD.23.347.
- [94] A. D. Linde, A new inflationary universe scenario: A possible solution of the horizon, flatness, homogeneity, isotropy and primordial monopole problems, Phys. Lett. B 108 (1982) 389–393. doi:10.1016/ 0370-2693(82)91219-9.
- [95] A. Albrecht, P. J. Steinhardt, Cosmology for grand unified theories with radiatively induced symmetry breaking, Phys. Rev. Lett.48 (1982) 1220– 1223. doi:10.1103/PhysRevLett.48.1220.
- [96] Planck Collaboration, P. A. R. Ade, N. Aghanim, M. Arnaud, F. Arroja, M. Ashdown, J. Aumont, C. Baccigalupi, M. Ballardini, A. J. Banday, et al., Planck 2015 results. XX. Constraints on inflation, ArXiv eprintsarXiv:1502.02114.
- [97] D. H. Lyth, Axions and inflation: Vacuum fluctuations, Phys. Rev. D45 (1992) 3394–3404. doi:10.1103/PhysRevD.45.3394.
- [98] G. Mangano, G. Miele, S. Pastor, M. Peloso, A precision calculation of the effective number of cosmological neutrinos, Phys. Lett. B 534 (2002) 8–16. arXiv:astro-ph/0111408, doi:10.1016/S0370-2693(02)01622-2.
- [99] T. Moroi, T. Takahashi, Effects of cosmological moduli fields on cosmic microwave background, Physics Letters B 522 (2001) 215-221. arXiv: hep-ph/0110096, doi:10.1016/S0370-2693(01)01295-3.

- [100] K. Enqvist, M. S. Sloth, Adiabatic CMB perturbations in pre-Big-Bang string cosmology, Nuclear Physics B 626 (2002) 395-409. arXiv:hep-ph/ 0109214, doi:10.1016/S0550-3213(02)00043-3.
- [101] D. H. Lyth, D. Wands, Generating the curvature perturbation without an inflaton, Physics Letters B 524 (2002) 5-14. arXiv:hep-ph/0110002, doi:10.1016/S0370-2693(01)01366-1.
- [102] L. Iliesiu, D. J. E. Marsh, K. Moodley, S. Watson, Constraining supersymmetry with heavy scalars: Using the CMB, Phys. Rev. D89 (10) (2014) 103513. arXiv:1312.3636, doi:10.1103/PhysRevD.89.103513.
- [103] Z. Hou, R. Keisler, L. Knox, M. Millea, C. Reichardt, How massless neutrinos affect the cosmic microwave background damping tail, Phys. Rev. D87 (8) (2013) 083008. arXiv:1104.2333, doi:10.1103/PhysRevD.87.083008.
- [104] B. Follin, L. Knox, M. Millea, Z. Pan, A First Detection of the Acoustic Oscillation Phase Shift Expected from the Cosmic Neutrino Background, ArXiv e-printsarXiv:1503.07863.
- [105] Planck Collaboration, Planck 2015 results. XIII. Cosmological parametersarXiv:1502.01589.
- [106] T. Higaki, K. Nakayama, F. Takahashi, Moduli-induced axion problem, Journal of High Energy Physics 7 (2013) 5. arXiv:1304.7987, doi:10. 1007/JHEP07(2013)005.
- [107] M. Cicoli, J. P. Conlon, F. Quevedo, Dark radiation in LARGE volume models, Phys. Rev. D87 (4) (2013) 043520. arXiv:1208.3562, doi:10.
 1103/PhysRevD.87.043520.
- [108] J. P. Conlon, M. C. D. Marsh, The cosmophenomenology of axionic dark radiation, JHEP10 (2013) 214. arXiv:1304.1804, doi:10.1007/JHEP10(2013)214.

- [109] G. D. Coughlan, W. Fischler, E. W. Kolb, S. Raby, G. G. Ross, Cosmological problems for the polonyi potential, Phys. Lett. B 131 (1983) 59–64.
 doi:10.1016/0370-2693(83)91091-2.
- [110] B. S. Acharya, K. Bobkov, G. Kane, J. Shao, S. Watson, P. Kumar, Nonthermal dark matter and the moduli problem in string frameworks, JHEP6 (2008) 64. arXiv:0804.0863, doi:10.1088/1126-6708/2008/06/064.
- [111] T. Hiramatsu, M. Kawasaki, K. Saikawa, T. Sekiguchi, Production of dark matter axions from collapse of string-wall systems, Phys. Rev. D85 (10) (2012) 105020. arXiv:1202.5851, doi:10.1103/PhysRevD.85.105020.
- T. W. B. Kibble, Topology of cosmic domains and strings, Journal of Physics A Mathematical General 9 (1976) 1387–1398. doi:10.1088/ 0305-4470/9/8/029.
- [113] O. Wantz, E. Shellard, Axion Cosmology Revisited, Phys. Rev. D82 (2010) 123508. arXiv:0910.1066, doi:10.1103/PhysRevD.82.123508.
- [114] R. L. Davis, Goldstone bosons in string models of galaxy formation, Phys. Rev. D32 (1985) 3172-3177. doi:10.1103/PhysRevD.32.3172.
- [115] D. Harari, P. Sikivie, On the evolution of global strings in the early universe, Phys. Lett. B 195 (1987) 361-365. doi:10.1016/0370-2693(87) 90032-3.
- [116] R. A. Battye, E. P. S. Shellard, Axion string constraints, Physical Review Letters 73 (1994) 2954-2957. arXiv:astro-ph/9403018, doi:10.1103/ PhysRevLett.73.2954.
- [117] R. A. Battye, E. P. S. Shellard, Axion String Constraints [Phys. Rev. Lett. 73, 2954 (1994)], Phys. Rev. Lett.76 (1996) 2203-2204. doi:10. 1103/PhysRevLett.76.2203.
- [118] N. Turok, Sub-degree Scale Microwave Anistropies from Cosmic Defects, ApJLett473 (1996) L5. arXiv:astro-ph/9606087, doi:10.1086/310384.

- [119] N. Kaiser, A. Stebbins, Microwave anisotropy due to cosmic strings, Nature310 (1984) 391–393. doi:10.1038/310391a0.
- [120] A. Salvio, A. Strumia, W. Xue, Thermal axion production, JCAP1 (2014)
 011. arXiv:1310.6982, doi:10.1088/1475-7516/2014/01/011.
- [121] M. Archidiacono, S. Hannestad, A. Mirizzi, G. Raffelt, Y. Y. Y. Wong, Axion hot dark matter bounds after Planck, JCAP10 (2013) 020. arXiv: 1307.0615, doi:10.1088/1475-7516/2013/10/020.
- [122] E. Di Valentino, S. Gariazzo, E. Giusarma, O. Mena, Robustness of cosmological axion mass limits, Phys. Rev. D91 (12) (2015) 123505. arXiv:1503.00911, doi:10.1103/PhysRevD.91.123505.
- [123] E. Di Valentino, E. Giusarma, M. Lattanzi, O. Mena, A. Melchiorri, J. Silk, Cosmological Axion and neutrino mass constraints from Planck 2015 temperature and polarization data, ArXiv e-printsarXiv:1507. 08665.
- [124] S. Hannestad, A. Mirizzi, G. Raffelt, A new cosmological mass limit on thermal relic axions, JCAP7 (2005) 002. arXiv:hep-ph/0504059, doi: 10.1088/1475-7516/2005/07/002.
- [125] S. Hannestad, A. Mirizzi, G. G. Raffelt, Y. Y. Y. Wong, Cosmological constraints on neutrino plus axion hot dark matter, JCAP8 (2007) 015. arXiv:0706.4198, doi:10.1088/1475-7516/2007/08/015.
- [126] S. Hannestad, A. Mirizzi, G. G. Raffelt, Y. Y. Y. Wong, Cosmological constraints on neutrino plus axion hot dark matter: update after WMAP-5, JCAP4 (2008) 019. arXiv:0803.1585, doi:10.1088/1475-7516/2008/04/019.
- [127] S. Hannestad, A. Mirizzi, G. G. Raffelt, Y. Y. Y. Wong, Neutrino and axion hot dark matter bounds after WMAP-7, JCAP8 (2010) 001. arXiv: 1004.0695, doi:10.1088/1475-7516/2010/08/001.

- M. Archidiacono, T. Basse, J. Hamann, S. Hannestad, G. Raffelt, Y. Y. Y.
 Wong, Future cosmological sensitivity for hot dark matter axions, JCAP5 (2015) 050. arXiv:1502.03325, doi:10.1088/1475-7516/2015/05/050.
- [129] D. Grin, T. L. Smith, M. Kamionkowski, Axion constraints in nonstandard thermal histories, Phys. Rev. D77 (8) (2008) 085020. arXiv:0711.1352, doi:10.1103/PhysRevD.77.085020.
- [130] B. Li, T. Rindler-Daller, P. R. Shapiro, Cosmological constraints on Bose-Einstein-condensed scalar field dark matter, Phys. Rev. D89 (8) (2014) 083536. arXiv:1310.6061, doi:10.1103/PhysRevD.89.083536.
- [131] R. Hlozek, D. Grin, D. J. E. Marsh, P. G. Ferreira, A search for ultra-light axions using precision cosmological data, Phys. Rev. D91 (2015) 103512. arXiv:1410.2896.
- [132] D. J. E. Marsh, P. G. Ferreira, Ultralight scalar fields and the growth of structure in the Universe, Phys. Rev. D82 (10) (2010) 103528. arXiv: 1009.3501, doi:10.1103/PhysRevD.82.103528.
- M. S. Turner, Coherent scalar-field oscillations in an expanding universe, Phys. Rev. D28 (1983) 1243–1247. doi:10.1103/PhysRevD.28.1243.
- [134] P. Fox, A. Pierce, S. Thomas, Probing a QCD String Axion with Precision Cosmological Measurements, ArXiv High Energy Physics - Theory e-printsarXiv:hep-th/0409059.
- [135] K. J. Bae, J.-H. Huh, J. E. Kim, Updating the axion cold dark matter energy density, JCAP9 (2008) 5. arXiv:0806.0497, doi:10.1088/ 1475-7516/2008/09/005.
- [136] E. Berkowitz, M. I. Buchoff, E. Rinaldi, Lattice QCD input for axion cosmology, Phys. Rev. D92 (3) (2015) 034507. arXiv:1505.07455, doi: 10.1103/PhysRevD.92.034507.

- [137] E. Berkowitz, Lattice QCD and Axion Cosmology, ArXiv e-printsarXiv: 1509.02976.
- [138] R. Kitano, N. Yamada, Topology in QCD and the axion abundance, JHEP10 (2015) 136. arXiv:1506.00370, doi:10.1007/JHEP10(2015) 136.
- [139] S. Borsanyi, M. Dierigl, Z. Fodor, S. D. Katz, S. W. Mages, D. Nogradi, J. Redondo, A. Ringwald, K. K. Szabo, Axion cosmology, lattice QCD and the dilute instanton gas, Physics Letters B 752 (2016) 175–181. arXiv: 1508.06917, doi:10.1016/j.physletb.2015.11.020.
- [140] G. Grilli di Cortona, E. Hardy, J. Pardo Vega, G. Villadoro, The QCD axion, precisely, ArXiv e-printsarXiv:1511.02867.
- [141] L. Visinelli, P. Gondolo, Axion cold dark matter in view of BICEP2 results, Phys. Rev. Lett.113 (2014) 011802. arXiv:1403.4594, doi: 10.1103/PhysRevLett.113.011802.
- [142] L. Visinelli, P. Gondolo, Dark matter axions revisited, Phys. Rev. D80 (3)
 (2009) 035024. arXiv:0903.4377, doi:10.1103/PhysRevD.80.035024.
- [143] P. J. Steinhardt, M. S. Turner, Saving the invisible axion, Phys. Lett. B 129 (1983) 51-56. doi:10.1016/0370-2693(83)90727-X.
- [144] F. Wilczek, A Model of Anthropic Reasoning, Addressing the Dark to Ordinary Matter Coincidence, ArXiv High Energy Physics - Phenomenology e-printsarXiv:hep-ph/0408167.
- M. Tegmark, A. Aguirre, M. J. Rees, F. Wilczek, Dimensionless constants, cosmology, and other dark matters, Phys. Rev. D73 (2) (2006) 023505.
 arXiv:astro-ph/0511774, doi:10.1103/PhysRevD.73.023505.
- M. P. Hertzberg, M. Tegmark, F. Wilczek, Axion Cosmology and the Energy Scale of Inflation, Phys. Rev. D78 (2008) 083507. arXiv:0807.
 1726, doi:10.1103/PhysRevD.78.083507.

- [147] K. J. Mack, Axions, inflation and the anthropic principle, JCAP7 (2011)
 021. arXiv:0911.0421, doi:10.1088/1475-7516/2011/07/021.
- [148] K. J. Mack, P. J. Steinhardt, Cosmological problems with multiple axion-like fields, JCAP5 (2011) 001. arXiv:0911.0418, doi:10.1088/ 1475-7516/2011/05/001.
- [149] C.-P. Ma, E. Bertschinger, Cosmological perturbation theory in the synchronous and conformal Newtonian gauges, ApJ455 (1995) 7-25. arXiv: astro-ph/9506072, doi:10.1086/176550.
- [150] W. Hu, Covariant Linear Perturbation Formalism, ArXiv Astrophysics e-printsarXiv:astro-ph/0402060.
- [151] L. D. Landau, E. M. Lifshitz, The classical theory of fields, 3rd rev. English edition, pp350, Pergamon, Oxford, 1971.
- [152] H. Kodama, M. Sasaki, Cosmological Perturbation Theory, Progress of Theoretical Physics Supplement 78 (1984) 1. doi:10.1143/PTPS.78.1.
- [153] A. Lewis, A. Challinor, A. Lasenby, Efficient Computation of Cosmic Microwave Background Anisotropies in Closed Friedmann-Robertson-Walker Models, ApJ538 (2000) 473–476. arXiv:astro-ph/9911177, doi:10.1086/309179.
- [154] J. Lesgourgues, The Cosmic Linear Anisotropy Solving System (CLASS)I: Overview, ArXiv e-printsarXiv:1104.2932.
- [155] D. Blas, J. Lesgourgues, T. Tram, The Cosmic Linear Anisotropy Solving System (CLASS). Part II: Approximation schemes, JCAP7 (2011) 34.
 arXiv:1104.2933, doi:10.1088/1475-7516/2011/07/034.
- [156] L. A. Ureña-López, A. X. Gonzalez-Morales, Towards accurate cosmological predictions for rapidly oscillating scalar fields as dark matter, ArXiv e-printsarXiv:1511.08195.

- [157] M. Bucher, K. Moodley, N. Turok, The General primordial cosmic perturbation, Phys. Rev. D62 (2000) 083508. arXiv:astro-ph/9904231, doi:10.1103/PhysRevD.62.083508.
- [158] K. A. Malik, D. Wands, Cosmological perturbations, Phys. Rep.475 (2009)
 1-51. arXiv:0809.4944, doi:10.1016/j.physrep.2009.03.001.
- [159] W. Hu, Structure formation with generalized dark matter, ApJ506 (1998)
 485–494. arXiv:astro-ph/9801234, doi:10.1086/306274.
- [160] J.-C. Hwang, H. Noh, Axion as a cold dark matter candidate, Phys. Lett. B 680 (2009) 1–3. arXiv:0902.4738, doi:10.1016/j.physletb.2009. 08.031.
- [161] C.-G. Park, J.-c. Hwang, H. Noh, Axion as a cold dark matter candidate: Low-mass case, Phys. Rev. D86 (8) (2012) 083535. arXiv:1207.3124, doi:10.1103/PhysRevD.86.083535.
- [162] M. Khlopov, B. Malomed, I. Zeldovich, Gravitational instability of scalar fields and formation of primordial black holes, MNRAS215 (1985) 575– 589.
- [163] M. Alcubierre, A. de la Macorra, A. Diez-Tejedor, J. M. Torres, Cosmological scalar field perturbations can growarXiv:1501.06918.
- [164] J. R. Bond, G. Efstathiou, J. Silk, Massive neutrinos and the large-scale structure of the universe, Physical Review Letters 45 (1980) 1980–1984. doi:10.1103/PhysRevLett.45.1980.
- [165] R. K. Schaefer, Q. Shafi, F. W. Stecker, Large-scale structure formation and cosmic microwave anisotropy in a cold plus hot dark matter universe, ApJ347 (1989) 575–589. doi:10.1086/168150.
- [166] S. Dodelson, E. Gates, A. Stebbins, Cold + Hot Dark Matter and the Cosmic Microwave Background, ApJ467 (1996) 10. arXiv:astro-ph/ 9509147, doi:10.1086/177581.
- [167] J. R. Bond, A. S. Szalay, M. S. Turner, Formation of galaxies in a gravitino-dominated universe, Physical Review Letters 48 (1982) 1636– 1639. doi:10.1103/PhysRevLett.48.1636.
- S. Dodelson, L. M. Widrow, Sterile neutrinos as dark matter, Physical Review Letters 72 (1994) 17-20. arXiv:hep-ph/9303287, doi:10.1103/ PhysRevLett.72.17.
- [169] S. Colombi, S. Dodelson, L. M. Widrow, Large-Scale Structure Tests of Warm Dark Matter, ApJ458 (1996) 1. arXiv:astro-ph/9505029, doi: 10.1086/176788.
- [170] P. Bode, J. P. Ostriker, N. Turok, Halo formation in warm dark matter models, ApJ556 (2001) 93-107. arXiv:astro-ph/0010389, doi: 10.1086/321541.
- [171] L. Amendola, R. Barbieri, Dark matter from an ultra-light pseudo-Goldsone-boson, Phys. Lett. B642 (2006) 192-196. arXiv:hep-ph/ 0509257, doi:10.1016/j.physletb.2006.08.069.
- [172] D. J. Marsh, E. Macaulay, M. Trebitsch, P. G. Ferreira, Ultra-light Axions: Degeneracies with Massive Neutrinos and Forecasts for Future Cosmological Observations, Phys. Rev. D85 (2012) 103514. arXiv:1110.0502, doi:10.1103/PhysRevD.85.103514.
- [173] J. M. Bardeen, J. R. Bond, N. Kaiser, A. S. Szalay, The statistics of peaks of Gaussian random fields, ApJ304 (1986) 15–61. doi:10.1086/164143.
- [174] D. J. Eisenstein, W. Hu, Baryonic Features in the Matter Transfer Function, ApJ496 (1998) 605-614. arXiv:astro-ph/9709112, doi:10.1086/ 305424.
- [175] P. J. E. Peebles, Principles of Physical Cosmology, Princeton University Press, 1993.

- [176] W. Hu, R. Barkana, A. Gruzinov, Cold and fuzzy dark matter, Phys. Rev. Lett.85 (2000) 1158-1161. arXiv:astro-ph/0003365, doi:10.1103/ PhysRevLett.85.1158.
- [177] M. Viel, J. Lesgourgues, M. G. Haehnelt, S. Matarrese, A. Riotto, Constraining warm dark matter candidates including sterile neutrinos and light gravitinos with WMAP and the Lyman-α forest, Phys. Rev. D71 (6) (2005) 063534. arXiv:astro-ph/0501562, doi:10.1103/PhysRevD.71. 063534.
- [178] D. J. E. Marsh, J. Silk, A model for halo formation with axion mixed dark matter, MNRAS437 (2014) 2652-2663. arXiv:1307.1705, doi:10.1093/ mnras/stt2079.
- [179] A. M. Green, S. Hofmann, D. J. Schwarz, The power spectrum of SUSY-CDM on subgalactic scales, MNRAS353 (2004) L23-L27. arXiv: astro-ph/0309621, doi:10.1111/j.1365-2966.2004.08232.x.
- [180] F. Bernardeau, S. Colombi, E. Gaztañaga, R. Scoccimarro, Largescale structure of the Universe and cosmological perturbation theory, Phys. Rep.367 (2002) 1-248. arXiv:astro-ph/0112551, doi:10.1016/ S0370-1573(02)00135-7.
- [181] P.-H. Chavanis, BEC dark matter, Zeldovich approximation, and generalized Burgers equation, Phys. Rev. D84 (6) (2011) 063518. arXiv: 1103.3219, doi:10.1103/PhysRevD.84.063518.
- [182] D. J. E. Marsh, Nonlinear hydrodynamics of axion dark matter: Relative velocity effects and quantum forces, Phys. Rev. D91 (12) (2015) 123520.
 arXiv:1504.00308, doi:10.1103/PhysRevD.91.123520.
- [183] R. E. Wyatt, Quantum Dynamics with Trajectories, Springer, 2005.
- [184] P. Mocz, S. Succi, Numerical solution of the non-linear Schrodinger equation using smoothed-particle hydrodynamics, Phys. Rev. E91 (2015) 053304. arXiv:1503.03869.

- [185] L. M. Widrow, N. Kaiser, Using the schrödinger equation to stimulate collisionless matter, ApJ416 (2) (1993) L71–L74.
- [186] C. Uhlemann, M. Kopp, T. Haugg, Schrdinger method as N-body double and UV completion of dust, Phys. Rev. D90 (2) (2014) 023517. arXiv: 1403.5567, doi:10.1103/PhysRevD.90.023517.
- [187] P. Coles, K. Spencer, A wave-mechanical approach to cosmic structure formation, Mon.Not.Roy.Astron.Soc. 342 (2003) 176. arXiv:astro-ph/ 0212433, doi:10.1046/j.1365-8711.2003.06529.x.
- [188] H.-Y. Schive, T. Chiueh, T. Broadhurst, K.-W. Huang, Contrasting Galaxy Formation from Quantum Wave Dark Matter, ψDM, with ΛCDM, using Planck and Hubble Data, ApJ818 (2016) 89. arXiv:1508.04621, doi:10.3847/0004-637X/818/1/89.
- [189] A. Sarkar, R. Mondal, S. Das, S. K. Sethi, S. Bharadwaj, D. J. E. Marsh, The effects of the small-scale DM power on the cosmological neutral hydrogen (HI) distribution at high redshifts, JCAP4 (2016) 012. arXiv:1512.03325, doi:10.1088/1475-7516/2016/04/012.
- [190] M. R. Lovell, C. S. Frenk, V. R. Eke, A. Jenkins, L. Gao, T. Theuns, The properties of warm dark matter haloes, MNRAS439 (2014) 300-317.
 arXiv:1308.1399, doi:10.1093/mnras/stt2431.
- [191] J. Wang, S. D. M. White, Discreteness effects in simulations of hot/warm dark matter, MNRAS380 (2007) 93–103. arXiv:astro-ph/0702575, doi: 10.1111/j.1365-2966.2007.12053.x.
- [192] A. Schneider, R. E. Smith, D. Reed, Halo mass function and the free streaming scale, MNRAS433 (2013) 1573-1587. arXiv:1303.0839, doi: 10.1093/mnras/stt829.
- [193] V. Springel, The cosmological simulation code GADGET-2, MN-RAS364 (2005) 1105–1134. arXiv:astro-ph/0505010, doi:10.1111/j. 1365-2966.2005.09655.x.

- [194] F.-Y. Cyr-Racine, K. Sigurdson, J. Zavala, T. Bringmann, M. Vogelsberger, C. Pfrommer, ETHOS - An Effective Theory of Structure Formation: From dark particle physics to the matter distribution of the Universe, ArXiv e-printsarXiv:1512.05344.
- [195] S. Bird, M. Viel, M. G. Haehnelt, Massive neutrinos and the nonlinear matter power spectrum, MNRAS420 (2012) 2551-2561. arXiv: 1109.4416, doi:10.1111/j.1365-2966.2011.20222.x.
- [196] D. Inman, J. D. Emberson, U.-L. Pen, A. Farchi, H.-R. Yu, J. Harnois-Déraps, Precision reconstruction of the cold dark matter-neutrino relative velocity from N -body simulations, Phys. Rev. D92 (2) (2015) 023502. arXiv:1503.07480, doi:10.1103/PhysRevD.92.023502.
- [197] J. Adamek, D. Daverio, R. Durrer, M. Kunz, gevolution: a cosmological Nbody code based on General Relativity, ArXiv e-printsarXiv:1604.06065.
- [198] H. A. Winther, F. Schmidt, A. Barreira, C. Arnold, S. Bose, C. Llinares, M. Baldi, B. Falck, W. A. Hellwing, K. Koyama, B. Li, D. F. Mota, E. Puchwein, R. E. Smith, G.-B. Zhao, Modified gravity N-body code comparison project, MNRAS454 (2015) 4208-4234. arXiv:1506.06384, doi:10.1093/mnras/stv2253.
- [199] H.-Y. Schive, T. Chiueh, T. Broadhurst, Cosmic structure as the quantum interference of a coherent dark wave, Nature Physics 10 (2014) 496-499.
 arXiv:1406.6586, doi:10.1038/nphys2996.
- [200] P. Sikivie, Q. Yang, Bose-Einstein Condensation of Dark Matter Axions, Phys. Rev. Lett.103 (11) (2009) 111301. arXiv:0901.1106, doi:10.1103/ PhysRevLett.103.111301.
- [201] S. Davidson, M. Elmer, Bose-Einstein condensation of the classical axion field in cosmology?, JCAP12 (2013) 34. arXiv:1307.8024, doi:10.1088/ 1475-7516/2013/12/034.

- [202] S. Davidson, Axions: Bose Einstein condensate or classical field?, Astroparticle Physics 65 (2015) 101-107. arXiv:1405.1139, doi:10.1016/ j.astropartphys.2014.12.007.
- [203] A. H. Guth, M. P. Hertzberg, C. Prescod-Weinstein, Do dark matter axions form a condensate with long-range correlation?, Phys. Rev. D92 (10) (2015) 103513. arXiv:1412.5930, doi:10.1103/PhysRevD.92.103513.
- [204] Y. Nambu, M. Sasaki, Quantum treatment of cosmological axion perturbations, Phys. Rev. D42 (1990) 3918-3924. doi:10.1103/PhysRevD.42. 3918.
- [205] B. Ratra, Expressions for linearized perturbations in a massive-scalarfield-dominated cosmological model, Phys. Rev. D44 (1991) 352–364. doi: 10.1103/PhysRevD.44.352.
- [206] P.-H. Chavanis, Mass-radius relation of Newtonian self-gravitating Bose-Einstein condensates with short-range interactions. I. Analytical results, Phys. Rev. D84 (4) (2011) 043531. arXiv:1103.2050, doi:10.1103/ PhysRevD.84.043531.
- [207] C. Cohen-Tannoudji, B. Diu, F. Laloe, Quantum Mechanics, Volume 1, Wiley, 1986.
- [208] C. J. Pethik, H. Smith, Bose-Einstein condensation in dilute gases, Cambridge University Press, 2002.
- [209] N. Banik, P. Sikivie, Axions and the galactic angular momentum distribution, Phys. Rev. D88 (12) (2013) 123517. arXiv:1307.3547, doi: 10.1103/PhysRevD.88.123517.
- [210] T. Rindler-Daller, P. R. Shapiro, Angular momentum and vortex formation in Bose-Einstein-condensed cold dark matter haloes, MNRAS422 (2012) 135-161. arXiv:1106.1256, doi:10.1111/j.1365-2966.2012. 20588.x.

- [211] D. J. E. Marsh, D. Grin, R. Hlozek, P. G. Ferreira, Axiverse cosmology and the energy scale of inflation, Phys. Rev. D87 (2013) 121701+. doi: 10.1103/physrevd.87.121701.
- [212] D. J. E. Marsh, D. Grin, R. Hlozek, P. G. Ferreira, Tensor Detection Severely Constrains Axion Dark Matter, Phys. Rev. Lett.113 (2014) 011801. arXiv:1403.4216, doi:10.1103/PhysRevLett.113.011801.
- [213] Planck Collaboration, P. A. R. Ade, N. Aghanim, M. I. R. Alves, C. Armitage-Caplan, M. Arnaud, M. Ashdown, F. Atrio-Barandela, J. Aumont, H. Aussel, et al., Planck 2013 results. I. Overview of products and scientific results, A&A571 (2014) A1. arXiv:1303.5062, doi:10.1051/ 0004-6361/201321529.
- [214] Planck Collaboration, P. A. R. Ade, N. Aghanim, C. Armitage-Caplan, M. Arnaud, M. Ashdown, F. Atrio-Barandela, J. Aumont, C. Baccigalupi, A. J. Banday, et al., Planck 2013 results. XV. CMB power spectra and likelihood, A&A571 (2014) A15. arXiv:1303.5075, doi: 10.1051/0004-6361/201321573.
- [215] C. L. Bennett, D. Larson, J. L. Weiland, N. Jarosik, G. Hinshaw, N. Odegard, K. M. Smith, R. S. Hill, B. Gold, M. Halpern, E. Komatsu, M. R. Nolta, L. Page, D. N. Spergel, E. Wollack, J. Dunkley, A. Kogut, M. Limon, S. S. Meyer, G. S. Tucker, E. L. Wright, Nine-year Wilkinson Microwave Anisotropy Probe (WMAP) Observations: Final Maps and Results, ApJS208 (2013) 20. arXiv:1212.5225, doi:10.1088/0067-0049/208/2/20.
- [216] S. Das, T. Louis, M. R. Nolta, G. E. Addison, E. S. Battistelli, J. R. Bond,
 E. Calabrese, D. Crichton, M. J. Devlin, S. Dicker, J. Dunkley, R. Dünner,
 J. W. Fowler, M. Gralla, A. Hajian, M. Halpern, M. Hasselfield, M. Hilton,
 A. D. Hincks, R. Hlozek, K. M. Huffenberger, J. P. Hughes, K. D. Irwin,
 A. Kosowsky, R. H. Lupton, T. A. Marriage, D. Marsden, F. Menanteau,
 K. Moodley, M. D. Niemack, L. A. Page, B. Partridge, E. D. Reese, B. L.

Schmitt, N. Sehgal, B. D. Sherwin, J. L. Sievers, D. N. Spergel, S. T. Staggs, D. S. Swetz, E. R. Switzer, R. Thornton, H. Trac, E. Wollack, The Atacama Cosmology Telescope: temperature and gravitational lensing power spectrum measurements from three seasons of data, JCAP4 (2014) 14. arXiv:1301.1037, doi:10.1088/1475-7516/2014/04/014.

- [217] R. Keisler, C. L. Reichardt, K. A. Aird, B. A. Benson, L. E. Bleem, J. E. Carlstrom, C. L. Chang, H. M. Cho, T. M. Crawford, A. T. Crites, T. de Haan, M. A. Dobbs, J. Dudley, E. M. George, N. W. Halverson, G. P. Holder, W. L. Holzapfel, S. Hoover, Z. Hou, J. D. Hrubes, M. Joy, L. Knox, A. T. Lee, E. M. Leitch, M. Lueker, D. Luong-Van, J. J. McMahon, J. Mehl, S. S. Meyer, M. Millea, J. J. Mohr, T. E. Montroy, T. Natoli, S. Padin, T. Plagge, C. Pryke, J. E. Ruhl, K. K. Schaffer, L. Shaw, E. Shirokoff, H. G. Spieler, Z. Staniszewski, A. A. Stark, K. Story, A. van Engelen, K. Vanderlinde, J. D. Vieira, R. Williamson, O. Zahn, A Measurement of the Damping Tail of the Cosmic Microwave Background Power Spectrum with the South Pole Telescope, ApJ743 (2011) 28. arXiv:1105.3182, doi:10.1088/0004-637X/743/1/28.
- [218] B. Bassett, R. Hlozek, Baryon acoustic oscillations, ArXiv e-printsarXiv: 0910.5224.
- [219] D. Parkinson, S. Riemer-Sorensen, C. Blake, G. B. Poole, T. M. Davis, et al., The WiggleZ Dark Energy Survey: Final data release and cosmological results, Phys. Rev. D86 (2012) 103518. arXiv:1210.2130, doi:10.1103/PhysRevD.86.103518.
- [220] M. LoVerde, Halo bias in mixed dark matter cosmologies, Phys. Rev. D90 (8) (2014) 083530. arXiv:1405.4855, doi:10.1103/PhysRevD.90. 083530.
- [221] A. Cooray, R. Sheth, Halo models of large scale structure, Phys. Rep.372
 (2002) 1–129. arXiv:astro-ph/0206508, doi:10.1016/S0370-1573(02)
 00276-4.

- [222] R. Laureijs, J. Amiaux, S. Arduini, J. . Auguères, J. Brinchmann, R. Cole, M. Cropper, C. Dabin, L. Duvet, A. Ealet, et al., Euclid Definition Study ReportarXiv:1110.3193.
- [223] R. E. Smith, K. Markovic, Testing the Warm Dark Matter paradigm with large-scale structures, Phys. Rev. D84 (2011) 063507. arXiv:1103.2134, doi:10.1103/PhysRevD.84.063507.
- [224] L. Amendola, et al., Cosmology and fundamental physics with the Euclid satellite, Living Rev.Rel. 16 (2013) 6. arXiv:1206.1225.
- [225] F. Feroz, M. P. Hobson, M. Bridges, MULTINEST: an efficient and robust Bayesian inference tool for cosmology and particle physics, MN-RAS398 (2009) 1601–1614. arXiv:arXiv:0809.3437, doi:10.1111/j. 1365-2966.2009.14548.x.
- [226] E. Komatsu, J. Dunkley, M. R. Nolta, C. L. Bennett, B. Gold, G. Hinshaw, N. Jarosik, D. Larson, M. Limon, L. Page, D. N. Spergel, M. Halpern, R. S. Hill, A. Kogut, S. S. Meyer, G. S. Tucker, J. L. Weiland, E. Wollack, E. L. Wright, Five-Year Wilkinson Microwave Anisotropy Probe Observations: Cosmological Interpretation, ApJS180 (2009) 330–376. arXiv:0803.0547, doi:10.1088/0067-0049/180/2/330.
- [227] T. Bringmann, P. Scott, Y. Akrami, Improved constraints on the primordial power spectrum at small scales from ultracompact minihalos, Phys. Rev. D85 (12) (2012) 125027. arXiv:1110.2484, doi:10.1103/ PhysRevD.85.125027.
- [228] J. Chluba, Science with CMB spectral distortions, ArXiv e-printsarXiv: 1405.6938.
- [229] K. N. Abazajian, K. Arnold, J. Austermann, B. A. Benson, C. Bischoff,J. Bock, J. R. Bond, J. Borrill, I. Buder, D. L. Burke, E. Calabrese,J. E. Carlstrom, C. S. Carvalho, C. L. Chang, H. C. Chiang, S. Church,

A. Cooray, T. M. Crawford, B. P. Crill, K. S. Dawson, S. Das, M. J. Devlin, M. Dobbs, S. Dodelson, O. Doré, J. Dunkley, J. L. Feng, A. Fraisse, J. Gallicchio, S. B. Giddings, D. Green, N. W. Halverson, S. Hanany, D. Hanson, S. R. Hildebrandt, A. Hincks, R. Hlozek, G. Holder, W. L. Holzapfel, K. Honscheid, G. Horowitz, W. Hu, J. Hubmayr, K. Irwin, M. Jackson, W. C. Jones, R. Kallosh, M. Kamionkowski, B. Keating, R. Keisler, W. Kinney, L. Knox, E. Komatsu, J. Kovac, C.-L. Kuo, A. Kusaka, C. Lawrence, A. T. Lee, E. Leitch, A. Linde, E. Linder, P. Lubin, J. Maldacena, E. Martinec, J. McMahon, A. Miller, V. Mukhanov, L. Newburgh, M. D. Niemack, H. Nguyen, H. T. Nguyen, L. Page, C. Pryke, C. L. Reichardt, J. E. Ruhl, N. Sehgal, U. Seljak, L. Senatore, J. Sievers, E. Silverstein, A. Slosar, K. M. Smith, D. Spergel, S. T. Staggs, A. Stark, R. Stompor, A. G. Vieregg, G. Wang, S. Watson, E. J. Wollack, W. L. K. Wu, K. W. Yoon, O. Zahn, M. Zaldarriaga, Inflation Physics from the Cosmic Microwave Background and Large Scale Structure, ArXiv e-printsarXiv:1309.5381.

- [230] K. Sigurdson, A. Cooray, Cosmic 21 cm Delensing of Microwave Background Polarization and the Minimum Detectable Energy Scale of Inflation, Phys. Rev. Lett.95 (21) (2005) 211303. arXiv:astro-ph/0502549, doi:10.1103/PhysRevLett.95.211303.
- [231] L. Book, M. Kamionkowski, F. Schmidt, Lensing of 21-cm Fluctuations by Primordial Gravitational Waves, Phys. Rev. Lett.108 (21) (2012) 211301.
 arXiv:1112.0567, doi:10.1103/PhysRevLett.108.211301.
- [232] S. Kachru, R. Kallosh, A. Linde, S. P. Trivedi, de Sitter vacua in string theory, Phys. Rev. D68 (4) (2003) 046005. arXiv:hep-th/0301240, doi: 10.1103/PhysRevD.68.046005.
- [233] J. Martin, C. Ringeval, V. Vennin, Encyclopædia Inflationaris, Physics of the Dark Universe 5 (2014) 75–235. arXiv:1303.3787, doi:10.1016/j. dark.2014.01.003.

- [234] M. Fairbairn, R. Hogan, D. J. E. Marsh, Unifying inflation and dark matter with the Peccei-Quinn field: Observable axions and observable tensors, Phys. Rev. D91 (2) (2015) 023509. arXiv:1410.1752, doi:10. 1103/PhysRevD.91.023509.
- [235] D. Budker, P. W. Graham, M. Ledbetter, S. Rajendran, A. O. Sushkov, Proposal for a Cosmic Axion Spin Precession Experiment (CASPEr), Phys. Rev. X 4 (2) (2014) 021030. arXiv:1306.6089, doi:10.1103/ PhysRevX.4.021030.
- [236] A. S. Rahlin, P. A. R. Ade, M. Amiri, S. J. Benton, J. J. Bock, J. R. Bond, S. A. Bryan, H. C. Chiang, C. R. Contaldi, B. P. Crill, O. Doré, M. Farhang, J. P. Filippini, L. M. Fissel, A. A. Fraisse, A. E. Gambrel, N. N. Gandilo, S. Golwala, J. E. Gudmundsson, M. Halpern, M. F. Hasselfield, G. Hilton, W. A. Holmes, V. V. Hristov, K. D. Irwin, W. C. Jones, Z. D. Kermish, C. L. Kuo, C. J. MacTavish, P. V. Mason, K. Megerian, L. Moncelsi, T. A. Morford, J. M. Nagy, C. B. Netterfield, R. O'Brient, C. Reintsema, J. E. Ruhl, M. C. Runyan, J. A. Shariff, J. D. Soler, A. Trangsrud, C. Tucker, R. S. Tucker, A. D. Turner, A. C. Weber, D. V. Wiebe, E. Y. Young, Pre-flight integration and characterization of the SPIDER balloon-borne telescope, in: Society of Photo-Optical Instrumentation Engineers (SPIE) Conference Series, 2014, p. 13. arXiv:1407.2906, doi:10.1117/12.2055683.
- [237] T. Higaki, K. S. Jeong, F. Takahashi, Solving the tension between high-scale inflation and axion isocurvature perturbations, Physics Letters B 734 (2014) 21-26. arXiv:1403.4186, doi:10.1016/j.physletb.2014.
 05.014.
- [238] B. Bozek, D. J. E. Marsh, J. Silk, R. F. G. Wyse, Galaxy UV-luminosity function and reionization constraints on axion dark matter, MNRAS450 (2015) 209-222. arXiv:1409.3544, doi:10.1093/mnras/stv624.

- [239] D. J. E. Marsh, A.-R. Pop, Axion dark matter, solitons and the cuspcore problem, MNRAS451 (2015) 2479-2492. arXiv:1502.03456, doi: 10.1093/mnras/stv1050.
- [240] R. Barkana, Z. Haiman, J. P. Ostriker, Constraints on Warm Dark Matter from Cosmological Reionization, ApJ558 (2001) 482-496. arXiv: astro-ph/0102304, doi:10.1086/322393.
- [241] A. J. Benson, A. Farahi, S. Cole, L. A. Moustakas, A. Jenkins, M. Lovell, R. Kennedy, J. Helly, C. Frenk, Dark matter halo merger histories beyond cold dark matter - I. Methods and application to warm dark matter, MNRAS428 (2013) 1774-1789. arXiv:1209.3018, doi:10.1093/mnras/ sts159.
- [242] E. W. Kolb, I. I. Tkachev, Nonlinear axion dynamics and the formation of cosmological pseudosolitons, Phys. Rev. D49 (1994) 5040-5051. arXiv: astro-ph/9311037, doi:10.1103/PhysRevD.49.5040.
- [243] E. Zackrisson, T. Riehm, Gravitational Lensing as a Probe of Cold Dark Matter Subhalos, Advances in Astronomy 2010 (2010) 9. arXiv:0905.
 4075, doi:10.1155/2010/478910.
- [244] F. Li, A. L. Erickcek, N. M. Law, A new probe of the small-scale primordial power spectrum: Astrometric microlensing by ultracompact minihalos, Phys. Rev. D86 (4) (2012) 043519. arXiv:1202.1284, doi: 10.1103/PhysRevD.86.043519.
- [245] R. J. Bouwens, G. D. Illingworth, P. A. Oesch, M. Trenti, I. Labbé,
 L. Bradley, M. Carollo, P. G. van Dokkum, V. Gonzalez, B. Holwerda,
 M. Franx, L. Spitler, R. Smit, D. Magee, UV Luminosity Functions at
 Redshifts z~4 to z~10: 10,000 Galaxies from HST Legacy Fields, ApJ803
 (2015) 34. arXiv:1403.4295, doi:10.1088/0004-637X/803/1/34.

[246] J. E. Gunn, B. A. Peterson, On the Density of Neutral Hydrogen in Intergalactic Space., ApJ142 (1965) 1633–1641. doi:10.1086/148444.

- [247] X. Fan, M. A. Strauss, R. H. Becker, R. L. White, J. E. Gunn, G. R. Knapp, G. T. Richards, D. P. Schneider, J. Brinkmann, M. Fukugita, Constraining the Evolution of the Ionizing Background and the Epoch of Reionization with z⁶ Quasars. II. A Sample of 19 Quasars, AJ132 (2006) 117–136. arXiv:astro-ph/0512082, doi:10.1086/504836.
- [248] C. Schultz, J. Oñorbe, K. N. Abazajian, J. S. Bullock, The high-z universe confronts warm dark matter: Galaxy counts, reionization and the nature of dark matter, MNRAS442 (2014) 1597–1609. arXiv:1401.3769, doi: 10.1093/mnras/stu976.
- [249] A. V. Kravtsov, A. A. Berlind, R. H. Wechsler, A. A. Klypin, S. Gottlöber,
 B. Allgood, J. R. Primack, The Dark Side of the Halo Occupation Distribution, ApJ609 (2004) 35-49. arXiv:astro-ph/0308519, doi: 10.1086/420959.
- [250] A. Vale, J. P. Ostriker, Linking halo mass to galaxy luminosity, MN-RAS353 (2004) 189-200. arXiv:astro-ph/0402500, doi:10.1111/j. 1365-2966.2004.08059.x.
- [251] R. A. Windhorst, S. H. Cohen, R. A. Jansen, C. Conselice, H. Yan, How JWST can measure first light, reionization and galaxy assembly, New Astronomy Reviews50 (2006) 113-120. arXiv:astro-ph/0506253, doi: 10.1016/j.newar.2005.11.018.
- [252] M. Kuhlen, C.-A. Faucher-Giguère, Concordance models of reionization: implications for faint galaxies and escape fraction evolution, MNRAS423 (2012) 862-876. arXiv:1201.0757, doi:10.1111/j.1365-2966.2012. 20924.x.
- [253] R. A. Sunyaev, Y. B. Zeldovich, Small-Scale Fluctuations of Relic Radiation, Ap&SS7 (1970) 3–19. doi:10.1007/BF00653471.

[254] A. Gruzinov, W. Hu, Secondary Cosmic Microwave Background

Anisotropies in a Universe Reionized in Patches, ApJ508 (1998) 435-439. arXiv:astro-ph/9803188, doi:10.1086/306432.

- [255] E. Calabrese, R. Hložek, N. Battaglia, J. R. Bond, F. de Bernardis, M. J. Devlin, A. Hajian, S. Henderson, J. C. Hil, A. Kosowsky, T. Louis, J. McMahon, K. Moodley, L. Newburgh, M. D. Niemack, L. A. Page, B. Partridge, N. Sehgal, J. L. Sievers, D. N. Spergel, S. T. Staggs, E. R. Switzer, H. Trac, E. J. Wollack, Precision epoch of reionization studies with next-generation CMB experiments, JCAP8 (2014) 10. arXiv:1406.4794, doi:10.1088/1475-7516/2014/08/010.
- [256] J. F. Navarro, C. S. Frenk, S. D. White, A Universal density profile from hierarchical clustering, ApJ490 (1997) 493-508. arXiv:astro-ph/ 9611107, doi:10.1086/304888.
- [257] H.-Y. Schive, M.-H. Liao, T.-P. Woo, S.-K. Wong, T. Chiueh, T. Broadhurst, W.-Y. P. Hwang, Understanding the Core-Halo Relation of Quantum Wave Dark Matter from 3D Simulations, Phys. Rev. Lett.113 (26) (2014) 261302. arXiv:1407.7762, doi:10.1103/PhysRevLett.113. 261302.
- [258] R. Ruffini, S. Bonazzola, Systems of Self-Gravitating Particles in General Relativity and the Concept of an Equation of State, Physical Review 187 (1969) 1767-1783. doi:10.1103/PhysRev.187.1767.
- [259] E. Seidel, W.-M. Suen, Oscillating soliton stars, Physical Review Letters 66 (1991) 1659–1662. doi:10.1103/PhysRevLett.66.1659.
- [260] A. R. Liddle, M. S. Madsen, The Structure and formation of boson stars, Int.J.Mod.Phys. D1 (1992) 101–144. doi:10.1142/S0218271892000057.
- [261] A. Diez-Tejedor, A. X. Gonzalez-Morales, No-go theorem for static scalar field dark matter halos with no Noether charges, Phys. Rev. D88 (6) (2013) 067302. arXiv:1306.4400, doi:10.1103/PhysRevD.88.067302.

- [262] F. S. Guzmán, L. A. Ureña-López, Gravitational Cooling of Selfgravitating Bose Condensates, ApJ645 (2006) 814-819. arXiv:astro-ph/ 0603613, doi:10.1086/504508.
- [263] E. Seidel, W.-M. Suen, Formation of solitonic stars through gravitational cooling, Physical Review Letters 72 (1994) 2516-2519. arXiv:gr-qc/9309015, doi:10.1103/PhysRevLett.72.2516.
- [264] A. X. González-Morales, A. Diez-Tejedor, L. A. Ureña-López, O. Valenzuela, Hints on halo evolution in scalar field dark matter models with galaxy observations, Phys. Rev. D87 (2) (2013) 021301. arXiv:1211. 6431, doi:10.1103/PhysRevD.87.021301.
- [265] A. Diez-Tejedor, A. X. Gonzalez-Morales, S. Profumo, Dwarf spheroidal galaxies and Bose-Einstein condensate dark matter, Phys. Rev. D90 (4) (2014) 043517. arXiv:1404.1054, doi:10.1103/PhysRevD.90.043517.
- [266] D. H. Weinberg, J. S. Bullock, F. Governato, R. Kuzio de Naray, A. H. G. Peter, Cold dark matter: controversies on small scales, ArXiv e-printsarXiv:1306.0913.
- [267] B. Moore, S. Ghigna, F. Governato, G. Lake, T. Quinn, J. Stadel, P. Tozzi, Dark Matter Substructure within Galactic Halos, ApJLett524 (1999) L19– L22. arXiv:astro-ph/9907411, doi:10.1086/312287.
- [268] A. Klypin, A. V. Kravtsov, O. Valenzuela, F. Prada, Where Are the Missing Galactic Satellites?, ApJ522 (1999) 82–92. arXiv:astro-ph/9901240, doi:10.1086/307643.
- [269] M. Boylan-Kolchin, J. S. Bullock, M. Kaplinghat, Too big to fail? The puzzling darkness of massive Milky Way subhaloes, MNRAS415 (2011) L40–L44. arXiv:1103.0007, doi:10.1111/j.1745-3933.2011.01074.x.
- [270] R. F. G. Wyse, G. Gilmore, Observed Properties of Dark Matter on Small Spatial Scales, in: J. I. Davies, M. J. Disney (Eds.), IAU Symposium,

Vol. 244 of IAU Symposium, 2008, pp. 44–52. arXiv:0708.1492, doi: 10.1017/S1743921307013828.

- [271] J. Oñorbe, M. Boylan-Kolchin, J. S. Bullock, P. F. Hopkins, D. Kerěs, C.-A. Faucher-Giguère, E. Quataert, N. Murray, Forged in FIRE: cusps, cores, and baryons in low-mass dwarf galaxies, ArXiv e-printsarXiv: 1502.02036.
- [272] A. Pontzen, F. Governato, Cold dark matter heats up, Nature506 (2014) 171-178. arXiv:1402.1764, doi:10.1038/nature12953.
- [273] D. N. Spergel, P. J. Steinhardt, Observational evidence for selfinteracting cold dark matter, Phys. Rev. Lett.84 (2000) 3760-3763. arXiv:astro-ph/ 9909386, doi:10.1103/PhysRevLett.84.3760.
- [274] A. Loeb, N. Weiner, Cores in Dwarf Galaxies from Dark Matter with a Yukawa Potential, Phys. Rev. Lett.106 (17) (2011) 171302. arXiv: 1011.6374, doi:10.1103/PhysRevLett.106.171302.
- [275] M. Vogelsberger, J. Zavala, C. Simpson, A. Jenkins, Dwarf galaxies in CDM and SIDM with baryons: observational probes of the nature of dark matter, MNRAS444 (2014) 3684-3698. arXiv:1405.5216, doi:10.1093/ mnras/stu1713.
- [276] J. A. Schewtschenko, R. J. Wilkinson, C. M. Baugh, C. Bœhm, S. Pascoli, Dark matter-radiation interactions: the impact on dark matter haloes, MNRAS449 (2015) 3587-3596. arXiv:1412.4905, doi:10.1093/mnras/ stv431.
- [277] H. W. Lin, A. Loeb, Scaling Relations of Halo Cores for Self-Interacting Dark Matter, ArXiv e-printsarXiv:1506.05471.
- [278] A. V. Macciò, S. Paduroiu, D. Anderhalden, A. Schneider, B. Moore, Cores in warm dark matter haloes: a Catch 22 problem, MNRAS424 (2012) 1105–1112. arXiv:1202.1282, doi:10.1111/j.1365-2966.2012. 21284.x.

- [279] A. V. Macciò, S. Paduroiu, D. Anderhalden, A. Schneider, B. Moore, Erratum: Cores in warm dark matter haloes: a Catch 22 problem, MNRAS428 (2013) 3715–3716. doi:10.1093/mnras/sts251.
- [280] S. Horiuchi, P. J. Humphrey, J. Oñorbe, K. N. Abazajian, M. Kaplinghat, S. Garrison-Kimmel, Sterile neutrino dark matter bounds from galaxies of the Local Group, Phys. Rev. D89 (2) (2014) 025017. arXiv:1311.0282, doi:10.1103/PhysRevD.89.025017.
- [281] S. Tremaine, J. E. Gunn, Dynamical role of light neutral leptons in cosmology, Phys. Rev. Lett.42 (1979) 407-410. doi:10.1103/PhysRevLett. 42.407.
- [282] M. G. Walker, J. Peñarrubia, A Method for Measuring (Slopes of) the Mass Profiles of Dwarf Spheroidal Galaxies, ApJ742 (2011) 20. arXiv: 1108.2404, doi:10.1088/0004-637X/742/1/20.
- [283] S. Shao, L. Gao, T. Theuns, C. S. Frenk, The phase-space density of fermionic dark matter haloes, MNRAS430 (2013) 2346-2357. arXiv: 1209.5563, doi:10.1093/mnras/stt053.
- [284] W. H. Press, B. S. Ryden, D. N. Spergel, Single mechanism for generating large-scale structure and providing dark missing matter, Physical Review Letters 64 (1990) 1084–1087. doi:10.1103/PhysRevLett.64.1084.
- [285] V. Sahni, L. Wang, New cosmological model of quintessence and dark matter, Phys. Rev. D62 (10) (2000) 103517. arXiv:astro-ph/9910097, doi:10.1103/PhysRevD.62.103517.
- [286] P. J. E. Peebles, Fluid Dark Matter, ApJLett534 (2000) L127–L129.
 arXiv:astro-ph/0002495, doi:10.1086/312677.
- [287] O. E. Gerhard, D. N. Spergel, Dwarf spheroidal galaxies and the mass of the neutrino, ApJLett389 (1992) L9–L11. doi:10.1086/186336.

- [288] M. L. Mateo, Dwarf Galaxies of the Local Group, ARA&A36 (1998) 435-506. arXiv:astro-ph/9810070, doi:10.1146/annurev.astro.36. 1.435.
- [289] V. Lora, J. Magaña, A. Bernal, F. J. Sánchez-Salcedo, E. K. Grebel, On the mass of ultra-light bosonic dark matter from galactic dynamics, JCAP2 (2012) 11. arXiv:1110.2684, doi:10.1088/1475-7516/2012/ 02/011.
- [290] F. Governato, D. Weisz, A. Pontzen, S. Loebman, D. Reed, A. M. Brooks,
 P. Behroozi, C. Christensen, P. Madau, L. Mayer, S. Shen, M. Walker,
 T. Quinn, B. W. Keller, J. Wadsley, Faint dwarfs as a test of DM models:
 WDM versus CDM, MNRAS448 (2015) 792-803. arXiv:1407.0022, doi:
 10.1093/mnras/stu2720.
- [291] C. P. Burgess, The Cosmological Constant Problem: Why it's hard to get Dark Energy from Micro-physics, ArXiv e-printsarXiv:1309.4133.
- [292] R. Bousso, J. Polchinski, Quantization of four-form fluxes and dynamical neutralization of the cosmological constant, Journal of High Energy Physics 6 (2000) 006. arXiv:hep-th/0004134, doi:10.1088/1126-6708/ 2000/06/006.
- [293] L. Susskind, The Anthropic Landscape of String Theory, in: The Davis Meeting On Cosmic Inflation, 2003, p. 26. arXiv:hep-th/0302219.
- [294] J. D. Brown, C. Teitelboim, Dynamical neutralization of the cosmological constant, Physics Letters B 195 (1987) 177–182. doi:10.1016/ 0370-2693(87)91190-7.
- [295] C. T. Hill, G. G. Ross, Models and new phenomenological implications of a class of pseudo-Goldstone bosons, Nuclear Physics B 311 (1988) 253–297.
 doi:10.1016/0550-3213(88)90062-4.
- [296] J. A. Frieman, C. T. Hill, A. Stebbins, I. Waga, Cosmology with Ultralight Pseudo Nambu-Goldstone Bosons, Physical Review Letters 75 (1995)

2077-2080. arXiv:astro-ph/9505060, doi:10.1103/PhysRevLett.75. 2077.

- [297] E. J. Copeland, M. Sami, S. Tsujikawa, Dynamics of Dark Energy, International Journal of Modern Physics D 15 (2006) 1753-1935. arXiv: hep-th/0603057, doi:10.1142/S021827180600942X.
- [298] V. Smer-Barreto, A. R. Liddle, Planck Satellite Constraints on Pseudo-Nambu-Goldstone Boson Quintessence, ArXiv e-printsarXiv:1503. 06100.
- [299] M. Kamionkowski, J. Pradler, D. G. E. Walker, Dark Energy from the String Axiverse, Phys. Rev. Lett.113 (25) (2014) 251302. arXiv:1409. 0549, doi:10.1103/PhysRevLett.113.251302.
- [300] T. C. Bachlechner, Axionic Band Structure of the Cosmological Constant, Phys. Rev. D93. arXiv:1510.06388.
- [301] D. Langlois, Inflation and Cosmological Perturbations, in: G. Wolschin (Ed.), Lecture Notes in Physics, Berlin Springer Verlag, Vol. 800 of Lecture Notes in Physics, Berlin Springer Verlag, 2010, pp. 1–57. arXiv:1001. 5259, doi:10.1007/978-3-642-10598-2_1.
- [302] D. Baumann, L. McAllister, Inflation and String Theory, ArXiv eprintsarXiv:1404.2601.
- [303] J. Martin, C. Ringeval, R. Trotta, V. Vennin, The best inflationary models after Planck, JCAP3 (2014) 39. arXiv:1312.3529, doi:10.1088/ 1475-7516/2014/03/039.
- [304] L. Kofman, A. Linde, A. A. Starobinsky, Towards the theory of reheating after inflation, Phys. Rev. D56 (1997) 3258-3295. arXiv:hep-ph/ 9704452, doi:10.1103/PhysRevD.56.3258.
- [305] R. Allahverdi, R. Brandenberger, F.-Y. Cyr-Racine, A. Mazumdar, Reheating in Inflationary Cosmology: Theory and Applications, Annual Re-

view of Nuclear and Particle Science 60 (2010) 27-51. arXiv:1001.2600, doi:10.1146/annurev.nucl.012809.104511.

- [306] R. Easther, R. Galvez, O. Özsoy, S. Watson, Supersymmetry, nonthermal dark matter, and precision cosmology, Phys. Rev. D89 (2) (2014) 023522. arXiv:1307.2453, doi:10.1103/PhysRevD.89.023522.
- [307] J. Martin, R. H. Brandenberger, Trans-Planckian problem of inflationary cosmology, Phys. Rev. D63 (12) (2001) 123501. arXiv:hep-th/0005209, doi:10.1103/PhysRevD.63.123501.
- [308] N. Kaloper, M. Kleban, A. Lawrence, S. Shenker, L. Susskind, Initial Conditions for Inflation, Journal of High Energy Physics 11 (2002) 037. arXiv:hep-th/0209231, doi:10.1088/1126-6708/2002/11/037.
- [309] K. Freese, J. A. Frieman, A. V. Olinto, Natural inflation with pseudo Nambu-Goldstone bosons, Physical Review Letters 65 (1990) 3233-3236. doi:10.1103/PhysRevLett.65.3233.
- [310] J. P. Conlon, Quantum gravity constraints on inflation, JCAP9 (2012) 19.
 arXiv:1203.5476, doi:10.1088/1475-7516/2012/09/019.
- [311] L. A. Boyle, R. R. Caldwell, M. Kamionkowski, Spintessence! New models for dark matter and dark energy, Physics Letters B 545 (2002) 17–22. arXiv:astro-ph/0105318, doi:10.1016/S0370-2693(02)02590-X.
- [312] E. Silverstein, A. Westphal, Monodromy in the CMB: Gravity waves and string inflation, Phys. Rev. D78 (10) (2008) 106003. arXiv:0803.3085, doi:10.1103/PhysRevD.78.106003.
- [313] L. McAllister, E. Silverstein, A. Westphal, Gravity waves and linear inflation from axion monodromy, Phys. Rev. D82 (4) (2010) 046003.
 arXiv:0808.0706, doi:10.1103/PhysRevD.82.046003.
- [314] G. Gur-Ari, Brane inflation and moduli stabilization on twisted tori, Journal of High Energy Physics 1 (2014) 179. arXiv:1310.6787, doi: 10.1007/JHEP01(2014)179.

- [315] D. Andriot, A no-go theorem for monodromy inflation, ArXiv eprintsarXiv:1510.02005.
- [316] R. Flauger, L. McAllister, E. Pajer, A. Westphal, G. Xu, Oscillations in the CMB from axion monodromy inflation, JCAP6 (2010) 9. arXiv: 0907.2916, doi:10.1088/1475-7516/2010/06/009.
- [317] P. D. Meerburg, D. N. Spergel, B. D. Wandelt, Searching for oscillations in the primordial power spectrum. I. Perturbative approach, Phys. Rev. D89 (6) (2014) 063536. arXiv:1308.3704, doi:10.1103/PhysRevD.89. 063536.
- [318] P. D. Meerburg, D. N. Spergel, B. D. Wandelt, Searching for oscillations in the primordial power spectrum. II. Constraints from Planck data, Phys. Rev. D89 (6) (2014) 063537. arXiv:1308.3705, doi:10.1103/PhysRevD. 89.063537.
- [319] R. Penrose, Gravitational Collapse: the Role of General Relativity, Nuovo Cimento Rivista Serie 1 (1969) 252.
- [320] R. Brito, V. Cardoso, P. Pani (Eds.), Superradiance, Vol. 906 of Lecture Notes in Physics, Berlin Springer Verlag, 2015. arXiv:1501.06570.
- [321] W. H. Press, S. A. Teukolsky, Floating Orbits, Superradiant Scattering and the Black-hole Bomb, Nature238 (1972) 211–212. doi:10.1038/ 238211a0.
- [322] W. H. Press, S. A. Teukolsky, Perturbations of a Rotating Black Hole.
 II. Dynamical Stability of the Kerr Metric, ApJ185 (1973) 649-674. doi: 10.1086/152445.
- [323] A. Arvanitaki, S. Dubovsky, Exploring the string axiverse with precision black hole physics, Phys. Rev. D83 (4) (2011) 044026. arXiv:1004.3558, doi:10.1103/PhysRevD.83.044026.

- [324] P. Pani, V. Cardoso, L. Gualtieri, E. Berti, A. Ishibashi, Black-Hole Bombs and Photon-Mass Bounds, Phys. Rev. Lett.109 (13) (2012) 131102.
 arXiv:1209.0465, doi:10.1103/PhysRevLett.109.131102.
- [325] R. Brito, V. Cardoso, P. Pani, Black holes as particle detectors: evolution of superradiant instabilities, Classical and Quantum Gravity 32 (13) (2015) 134001. arXiv:1411.0686, doi:10.1088/0264-9381/32/ 13/134001.
- [326] G. M. Harry, LIGO Scientific Collaboration, Advanced LIGO: the next generation of gravitational wave detectors, Classical and Quantum Gravity 27 (8) (2010) 084006. doi:10.1088/0264-9381/27/8/084006.
- [327] P. Amaro-Seoane, S. Aoudia, S. Babak, P. Binétruy, E. Berti, A. Bohé, C. Caprini, M. Colpi, N. J. Cornish, K. Danzmann, J.-F. Dufaux, J. Gair, O. Jennrich, P. Jetzer, A. Klein, R. N. Lang, A. Lobo, T. Littenberg, S. T. McWilliams, G. Nelemans, A. Petiteau, E. K. Porter, B. F. Schutz, A. Sesana, R. Stebbins, T. Sumner, M. Vallisneri, S. Vitale, M. Volonteri, H. Ward, Low-frequency gravitational-wave science with eLISA/NGO, Classical and Quantum Gravity 29 (12) (2012) 124016. arXiv:1202.0839, doi:10.1088/0264-9381/29/12/124016.
- [328] A. Khmelnitsky, V. Rubakov, Pulsar timing signal from ultralight scalar dark matter, JCAP2 (2014) 019. arXiv:1309.5888, doi:10.1088/ 1475-7516/2014/02/019.
- [329] F. A. Jenet, G. B. Hobbs, W. van Straten, R. N. Manchester, M. Bailes, J. P. W. Verbiest, R. T. Edwards, A. W. Hotan, J. M. Sarkissian, S. M. Ord, Upper Bounds on the Low-Frequency Stochastic Gravitational Wave Background from Pulsar Timing Observations: Current Limits and Future Prospects, ApJ653 (2006) 1571–1576. arXiv:astro-ph/0609013, doi: 10.1086/508702.

[330] N. K. Porayko, K. A. Postnov, Constraints on ultralight scalar dark matter

from pulsar timing, Phys. Rev. D90 (6) (2014) 062008. arXiv:1408.4670, doi:10.1103/PhysRevD.90.062008.

- [331] R. van Haasteren, Y. Levin, Understanding and analysing time-correlated stochastic signals in pulsar timing, MNRAS428 (2013) 1147–1159. arXiv: 1202.5932, doi:10.1093/mnras/sts097.
- [332] P. W. Graham, D. E. Kaplan, J. Mardon, S. Rajendran, W. A. Terrano, Dark Matter Direct Detection with Accelerometers, ArXiv e-printsarXiv: 1512.06165.
- [333] P. Sikivie, Experimental tests of the 'invisible' axion, Phys. Rev. Lett.51 (1983) 1415-1417. doi:10.1103/PhysRevLett.51.1415.
- [334] K. van Bibber, N. R. Dagdeviren, S. E. Koonin, A. K. Kerman, H. N. Nelson, Proposed experiment to produce and detect light pseudoscalars, Phys. Rev. Lett.59 (1987) 759–762. doi:10.1103/PhysRevLett.59.759.
- [335] H. Primakoff, Photo-Production of Neutral Mesons in Nuclear Electric Fields and the Mean Life of the Neutral Meson, Physical Review 81 (1951) 899-899. doi:10.1103/PhysRev.81.899.
- [336] K. Zioutas, S. Andriamonje, V. Arsov, S. Aune, D. Autiero, F. T. Avignone, K. Barth, A. Belov, B. Beltrán, H. Bräuninger, J. M. Carmona, S. Cebrián, E. Chesi, J. I. Collar, R. Creswick, T. Dafni, M. Davenport, L. di Lella, C. Eleftheriadis, J. Englhauser, G. Fanourakis, H. Farach, E. Ferrer, H. Fischer, J. Franz, P. Friedrich, T. Geralis, I. Giomataris, S. Gninenko, N. Goloubev, M. D. Hasinoff, F. H. Heinsius, D. H. Hoffmann, I. G. Irastorza, J. Jacoby, D. Kang, K. Königsmann, R. Kotthaus, M. Krčmar, K. Kousouris, M. Kuster, B. Lakić, C. Lasseur, A. Liolios, A. Ljubičić, G. Lutz, G. Luzón, D. W. Miller, A. Morales, J. Morales, M. Mutterer, A. Nikolaidis, A. Ortiz, T. Papaevangelou, A. Placci, G. Raffelt, J. Ruz, H. Riege, M. L. Sarsa, I. Savvidis, W. Serber,

P. Serpico, Y. Semertzidis, L. Stewart, J. D. Vieira, J. Villar, L. Walckiers, K. Zachariadou, First Results from the CERN Axion Solar Telescope, Phys. Rev. Lett.94 (12) (2005) 121301. arXiv:hep-ex/0411033, doi:10.1103/PhysRevLett.94.121301.

- [337] S. Andriamonje, S. Aune, D. Autiero, K. Barth, A. Belov, B. Beltrán, H. Bräuninger, J. M. Carmona, S. Cebrián, J. I. Collar, T. Dafni, M. Davenport, L. Di Lella, C. Eleftheriadis, J. Englhauser, G. Fanourakis, E. Ferrer Ribas, H. Fischer, J. Franz, P. Friedrich, T. Geralis, I. Giomataris, S. Gninenko, H. Gómez, M. Hasinoff, F. H. Heinsius, D. H. H. Hoffmann, I. G. Irastorza, J. Jacoby, K. Jakovcic, D. Kang, K. Königsmann, R. Kotthaus, M. Krcmar, K. Kousouris, M. Kuster, B. Lakic, C. Lasseur, A. Liolios, A. Ljubicic, G. Lutz, G. Luzón, D. Miller, A. Morales, J. Morales, A. Ortiz, T. Papaevangelou, A. Placci, G. Raffelt, H. Riege, A. Rodríguez, J. Ruz, I. Savvidis, Y. Semertzidis, P. Serpico, L. Stewart, J. Vieira, J. Villar, J. Vogel, L. Walckiers, K. Zioutas, CAST Collaboration, An improved limit on the axion photon coupling from the CAST experiment, JCAP4 (2007) 10. arXiv:hep-ex/0702006, doi:10.1088/1475-7516/2007/04/ 010.
- [338] M. Arik, S. Aune, K. Barth, A. Belov, S. Borghi, H. Bräuninger, G. Cantatore, J. M. Carmona, S. A. Cetin, J. I. Collar, E. Da Riva, T. Dafni, M. Davenport, C. Eleftheriadis, N. Elias, G. Fanourakis, E. Ferrer-Ribas, P. Friedrich, J. Galán, J. A. García, A. Gardikiotis, J. G. Garza, E. N. Gazis, T. Geralis, E. Georgiopoulou, I. Giomataris, S. Gninenko, H. Gómez, M. Gómez Marzoa, E. Gruber, T. Guthörl, R. Hartmann, S. Hauf, F. Haug, M. D. Hasinoff, D. H. H. Hoffmann, F. J. Iguaz, I. G. Irastorza, J. Jacoby, K. Jakovčić, M. Karuza, K. Königsmann, R. Kotthaus, M. Krčmar, M. Kuster, B. Lakić, P. M. Lang, J. M. Laurent, A. Liolios, A. Ljubičić, G. Luzón, S. Neff, T. Niinikoski, A. Nordt, T. Papaevangelou, M. J. Pivovaroff, G. Raffelt, H. Riege, A. Rodríguez, M. Rosu, J. Ruz, I. Savvidis, I. Shilon, P. S. Silva, S. K. Solanki, L. Stew-

art, A. Tomás, M. Tsagri, K. van Bibber, T. Vafeiadis, J. Villar, J. K. Vogel, S. C. Yildiz, K. Zioutas, CAST Collaboration, Search for Solar Axions by the CERN Axion Solar Telescope with He3 Buffer Gas: Closing the Hot Dark Matter Gap, Phys. Rev. Lett.112 (9) (2014) 091302. arXiv:1307.1985, doi:10.1103/PhysRevLett.112.091302.

- [339] J. K. Vogel, F. T. Avignone, G. Cantatore, J. M. Carmona, S. Caspi, S. A. Cetin, F. E. Christensen, A. Dael, T. Dafni, M. Davenport, A. V. Derbin, K. Desch, A. Diago, A. Dudarev, C. Eleftheriadis, G. Fanourakis, E. Ferrer-Ribas, J. Galan, J. A. Garcia, J. G. Garza, T. Geralis, B. Gimeno, I. Giomataris, S. Gninenko, H. Gomez, C. J. Hailey, T. Hiramatsu, D. H. H. Hoffmann, F. J. Iguaz, I. G. Irastorza, J. Isern, J. Jaeckel, K. Jakovcic, J. Kaminski, M. Kawasaki, M. Krcmar, C. Krieger, B. Lakic, A. Lindner, A. Liolios, G. Luzon, I. Ortega, T. Papaevangelou, M. J. Pivovaroff, G. Raffelt, J. Redondo, A. Ringwald, S. Russenschuck, J. Ruz, K. Saikawa, I. Savvidis, T. Sekiguchi, I. Shilon, H. Silva, H. H. J. ten Kate, A. Tomas, S. Troitsky, K. van Bibber, P. Vedrine, J. A. Villar, L. Walckiers, W. Wester, S. C. Yildiz, K. Zioutas, IAXO The International Axion Observatory, ArXiv e-printsarXiv:1302.3273.
- [340] A. Ayala, I. Domínguez, M. Giannotti, A. Mirizzi, O. Straniero, Revisiting the Bound on Axion-Photon Coupling from Globular Clusters, Physical Review Letters 113 (19) (2014) 191302. arXiv:1406.6053, doi:10.1103/ PhysRevLett.113.191302.
- [341] A. Payez, C. Evoli, T. Fischer, M. Giannotti, A. Mirizzi, A. Ringwald, Revisiting the SN1987A gamma-ray limit on ultralight axion-like particles, JCAP2 (2015) 006. arXiv:1410.3747, doi:10.1088/1475-7516/2015/ 02/006.
- [342] E. Massó, R. Toldrà, Light spinless particle coupled to photons, Phys.
 Rev. D52 (1995) 1755-1763. arXiv:hep-ph/9503293, doi:10.1103/ PhysRevD.52.1755.

- [343] G. Raffelt, A. Weiss, Red giant bound on the axion-electron coupling reexamined, Phys. Rev. D51 (1995) 1495-1498. arXiv:hep-ph/9410205, doi:10.1103/PhysRevD.51.1495.
- [344] M. Koshiba, Observational neutrino astrophysics, Phys. Rep.220 (1992)
 229–381. doi:10.1016/0370-1573(92)90083-C.
- [345] K. Ehret, M. Frede, S. Ghazaryan, M. Hildebrandt, E.-A. Knabbe, D. Kracht, A. Lindner, J. List, T. Meier, N. Meyer, D. Notz, J. Redondo, A. Ringwald, G. Wiedemann, B. Willke, New ALPS results on hidden-sector lightweights, Physics Letters B 689 (2010) 149–155. arXiv:1004.1313, doi:10.1016/j.physletb.2010.04.066.
- [346] R. Bähre, B. Döbrich, J. Dreyling-Eschweiler, S. Ghazaryan, R. Hodajerdi,
 D. Horns, F. Januschek, E.-A. Knabbe, A. Lindner, D. Notz, A. Ringwald,
 J. E. von Seggern, R. Stromhagen, D. Trines, B. Willke, Any light particle
 search II Technical Design Report, Journal of Instrumentation 8 (2013)
 9001. arXiv:1302.5647, doi:10.1088/1748-0221/8/09/T09001.
- [347] E. Zavattini, G. Zavattini, G. Ruoso, E. Polacco, E. Milotti, M. Karuza, U. Gastaldi, G. di Domenico, F. Della Valle, R. Cimino, S. Carusotto, G. Cantatore, M. Bregant, Experimental Observation of Optical Rotation Generated in Vacuum by a Magnetic Field, Physical Review Letters 96 (11) (2006) 110406. arXiv:hep-ex/0507107, doi:10.1103/PhysRevLett.96.110406.
- [348] A. S. Chou, W. Wester, A. Baumbaugh, H. R. Gustafson, Y. Irizarry-Valle, P. O. Mazur, J. H. Steffen, R. Tomlin, X. Yang, J. Yoo, Search for Axionlike Particles Using a Variable-Baseline Photon-Regeneration Technique, Physical Review Letters 100 (8) (2008) 080402. arXiv:0710.3783, doi:10.1103/PhysRevLett.100.080402.
- [349] E. Zavattini, G. Zavattini, G. Ruoso, G. Raiteri, E. Polacco, E. Milotti,V. Lozza, M. Karuza, U. Gastaldi, G. di Domenico, F. Della Valle,

R. Cimino, S. Carusotto, G. Cantatore, M. Bregant, New PVLAS results and limits on magnetically induced optical rotation and ellipticity in vacuum, Phys. Rev. D77 (3) (2008) 032006. arXiv:0706.3419, doi:10.1103/PhysRevD.77.032006.

- [350] J. Moody, F. Wilczek, NEW MACROSCOPIC FORCES?, Phys. Rev. D30 (1984) 130. doi:10.1103/PhysRevD.30.130.
- [351] G. Raffelt, Limits on a CP-violating scalar axion-nucleon interaction, Phys. Rev. D86 (1) (2012) 015001. arXiv:1205.1776, doi:10.1103/ PhysRevD.86.015001.
- [352] A. Arvanitaki, A. A. Geraci, Resonantly Detecting Axion-Mediated Forces with Nuclear Magnetic Resonance, Phys. Rev. Lett.113 (16) (2014) 161801. arXiv:1403.1290, doi:10.1103/PhysRevLett.113.161801.
- [353] G. Rybka, A. Wagner, K. Patel, R. Percival, K. Ramos, A. Brill, Search for dark matter axions with the Orpheus experiment, Phys. Rev. D91 (1) (2015) 011701. arXiv:1403.3121, doi:10.1103/PhysRevD.91.011701.
- [354] P. Sikivie, N. Sullivan, D. B. Tanner, Proposal for Axion Dark Matter Detection Using an LC Circuit, Physical Review Letters 112 (13) (2014) 131301. arXiv:1310.8545, doi:10.1103/PhysRevLett.112.131301.
- [355] Y. Kahn, B. R. Safdi, J. Thaler, A Broadband/Resonant Approach to Axion Dark Matter Detection, ArXiv e-printsarXiv:1602.01086.
- [356] M. Millea, L. Knox, B. D. Fields, New bounds for axions and axionlike particles with keV-GeV masses, Phys. Rev. D92 (2) (2015) 023010. arXiv:1501.04097, doi:10.1103/PhysRevD.92.023010.
- [357] E. Massó, R. Toldrà, New constraints on a light spinless particle coupled to photons, Phys. Rev. D55 (1997) 7967-7969. arXiv:hep-ph/9702275, doi:10.1103/PhysRevD.55.7967.

- [358] D. Cadamuro, J. Redondo, Cosmological bounds on pseudo Nambu-Goldstone bosons, JCAP2 (2012) 032. arXiv:1110.2895, doi:10.1088/ 1475-7516/2012/02/032.
- [359] J. Ellis, G. B. Gelmini, J. L. Lopez, D. V. Nanopoulos, S. Sarkar, Astrophysical constraints on massive unstable neutral relic particles, Nuclear Physics B 373 (1992) 399–437. doi:10.1016/0550-3213(92)90438-H.
- [360] V. Mukhanov, Nucleosynthesis Without Computer, International Journal of Theoretical Physics 43 (2004) 669–693. arXiv:astro-ph/0303073, doi:10.1023/B:IJTP.0000048169.69609.77.
- [361] G. Abbiendi, et al., Photonic events with missing energy in e+ e- collisions at S**(1/2) = 189-GeV, Eur. Phys. J. C18 (2000) 253-272. arXiv: hep-ex/0005002, doi:10.1007/s100520000522.
- [362] A. Heister, et al., Single photon and multiphoton production in e⁺e⁻ collisions at √s up to 209-GeV, Eur. Phys. J. C28 (2003) 1–13. doi: 10.1140/epjc/s2002-01129-7.
- [363] J. Abdallah, et al., Photon events with missing energy in e+ e- collisions at s**(1/2) = 130-GeV to 209-GeV, Eur. Phys. J. C38 (2005) 395-411.
 arXiv:hep-ex/0406019, doi:10.1140/epjc/s2004-02051-8.
- [364] C. Hearty, J. E. Rothberg, K. K. Young, A. S. Johnson, J. S. Whitaker, R. J. Wilson, G. Bartha, D. L. Burke, P. Extermann, P. H. Garbincius, C. A. Hawkins, M. J. Jonker, L. Keller, C. Matteuzzi, N. A. Roe, et al., Search for the anomalous production of single photons in e⁺e⁻ annihilation at sqrs =29 GeV, Phys. Rev. D39 (1989) 3207-3226. doi:10.1103/PhysRevD.39.3207.
- [365] K. Jedamzik, Big bang nucleosynthesis constraints on hadronically and electromagnetically decaying relic neutral particles, Phys. Rev. D74 (10) (2006) 103509. arXiv:hep-ph/0604251, doi:10.1103/PhysRevD.74. 103509.

- [366] F. Tavecchio, G. Ghisellini, L. Foschini, G. Bonnoli, G. Ghirlanda, P. Coppi, The intergalactic magnetic field constrained by Fermi/Large Area Telescope observations of the TeV blazar 1ES0229+200, MNRAS406 (2010) L70-L74. arXiv:1004.1329, doi:10.1111/j.1745-3933.2010. 00884.x.
- [367] A. Neronov, I. Vovk, Evidence for Strong Extragalactic Magnetic Fields from Fermi Observations of TeV Blazars, Science 328 (2010) 73-. arXiv: 1006.3504, doi:10.1126/science.1184192.
- [368] T. Yanagida, M. Yoshimura, Resonant axion-photon conversion in the early universe, Phys. Lett. B 202 (1988) 301–306. doi:10.1016/ 0370-2693(88)90475-3.
- [369] J. C. Mather, E. S. Cheng, D. A. Cottingham, R. E. Eplee, Jr., D. J. Fixsen, T. Hewagama, R. B. Isaacman, K. A. Jensen, S. S. Meyer, P. D. Noerdlinger, S. M. Read, L. P. Rosen, R. A. Shafer, E. L. Wright, C. L. Bennett, N. W. Boggess, M. G. Hauser, T. Kelsall, S. H. Moseley, Jr., R. F. Silverberg, G. F. Smoot, R. Weiss, D. T. Wilkinson, Measurement of the cosmic microwave background spectrum by the COBE FIRAS instrument, ApJ420 (1994) 439–444. doi:10.1086/173574.
- [370] D. J. Fixsen, E. S. Cheng, J. M. Gales, J. C. Mather, R. A. Shafer,
 E. L. Wright, The Cosmic Microwave Background Spectrum from the Full COBE FIRAS Data Set, ApJ473 (1996) 576. arXiv:astro-ph/9605054, doi:10.1086/178173.
- [371] H. Tashiro, J. Silk, D. J. E. Marsh, Constraints on primordial magnetic fields from CMB distortions in the axiverse, Phys. Rev. D88 (12) (2013) 125024. arXiv:1308.0314, doi:10.1103/PhysRevD.88.125024.
- [372] A. Mirizzi, J. Redondo, G. Sigl, Constraining resonant photon-axion conversions in the early universe, JCAP8 (2009) 1. arXiv:0905.4865, doi:10.1088/1475-7516/2009/08/001.

[373] A. Kogut, D. J. Fixsen, D. T. Chuss, J. Dotson, E. Dwek, M. Halpern, G. F. Hinshaw, S. M. Meyer, S. H. Moseley, M. D. Seiffert, D. N. Spergel, E. J. Wollack, The Primordial Inflation Explorer (PIXIE): a nulling polarimeter for cosmic microwave background observations, JCAP7 (2011) 25. arXiv:1105.2044, doi:10.1088/1475-7516/2011/07/025.

[374] P. André, C. Baccigalupi, A. Banday, D. Barbosa, B. Barreiro, J. Bartlett, N. Bartolo, E. Battistelli, R. Battye, G. Bendo, A. Benoît, J.-P. Bernard, M. Bersanelli, M. Béthermin, P. Bielewicz, A. Bonaldi, F. Bouchet, F. Boulanger, J. Brand, M. Bucher, C. Burigana, Z.-Y. Cai, P. Camus, F. Casas, V. Casasola, G. Castex, A. Challinor, J. Chluba, G. Chon, S. Colafrancesco, B. Comis, F. Cuttaia, G. D'Alessandro, A. Da Silva, R. Davis, M. de Avillez, P. de Bernardis, M. de Petris, A. de Rosa, G. de Zotti, J. Delabrouille, F.-X. Désert, C. Dickinson, J. M. Diego, J. Dunkley, T. Enßlin, J. Errard, E. Falgarone, P. Ferreira, K. Ferrière, F. Finelli, A. Fletcher, P. Fosalba, G. Fuller, S. Galli, K. Ganga, J. García-Bellido, A. Ghribi, M. Giard, Y. Giraud-Héraud, J. Gonzalez-Nuevo, K. Grainge, A. Gruppuso, A. Hall, J.-C. Hamilton, M. Haverkorn, C. Hernandez-Monteagudo, D. Herranz, M. Jackson, A. Jaffe, R. Khatri, M. Kunz, L. Lamagna, M. Lattanzi, P. Leahy, J. Lesgourgues, M. Liguori, E. Liuzzo, M. Lopez-Caniego, J. Macias-Perez, B. Maffei, D. Maino, A. Mangilli, E. Martinez-Gonzalez, C. J. A. P. Martins, S. Masi, M. Massardi, S. Matarrese, A. Melchiorri, J.-B. Melin, A. Mennella, A. Mignano, M.-A. Miville-Deschênes, A. Monfardini, A. Murphy, P. Naselsky, F. Nati, P. Natoli, M. Negrello, F. Noviello, C. O'Sullivan, F. Paci, L. Pagano, R. Paladino, N. Palanque-Delabrouille, D. Paoletti, H. Peiris, F. Perrotta, F. Piacentini, M. Piat, L. Piccirillo, G. Pisano, G. Polenta, A. Pollo, N. Ponthieu, M. Remazeilles, S. Ricciardi, M. Roman, C. Rosset, J.-A. Rubino-Martin, M. Salatino, A. Schillaci, P. Shellard, J. Silk, A. Starobinsky, R. Stompor, R. Sunyaev, A. Tartari, L. Terenzi, L. Toffolatti, M. Tomasi, N. Trappe, M. Tristram, T. Trombetti, M. Tucci, R. Van de Weijgaert,

B. Van Tent, L. Verde, P. Vielva, B. Wandelt, R. Watson, S. Withington, PRISM (Polarized Radiation Imaging and Spectroscopy Mission): an extended white paper, JCAP2 (2014) 6. arXiv:1306.2259, doi:10.1088/1475-7516/2014/02/006.

- [375] J. P. Conlon, M. C. D. Marsh, Excess Astrophysical Photons from a 0.1-1 keV Cosmic Axion Background, Phys. Rev. Lett.111 (15) (2013) 151301.
 arXiv:1305.3603, doi:10.1103/PhysRevLett.111.151301.
- [376] R. Lieu, J. P. D. Mittaz, S. Bowyer, J. O. Breen, F. J. Lockman, E. M. Murphy, C.-Y. Hwang, Diffuse Extreme-Ultraviolet Emission from the Coma Cluster: Evidence for Rapidly Cooling Gases at Submegakelvin Temperatures, Science 274 (1996) 1335–1338. doi:10.1126/science. 274.5291.1335.
- [377] F. V. Day, Cosmic axion background propagation in galaxies, ArXiv eprintsarXiv:1506.05334.
- [378] R. C. Hickox, M. Markevitch, Resolving the Unresolved Cosmic X-Ray Background in the Chandra Deep Fields, ApJLett661 (2007) L117–L121. arXiv:astro-ph/0702556, doi:10.1086/519003.
- [379] M. Pospelov, A. Ritz, C. Skordis, Pseudoscalar Perturbations and Polarization of the Cosmic Microwave Background, Phys. Rev. Lett.103 (5) (2009) 051302. arXiv:0808.0673, doi:10.1103/PhysRevLett.103. 051302.
- [380] D. Harari, P. Sikivie, Effects of a Nambu-Goldstone boson on the polarization of radio galaxies and the cosmic microwave background, Phys. Lett. B 289 (1992) 67–72. doi:10.1016/0370-2693(92)91363-E.
- [381] S. M. Carroll, G. B. Field, R. Jackiw, Limits on a Lorentz- and parityviolating modification of electrodynamics, Phys. Rev. D41 (1990) 1231– 1240. doi:10.1103/PhysRevD.41.1231.

- [382] H.-H. Mei, W.-T. Ni, W.-P. Pan, L. Xu, S. di Serego Alighieri, New Constraints on Cosmic Polarization Rotation from the ACTPol Cosmic Microwave Background B-mode Polarization Observation and the BICEP2 Constraint Update, ApJ805 (2015) 107. arXiv:1412.8569, doi:10.1088/0004-637X/805/2/107.
- [383] A. Lue, L. Wang, M. Kamionkowski, Cosmological Signature of New Parity-Violating Interactions, Physical Review Letters 83 (1999) 1506– 1509. arXiv:astro-ph/9812088, doi:10.1103/PhysRevLett.83.1506.
- [384] V. Gluscevic, D. Hanson, M. Kamionkowski, C. M. Hirata, First CMB constraints on direction-dependent cosmological birefringence from WMAP-7, Phys. Rev. D86 (10) (2012) 103529. arXiv:1206.5546, doi:10.1103/ PhysRevD.86.103529.
- [385] W. Zhao, M. Li, Fluctuations of cosmological birefringence and the effect on CMB B-mode polarization, Phys. Rev. D89 (10) (2014) 103518. arXiv: 1403.3997, doi:10.1103/PhysRevD.89.103518.
- [386] POLARBEAR Collaboration, P. A. R. Ade, K. Arnold, M. Atlas, C. Baccigalupi, D. Barron, D. Boettger, J. Borrill, S. Chapman, Y. Chinone, A. Cukierman, M. Dobbs, A. Ducout, R. Dunner, T. Elleflot, J. Errard, G. Fabbian, S. Feeney, C. Feng, A. Gilbert, N. Goeckner-Wald, J. Groh, G. Hall, N. W. Halverson, M. Hasegawa, K. Hattori, M. Hazumi, C. Hill, W. L. Holzapfel, Y. Hori, L. Howe, Y. Inoue, G. C. Jaehnig, A. H. Jaffe, O. Jeong, N. Katayama, J. P. Kaufman, B. Keating, Z. Kermish, R. Keskitalo, T. Kisner, A. Kusaka, M. Le Jeune, A. T. Lee, E. M. Leitch, D. Leon, Y. Li, E. Linder, L. Lowry, F. Matsuda, T. Matsumura, N. Miller, J. Montgomery, M. J. Myers, M. Navaroli, H. Nishino, T. Okamura, H. Paar, J. Peloton, L. Pogosian, D. Poletti, G. Puglisi, C. Raum, G. Rebeiz, C. L. Reichardt, P. L. Richards, C. Ross, K. M. Rotermund, D. E. Schenck, B. D. Sherwin, M. Shimon, I. Shirley, P. Siritanasak, G. Smecher, N. Stebor, B. Steinbach, A. Suzuki, J.-i. Suzuki, O. Tajima,

S. Takakura, A. Tikhomirov, T. Tomaru, N. Whitehorn, B. Wilson, A. Yadav, A. Zahn, O. Zahn, POLARBEAR Constraints on Cosmic Birefringence and Primordial Magnetic Fields, ArXiv e-printsarXiv:1509.02461.

- [387] R. Bott, Bull. Soc. Math. France 84 (1956) 251.
- [388] C. P. Burgess, An Introduction to Effective Field Theory, Annual Review of Nuclear and Particle Science 57 (2007) 329-362. arXiv:hep-th/0701053, doi:10.1146/annurev.nucl.56.080805.140508.
- [389] M. E. Peskin, D. V. Schroeder, An Introduction to Quantum Field Theory, Westview Press, 1995.
- [390] M. Srednicki, Quantum Field Theory, Cambridge University Press, 2007.
- [391] S. Dodelson, Modern cosmology, Academic Press, 2003.
- [392] M. P. Hobson, A. H. Jaffe, A. R. Liddle, P. Mukherjee, D. Parkinson, Bayesian Methods in Cosmology, Cambridge University Press, 2009.
- [393] D. J. E. Marsh, P. Bull, P. G. Ferreira, A. Pontzen, Quintessence in a quandary: Prior dependence in dark energy models, Phys. Rev. D90 (10) (2014) 105023. arXiv:1406.2301, doi:10.1103/PhysRevD.90.105023.
- [394] C. Long, L. McAllister, P. McGuirk, Heavy tails in Calabi-Yau moduli spaces, JHEP10 (2014) 187. arXiv:1407.0709, doi:10.1007/ JHEP10(2014)187.
- [395] B. Freivogel, Anthropic explanation of the dark matter abundance, JCAP3 (2010) 21. arXiv:0810.0703, doi:10.1088/1475-7516/2010/03/021.
- [396] R. Bousso, L. Hall, Why comparable? A multiverse explanation of the dark matter-baryon coincidence, Phys. Rev. D88 (6) (2013) 063503.
 arXiv:1304.6407, doi:10.1103/PhysRevD.88.063503.
- [397] A. Lewis, S. Bridle, Cosmological parameters from CMB and other data: A Monte Carlo approach, Phys. Rev. D66 (2002) 103511. arXiv: astro-ph/0205436, doi:10.1103/PhysRevD.66.103511.

- [398] J. Dunkley, M. Bucher, P. G. Ferreira, K. Moodley, C. Skordis, Fast and reliable Markov chain Monte Carlo technique for cosmological parameter estimation, MNRAS356 (2005) 925–936. arXiv:astro-ph/0405462, doi: 10.1111/j.1365-2966.2004.08464.x.
- [399] D. Foreman-Mackey, D. W. Hogg, D. Lang, J. Goodman, emcee: The MCMC Hammer, PASP125 (2013) 306-312. arXiv:1202.3665, doi:10. 1086/670067.
- [400] R. K. Sheth, G. Tormen, Large-scale bias and the peak background split, MNRAS308 (1999) 119-126. arXiv:astro-ph/9901122, doi:10.1046/ j.1365-8711.1999.02692.x.